\documentstyle[epsf,mystyle]{jarticle}
\setcounter{secnumdepth}{6}
\setcounter{tocdepth}{6}
\topsep=0.1cm
\parsep=0.1cm
\itemsep=0.0cm
\newenvironment{nquote}[1]%
{\list{}{\leftmargin=#1}\item[]}%
{\endlist}
\begin{document}
\input{a47}
\Ayonnana{$@MxMQ<T3F0L(J\\
}
{$@:n?^%D!<%k(J\\
tgif+$@!J%P!<%8%g%s(J2.2.4$@!K$N(J\\
$@;H$$J}!JBh(J1$@HG!K(J\\
$-$$@F|K\8l2=(J+$@&A$5$l$?(Jtgif$-$\\
}
{[$@J8=qHV9f(J]
}
{1991$@G/(J4$@7n(J22$@F|!J(J2$@7n(J22$@F|!A!K(J
}
{C\&C $@6&DL%=%U%H%&%'%"3+H/K\It(J\\
$@4D6-%7%9%F%`3+H/It(J\\
}
{$@>>ED980l(J
}
{}
\medskip
\pagestyle{tgif}
\medskip

\section{tgif+ $@$K$D$$$F(J}
\label{introduction}
\medskip
tgif+ $@$O!$4pK\E*$K(J X $@%&%#%s%I%&$rMQ$$$?:n?^%D!<%k$G$"$k(J tgif $@!J%P!<%8%g%s(J 2.2$@!$:n@.<T(J: William Chia-Wei Cheng$@!$(Jwilliam@cs.ucla.edu$@!K(J $@$rF|K\8l2=(J + $@&A(J $@$7$?$b$N$G$9(J
\footnote{$@F|K\8l2=(J + $@&A(J $@$N:n6H$O!$>>ED!J(J+ $@&A!K$,8D?M%Y!<%9$G9T$J$C$?$b$N$G$"$j!$(J
$@EvK\It$N%W%m%8%'%/%H!$$*$h$S6HL3$H$OA4$/4X78$J$$$3$H$rL@5-$7$F$*$-$^$9!%$J$*!$(JSystemV
$@2=$O!$EvK\It$NJ?>>7/!&Bg?{$5$s$,!$$^$?!$K\J8=q$N%G%P%C%0$O!$EvK\It$N(J
$@>.>B$5$s$,9T$J$C$F2<$5$$$^$7$?!%46<U!*(J}$@!%(J

$@K\J8=q(J\footnote{$@K\J8=q$O!$(Jnemacs + plain2 + jLaTeX + tgif+ $@$G:n@.$7$?$b$N$G$9!%(J}
$@$O!$(Jtgif+  $@$N;HMQJ}K!$K$D$$$F@bL@$7$?$b$N$G$9$,!$;HMQ$9$k>e$G:GDc8B$N(J
$@5!G=$K$D$$$F$7$+@bL@$7$F$"$j$^$;$s!%>\$7$$5!G=$O!$%=!<%9%3!<%I$r!$(J
$@$^$?!$%=!<%9%3!<%I$KIU?o$9$k(J tgif.man, prtgif.man$@$b;2>H$7$F2<$5$$!%(J
\medskip
tgif+ $@$G(J tgif $@$K$I$N$h$&$J2~B$$,$5$l$F$$$k$+$O!$<!$N!V(J
\S~\ref{historyAndBug}~$@2~B$MzNr$H%P%0%l%]!<%H!W$r;2>H$7$F2<$5$$(J
\footnote{tgif $@$K9T$J$C$?2~B$2U=j$NB?$/$O!$(Jtgif $@$N:n@.<T$KAw$C$F$*$j!$%*%j%8%J%k$N(J
tgif $@$KH?1G$5$l$kM=Dj$G$9$N$G!$$3$N(J tgif+ $@$N%$%s%?%U%'!<%9$O!$:#8e$N(J
tgif $@$N%$%s%?%U%'!<%9$H$=$l$[$I(J
$@$+$1$O$J$l$?$b$N$K$O$J$i$J$$$O$:$G$9!%(J}$@!%(J
\medskip
\par
$@%=!<%9%3!<%ICf$G!$!V(JUC$@!W!$!V(JKANJI$@!W!$!V(JKINPUT$@!W!$!V(JSYSV$@!W!$!V(J
DEBUG$@!W$G!$(J\#ifdef $@$"$k$$$O(J \#ifndef $@$5$l$F$$$kItJ,$O!$JQ99$r9T$J$C(J
$@$?2U=j$rI=$7$^$9!%(Jtgif $@$NF|K\8l2=$K:]$7$F!$(JKinput$@!$(JNemacs $@$N%=!<%9$r$+(J
$@$J$j0zMQ$5$;$FD:$-$^$7$?!%$3$N>l$r<Z$j$F46<U$N0U$rI=$7$^$9!%(J
\medskip
\par
$@$J$*!$$3$l$O$*4j$$$G$9$,!$(Jtgif$@!$(Jtgif+$@!$$"$k$$$OK\J8=q$K4X$9$k(J $@%P%0!$MWK>!$<B:]$K(J
$@9T$J$C$?JQ99Ey$,$4$6$$$^$7$?$i;d$K$b$*CN$i$;2<$5$$!%$^$?!$%=!<%9%3!<(J
$@%I$N:FG[IU$N:]$K$O!$$G$-$^$7$?$i!$G[IU@h$r$*CN$i$;2<$5$$!%(J
\medskip
\par
$@$I$J$?$+K\J8=q$r1Q8l$KK]Lu$7$F2<$5$$!%(J
\medskip
%$@$3$NJ8=q$O!$>>ED$,8D?M$N@UG$$G:n@.$7$?$b$N$G$"$j!$2q<R$OL54X78$G$9!%(J
\medskip
\begin{center}
Copyright (c) 1991$@!$>>ED980l(J
\\
\end{center}
\medskip
   \begin{center}
        \epsfile{file=FIGURE/kao.eps,hscale=0.8,vscale=0.8}

$@Cx<T6a1F!J%$%s%I$K$F!K(J

\vspace*{0.5cm}
$@F|K\EE5$3t<02q<R(J

C\&C$@6&DL%=%U%H%&%'%"3+H/K\It(J $@4D6-%7%9%F%`3+H/It(J

$@>>ED980l(J (Kou1 Ma2da)

(matsuda@ccs.mt.nec.co.jp, ex 820-24822, mail 20-24820, tel 5476-1106)
\end{center}
\medskip
\newpage
\medskip

\section{$@$O$8$a$F;H$o$l$kJ}$N$?$a$N%a%b(J}
\label{forBeginner}
\medskip

\subsection{$@K\J8=q$NFI$_J}(J}
\medskip
\par
$@=i$a$F(J tgif+ $@$r;H$o$l$kJ}$O!$(J
$@!V(J\S~\ref{facility}~$@5!G=35MW!W$H!V(J\S~\ref{howToDo}~$@5/F0!$:n?^!$J]B8!$0u:~!$=*N;!W(J
$@$H!V(J\S~\ref{howToInclude}~jLaTeX $@$X$N?^$N<h$j9~$_!W$@$1$rFI$a$P!$$H$j$"$((J
$@$:;H$($k$h$&$K$J$j$^$9!%$^$?!$$[$H$s$I$N>l9g!$$=$l$G==J,$G$9!%(J

tgif+ $@$,!$(JUNIX $@>e$NJ8=q:n@.:n6H$G!$$I$N$h$&$J0LCV$K$"$k$N$+$O!$?^(J~\ref{position}~$@$r(J
$@;2>H$7$F2<$5$$!%(J
\medskip
\par
$@$J$*!$$3$N%^%K%e%"%k$O!$:G=i$+$i:G8e$^$GFI$`$H8@$&7A<0$G$O$J$/!$L\<!$r8+$J$,$i(J
$@I,MW$J>l=j$rFI$a$P$$$$$h$&$K!J$H$J$k$3$H$r5'$C$F(J:$-$)$@!K=q$+$l$F$$$^$9!%(J
$@$=$N$?$aF1$8$h$&$J5-=R$r$$$m$$$m$J>l=j$G;H$C$F$*$j!$%^%K%e%"%k$,8|$/$J$C$F(J
$@$7$^$C$F$$$^$9(J:$-$($@!%(J
\medskip

\subsection{$@$3$l$^$G!$B>$N:n?^%D!<%k!J(JMacDraw $@Ey!K$r;H$C$F$i$7$?>l9g(J}
\medskip
\par
$@<!$NE@$KCm0U$7$F2<$5$$!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@5!G=$O!$(JMacDraw + $@&ADxEY$G$9!%(J
\medskip
\item $@0l$D$NJ8;zNs%*%V%8%'%/%H$NCf$GJ#?t$N%U%)%s%H$r;HMQ$9$k$3$H$O$G$-$^$;$s!%(J
\medskip
\item $@IA$1$kKg?t$O!$(J1 $@Kg$G$9!%(J
\medskip
\item $@J8;zNs!$%S%C%H%^%C%W%G!<%?$N3HBg!$=L>.$O$G$-$^$;$s(J
\footnote{$@J8;zNs$N%U%)%s%H%5%$%:$O!$JQ$($i$l$^$9!%(J}$@!%(J
\medskip
\item $@%*%V%8%'%/%H$N2sE>$O!$(J90 $@EYC10L$G$7$+$G$-$^$;$s!%(J
\medskip
\medskip
\medskip
\medskip
\end{enumerate}
}

\section{$@5l%P!<%8%g%s(J(tgif+ 1.15.3)$@$N%f!<%6$N$?$a$N%a%b(J}
\label{specialNote}
\medskip

\subsection{$@%*%j%8%J%k$N(J tgif $@$NJQ99(J}
\medskip
\begin{itemize}
\item $@N)$A>e$,$j$,CY$/$J$j$^$7$?!%(J
\medskip
\item $@07$($kIA2h%*%V%8%'%/%H$,A}$($^$7$?!%(J
\medskip
\par
$@1_8L!$3Q4];M3Q!$(JX11$@$N%S%C%H%^%C%W!$GK@~!%(J
\medskip

        xwd $@$r(J X11 $@$N(J $@%S%C%H%^%C%W$KJQ49$9$kJ}K!$O!$(Jpbmplus $@$,(J
        $@%$%s%9%H!<%k$5$l$F$$$l$P!$<!$N$h$&$K$9$l$P2DG=$G$9(J
        $@!J!VIUO?(J~\ref{generateXbm}~xwd $@$N7k2L$r(J X11 $@$N%S%C%H%^%C%W7A<0!J(J.xbm$@!KJQ49$9$kJ}K!!W;2>H!K!%(J
\medskip
\baselineskip=10pt
\begin{verbatim}
% xwd | xwdtopnm | pbmtoxbm > nantoka.xbm
\end{verbatim}
\medskip
\item $@3HD%;R$,JQ$o$j$^$7$?!%(J
\medskip
        $@=>Mh!$!V(J.ps$@!W(J $@$@$C$?$b$N$O!$!V(J.eps$@!W$K$J$j$^$7$?!%(J
        $@$^$?!$%S%C%H%^%C%W%U%!%$%k$K$O!$!V(J.xbm$@!W$N3HD%;R$,I,MW$G$9!J!V(J\S~\ref{files}~tgif+ $@$N@8@.$9$k%U%!%$%k!W;2>H!K!%(J
\medskip
\end{itemize}

\subsection{tgif+ $@$NJQ99(J}
\medskip
\par
$@FC$K!$<!$N(J4$@E@$,Bg$-$/JQ99$K$J$j$^$7$?$N$G!$Cm0U$7$F2<$5$$!%(J
\medskip
\begin{itemize}
\item $@%a%$%s%a%K%e!<$N(J wired down $@5!G=$NDI2C(J
\medskip

        $@"*(J $@1&%\%?%s$G!$%a%$%s%a%K%e!<$rI=<($7$?>uBV$G!$%I%i%C%0$9$k$H%a%$%s%a(J
        $@%K%e!<$,D%$jIU$/!J(J2$@8D$^$G!K!%??Cf%\%?%s$G>C5n!%1&%\%?%s$G0\F0!J!V(J\S~\ref{displayMainMenu}~$@%a%$%s%a%K%e!<$NI=<(!W;2>H!K!%(J
\medskip
\item Fake $@%U%)%s%H5!G=$NDI2C(J
\medskip
\end{itemize}
\begin{quote}
        $@"*(JX $@%5!<%P$,!$4A;z%U%)%s%H$H$7$F(J k14 $@$7$+;}$C$F$$$J$$>l9g!J$[$H$s$I$N(J X $@%5!<%P$,$I$&$G$9$,!K$K!$(Jtgif+ $@$G;HMQ$7$F$$$kB>$N%5%$%:$N%U%)%s%H$r%G%#%9%W%l%$$KI=<($9$k>l9g$K(J k14 $@$GBeMQ(J
        $@$9$k5!G=!%BeMQ$7$?>l9g$O!$J8;zNs$N<B:]$N%5%$%:$,J8;zNsOH$H$7$FI=<($5$l(J
        $@$k!J!V(J\S~\ref{displayKanji}~$@<B:]$NI=<(Iw7J!W(J $@;2>H!K!%(J
\end{quote}
\medskip
\par

$@$3$l0J30$NJQ99FbMF$O!$!V(J\S~\ref{historyAndBug} $@2~B$MzNr$H%P%0%l%]!<%H!W$N(J
$@!V(J2.2.1$@!W0J9_$r(J
$@;2>H$7$F2<$5$$!%(J
\medskip
\medskip
\medskip
        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/position.eps}
                \caption{tgif+ $@$N0LCVIU$1(J}
                \label{position}
           \end{center}
        \end{figure}
\newpage
        \tableofcontents
        \listoffigures

        \vspace*{3cm}
           \begin{center}
                \epsfile{file=FIGURE/tgif.eps,vscale=0.8,hscale=0.8}
           \end{center}
\medskip
\newpage

\section{$@2~B$MzNr$H%P%0%l%]!<%H(J}
\label{historyAndBug}
\medskip

\subsection{$@2~HGMzNr(J}
\label{history}
\medskip
~\\
\begin{tabular}{|l|l|l|l|}
\hline
$@HG(J     & \multicolumn{1}{|c|}{ $@FbMF(J                                                } & contrib  & $@:NH](J\\
\hline\hline
1.9.1  & \multicolumn{1}{|c|}{       17/07/90$@!J(JCCS $@%0%k!<%WFb%j%j!<%9!K(J            } &          &     \\
       & $@!|(J $-$display $@%*%W%7%g%s$NIU2C(J                          &          &     \\
       & $@!|%a%K%e!<%P!<$rIU$1$k!%(J                              &          &     \\
       & $@!|(JButton2 $@$G!V(JMode $@%a%K%e!<!W$,%]%C%W%"%C%W$9$k$h$&$K(J & 23/03/91 & $@"$(J  \\
       & $@$9$k!%(J                                                &          &     \\
       & $@!|F|K\8l$,I=<(!$0u:~2DG=$K$9$k!%(J                      &          &     \\
       & $@!|%a%K%e!<$K%+%l%s%H$NIA2hB0@-$rI=<($9$k$h$&$K$9$k(J    & 04/04/91 & $@!{(J  \\
\hline
1.9.2  & \multicolumn{1}{|c|}{       24/07/90$@!J(JCCS $@%0%k!<%WFb%j%j!<%9!K(J            } &          &     \\
       & $@!|(JNemacs $@$+$i(J Copy\&Paste $@2DG=$K$9$k(J($@F|K\8l$b2DG=!K(J    &          &     \\
\hline
1.12.1 & \multicolumn{1}{|c|}{       08/08/91$@!J(JCCS $@%0%k!<%WFb%j%j!<%9!K(J            } &          &     \\
       & $@!|(JDotOnGrid $@5!G=$NDI2C(J                                &          &     \\
       & $@"*%0%j%C%IE@$NI=<(4V3V$r%0%j%C%I4V3V$K$9$k5!G=(J        &          &     \\
       & $@!|(J tgif+ $@4V$N(J $@%3%T!<(J\&$@%Z!<%9%H5!G=$NDI2C(J               & 15/08/90 & $@!{(J  \\
       & $@"*(J tgif+ $@4V$G(J $@%*%V%8%'%/%H$N%3%T!<(J\&$@%Z!<%9%H$,$G$-$k$h(J &          &     \\
       & $@$&$K$9$k!%(J                                            &          &     \\
       & $@!|(JJstar $@Iw$N!VF1MM!W5!G=!J(JSameAs$@!K$NDI2C(J              & 15/08/90 & $@!_(J  \\
\hline
1.12.2 & \multicolumn{1}{|c|}{       ??/08/90$@!J(JCCS $@%0%k!<%WFb%j%j!<%9!K(J            } &          &     \\
       & $@!|(J $-$r $@%*%W%7%g%s(J $@$NIU2C(J                               &          &     \\
       & $@!|(J $@1_$NIA2h$r9bB.2=(J                                   &          &     \\
\hline
1.15.1 & \multicolumn{1}{|c|}{       ??/??/90$@!J(JCCS $@%0%k!<%WFb%j%j!<%9!K(J            } &          &     \\
       & $@!|(JUNDO $@5!G=$NDI2C(J                                     & 04/09/90 & $@!_(J  \\
       & $@!|%Z!<%9%H5!G=$N3HD%(J                                  &          &     \\
       & $@"*%F%-%9%H%+!<%=%k$,I=<($5$l$F$$$k;~$O!$$=$N>l=j$K(J    &          &     \\
\hline
1.15.3 & \multicolumn{1}{|c|}{       12/09/90$@!J(JCCS $@%0%k!<%WFb%j%j!<%9!K(J            } &          &     \\
       & $@!|(Jkinput $@$,;H$($k$h$&$K$9$k(J                           &          &     \\
\hline
       & \multicolumn{1}{|c|}{       15/10/90$@!J<RFb%j%j!<%9!K(J                      } &          &     \\
       & $@!|(JEXT $@%U%)%s%HBP1~DI2C(J                                &          &     \\
       & $@"*(JEUC $@%(%s%3!<%G%#%s%0$N4A;z%U%)%s%H$,$J$$(J PostScript &          &     \\
       & $@%W%j%s%?$KBP1~(J                                        &          &     \\
\hline
\end{tabular}\\
\medskip
\medskip
\par
$@I=Cf!$!V(Jcontrib$@!W$O!$:n@.<T$X$N%Q%C%A%U%!%$%k$NAwIUF|$r!$!V:NH]!W$OAwIU$7$?(J
$@5!G=$N:NH]$rI=$7$F$$$^$9!J!{$O:NMQ!$!_$OIT:NMQ!$"$$O=$@5$5$l$F:NMQ!K!%(J
\medskip
\newpage
~\\
\begin{tabular}{|c|l|l|l|}
\hline
$@HG(J     & \multicolumn{1}{|c|}{ $@FbMF(J                                                 } & contrib  & $@:NH](J\\
\hline\hline
 2.2.1 & \multicolumn{1}{|c|}{       19/03/91$@!J(JCCS $@%0%k!<%WFb%j%j!<%9!K(J             } &          &     \\
       & $@!|!V(JChoice$@%&%#%s%I%&!WFb$N%S%C%H%^%C%W$NJQ99(J           & 10/03/91 & $@"$(J  \\
       & $@"*@~$N!VB@$5!W!VLp0u!W!VGK@~!W$N%"%$%3%s$,!VB@$5(J1$@!W(J    &          &     \\
       & $@!VLp0u$J$7!W!V<B@~!W$N>l9g$KJ,$+$j$K$/$$$N$G!$!V(JWIDTH  &          &     \\
       & $@!W!V(JARROW$@!W!V(JDASH$@!W$NJ8;z$r%"%$%3%s$KIU2C$7$?!%(J        &          &     \\
       & $@!|%a%$%s%a%K%e!<$N(J wired down $@5!G=$NDI2C(J               & 04/04/91 & $@!{(J  \\
       & $@"*(J $@%a%$%s%a%K%e!<$rI=<($7$?>uBV$G!$%^%&%9$r%I%i%C%0$9(J  &          &     \\
       & $@$k$H%a%$%s%a%K%e!<$,D%$jIU$/!J(J2$@8D$^$G!K!%(J              &          &     \\
       & $@??Cf%\%?%s$G>C5n!%1&%\%?%s$G0\F0!%(J                     &          &     \\
       & $@!|(J ShowSize $@5!G=$NDI2C(J                                 &          &     \\
       & $@"*IA2h;~$K%*%V%8%'%/%H$N%5%$%:$rI=<($9$k!%(J             &          &     \\
       & $@!|(J ZoomIn $@5!G=$N3HD%(J                                   &          &     \\
       & $@"*(J 2 $@G\!$(J4 $@G\$^$G3HBg2DG=!%%U%)%s%H$O!$3HBg;~$K$O!$(J    &          &     \\
       & Fake $@%U%)%s%H$H$J$k!%(J                                  &          &     \\
       & $@!|(Jkinput $@$N<+F0@Z$jN%$75!G=(J                            &          &     \\
       & $@"*%F%-%9%H%b!<%I$r=*N;$9$k$H<+F0E*$K(J kinput $@$r=*N;$9$k(J &          &     \\
       & $@!|(JDotOnGrid $@5!G=$r(JLayout$@%a%K%e!<$KDI2C(J                 &          &     \\
       & $@!|(JFake $@%U%)%s%H5!G=$NDI2C(J                              &          &     \\
       & $@"*(Jk14 $@$7$+4A;z%U%)%s%H$,$J$$>l9g$K!$B>$N%5%$%:$N%U%)(J   &          &     \\
       & $@%s%H$r(J k14 $@$GBeMQ$9$k5!G=!%BeMQ$7$?>l9g$O!$J8;zNs$N(J    &          &     \\
       & $@<B:]$N%5%$%:$,J8;zNsOH$H$7$FI=<($5$l$k!%(J               &          &     \\
       & $@!|;HMQ$9$k4A;z%U%)%s%H$r(J\verb+~+/.Xdefaults $@$G;XDj2DG=$K$9$k(J  &          &     \\
       & $@!|%+%l%s%H%b!<%I$N%W%C%7%e5!G=(J                         & 23/03/91 & $@!{(J  \\
       & $@"*(J\verb+^+E $@$G!$%+%l%s%H$NIA2h%b!<%I$rA*Br%b!<%I$K$9$k!%(J\verb+^+E $@$G(J &          &     \\
       & $@85$NIA2h%b!<%I$KLa$k!J>>2<(J@CS$@8&$5$s!K!%(J                &          &     \\
       & $@!|%G%U%)%k%H%+!<%=%k$N;XDj5!G=(J                         & 23/03/91 & $@!{(J  \\
       & $@"*!VA*Br!W%b!<%I$N%+!<%=%k$N%+!<%=%k$N7A>u$,;XDj2DG=(J   &          &     \\
       & $@!J!V(JTgif*DefaultCursor: $@!W$rDI2C!K(J                     &          &     \\
       & $@!|IA2h%+!<%=%k$N;XDj5!G=(J                               & 23/03/91 & $@!{(J  \\
       & $@"*IA2h;~$N%+!<%=%k$N%+!<%=%k$N7A>u$,;XDj2DG=(J           &          &     \\
       & $@!J!V(JTgif*DrawCursor: $@!W$rDI2C!K(J                        &          &     \\
       & $@!|(JHideDragCursor $@5!G=$rDI2C(J                            &          &     \\
       & $@"*%I%i%C%0Cf$N%+!<%=%k$rI=<($7$J$$(J                     &          &     \\
       & $@!J!V(J$-$hc$@!W!$!V(JTgif*HideDragCursor: on$@!W!K(J               &          &     \\
       & $@!|(JEXT $@%U%)%s%H$r%G%U%)%k%H$K(J                           &          &     \\
       & $@"*!V(J$-$euc$@!W!V(JTgif*Euc: on$@!W$G(JEUC$@%U%)%s%H$,;XDj2DG=(J      &          &     \\
       & $@!|%G%U%)%k%H$N%W%j%s%H%"%&%H@h$,;XDj2DG=(J               & 23/03/91 & $@!{(J  \\
       & $@"*(J $@!V(JTgif*WhereToPrint:$@!W$G;XDj2DG=(J($@!V(J$-$eps$@!W$G$b2D!K!%(J &          &     \\
       & $@!|%;%s%A%a!<%?C10L7O$NF3F~!J%P%0$"$j!K(J                 &          &     \\
       & $@"*%;%s%A%a!<%?C10L$GIA2h$G$-$k$h$&$K$9$k!%(J             &          &     \\
       &                                                        &          &     \\
\hline
\end{tabular}\\
\newpage
~\\
\begin{tabular}{|l|l|r|r|}
\hline
$@HG(J    & \multicolumn{1}{|c|}{ $@FbMF(J                                                } & contrib & $@:NH](J\\
\hline\hline
2.2.2 & \multicolumn{1}{|c|}{       25/03/91$@!J(JCCS $@%0%k!<%WFb%j%j!<%9!K(J            } &         &     \\
      & $@!|%P%0%l%]!<%H!J(J2.2.2 $@HG!K;2>H(J                        &         &     \\
\hline
2.2.3 & \multicolumn{1}{|c|}{       28/03/91$@!J(JCCS $@%0%k!<%WFb%j%j!<%9!K(J            } &         &     \\
      & $@!|(J $-$noKanji $@%*%W%7%g%s$rIU2C(J                          &         &     \\
      & $@"*4A;z$r4^$^$J$$(J .eps $@$r:n@.$9$k>l9g$K;XDj(J            &         &     \\
      & $@!|%P%0%l%]!<%H!J(J2.2.3 $@HG!K;2>H(J                        &         &     \\
\hline
2.2.4 & \multicolumn{1}{|c|}{       10/05/91$@!J<RFb%j%j!<%9!K(J                      } &         &     \\
      & $@!|(Jkinput $@5/F0;~$K!V(JTitle $@%&%#%s%I%&!W$K!V(J*KINPUT*$@!W$H(J &         &     \\
      & $@I=<($5$l$k$h$&$K$9$k!%(J                                &         &     \\
      & $@!|(Jkinput $@$r<+F0E*$K5/F0$9$k5!G=$rIU2C(J                 &         &     \\
      & $@"*!V(J$-$autokinput $@$^$?$O(J $-$ak$@!W$+!V(JTgif*AutoKinput: on$@!W(J &         &     \\
      & $@!|%U%!%$%k%;!<%V;~$K!V(J.eps$@%U%!%$%k!W$r@8@.$9$k5!G=$r(J  &         &     \\
      & $@IU2C$9$k"*!V(J$-$swe$@!W!V(JTgif*SaveWithEPS: on$@!W(J            &         &     \\
      & $@!|%P%0%l%]!<%H!J(J2.2.4 $@HG!K;2>H(J                        &         &     \\
\hline
\end{tabular}\\
\medskip

\subsection{$@%P%0%l%]!<%H(J}
\label{bugReport}
\medskip
~\\
\begin{tabular}{|l|l|l|l|}
\hline
$@HG(J    & \multicolumn{1}{|c|}{ $@FbMF(J                                                 } & contrib  & $@:NH](J\\
\hline\hline
2.2.2 & $@!|?eJ?!J?bD>!K@~$r%3%T!<(J\&$@%Z!<%9%H$9$k$H%Z!<%9%H;~$K8+(J  & 21/03/91 & $@!{(J  \\
      & $@$($J$/$J$k!%(J                                           &          &     \\
      & $@"*(J SelBox() in seletc.c                                &          &     \\
      & $@?eJ?!J?bD>!K@~$NMM$J(J $@;O=*E@$N(J x $@:BI8$^$?$O(J y $@:BI8$,(J    &          &     \\
      & $@F1$8>l9g$K(J XOR $@$GIA2h$7$F$$$k$?$a2hLL$KI=<($5$l$J$$!%(J  &          &     \\
      & $@?eJ?!J?bD>!K@~(J $@$N%A%'%C%/$r$9$k$h$&$K$9$k(J              &          &     \\
      & $@!|%W%j%s%H%"%&%H$9$k$H?^$,C<$,@Z$l$k!%(J                 & 24/03/91 & $@!{(J  \\
      & $@"*(J DumpBBox() in file.c                                &          &     \\
      & $@!V(J\%\%BoundingBox: $@!W$N7W;;$,!$(Jprintf()$@$G@Z$j<N$F$i(J      &          &     \\
      & $@<B:]$h$j>.$5$/$J$C$F$$$k$?$a!$@Z$j>e$2$k$h$&$K$9$k!%(J   &          &     \\
\hline
2.2.3 & $@!|(JGroup$@2=!$(JUngroup$@2=$N:]$K!$ITI,MW$J:FIA2h$,9T$o$l$k(J   & 27/03/91 & $@!_(J  \\
      & $@"*(J Group() in select.c, Ungroup() in group.c           &          &     \\
      & $@ITI,MW$J:FIA2h$7$J$$$h$&$K$9$k!%(J                       &          &     \\
\hline
2.2.4 & $@!|%+%i!<%G%#%9%W%l%$$N>l9g$HGr9u$N>l9g$G!$%G%U%)%k%H(J   & 30/03/91 & $@!{(J  \\
      & $@$N%&%#%s%I%&$N?'$,0c$&!%(J                               &          & 1/4 \\
      & $@"*(J InitColor() in color.c                              &          &     \\
      & $@C1$J$k%W%m%0%i%_%s%0%_%9(J                               &          &     \\
      & $@!|%+%i!<%G%#%9%W%l%$$N>l9g!$(JReverseVideo $@$r;XDj$9$k(J    & 30/03/91 & $@"$(J  \\
      & $@$H%G%U%)%k%H$G$O!$%&%#%s%I%&$NOH$,I=<($5$l$J$$!%(J       &          & 1/4 \\
      & $@"*(J InitColor () in color.c                             &          &     \\
      & $@%+%i!<%G%#%9%W%l%$$N>l9g!$(JReverseVideo $@$N=hM}$r9T$C$F(J  &          &     \\
      & $@$$$J$$$N$G!$%&%#%s%I%&$N(JForeground$@%+%i!<$HOH$N?'$,F1$8(J &          &     \\
      & $@>l9g$K$O!$(JBackground$@%+%i!<$rOH$N?'$K$9$k!%(J             &          &     \\
      & $@!|(J EmergensySave () $@$r8F$P$:$K(J exit($-$1)$@$7$F$$$k(J        & 30/03/91 & $@!{(J  \\
      & $@"*3F=j(J                                                 &          & 1/4 \\
      & EmergensySave()$@$r%3!<%k$9$k$h$&$K$9$k!JCfED(J@CS$@8&$5$s!K(J &          &     \\
\hline
\end{tabular}\\
\medskip
\medskip
\medskip
\medskip
\medskip
\medskip
\medskip
\medskip
\medskip
\medskip
\newpage

\section{tgif+ $@$N%P%0(J}
\label{bug}
\medskip
\par
tgif+ $@$r;HMQ$9$k:]$K$O!$<!$NE@$KCm0U$7$F2<$5$$!%(J
\medskip
\par
Undo $@5!G=(J
\begin{nquote}{4ex}
\begin{itemize}
\item $@%5%$%:$,(J 0 $@$K$J$k$^$G(J $@%j%5%$%:$5$l$?(J $@%*%V%8%'%/%H$O!$$b$H$K(J
$@La$i$J$$>l9g$,$"$j$^$9!%(J
\medskip
\end{itemize}
\end{nquote}
\par
$@%;%s%A%a!<%?(J $@C10L7O(J
\begin{nquote}{4ex}
\begin{itemize}
\item $@%;%s%A%a!<%?C10L7O$O!$(JZoomIn$@!$(JZoomOut $@;~$KL\@9$jI=<($,$*$+$7(J
$@$/$J$j$^$9!%?^$K$O1F6A$O$"$j$^$;$s!%(J
\medskip
\end{itemize}
\end{nquote}
\par
Fake $@%U%)%s%H5!G=(J
\begin{nquote}{4ex}
\begin{itemize}
\item Fake $@%U%)%s%H$O!$(Jk14 $@$h$j$b%5%$%:$,>.$5$$;~$K$O!$%4%_$rI=<((J
$@$7$^$9$,LdBj$O$"$j$^$;$s!%$^$?!$2s$j$KI=<($5$l$kJ8;zOH$O!$<B:]$N(J
$@J8;zNs$N%5%$%:!J(JBounding Box$@!K$h$j$b<c43Bg$-$a$G$9!%(J
\medskip
\end{itemize}
\end{nquote}
\par
wired down $@5!G=(J
\begin{nquote}{4ex}
\begin{itemize}
\item $@%a%$%s%a%K%e!<$r2hLL$KD%$jIU$1$?>uBV$G(J tgif+ $@$r(J lower $@$9$k$H(J
$@%&%#%s%I%&%^%M!<%8%c$K$h$C$F$O!$%a%$%s%a%K%e!<$N%&%#%s%I%&$r(J
lower $@$7$J$$>l9g$,$"$j$^$9!%$=$N$h$&$J%&%#%s%I%&%^%M!<%8%c$N$b$H$G(J
$@;HMQ$9$k>l9g$O!$(Jtgif+ $@$N%&%#%s%I%&$,<YKb$J;~$O!$(Jlower $@$9$k$+$o$j$K(J
$@%"%$%3%s2=$7$F2<$5$$!%(J
\medskip
\end{itemize}
\end{nquote}
\par
ZoomIn $@5!G=(J
\begin{nquote}{4ex}
\begin{itemize}
\item $@J8;z$NBg$-$5$O!$Bg$-$/$J$j$^$;$s$,!$$=$N$+$o$j!$$A$g$C$HBg$-(J
$@L\$NJ8;zOH$,$D$-$^$9!J$A$g$C$HJ8;z$N:FIA2h$K%P%0$,$"$C$?$j$7$^(J
$@$9$,!$L@<(E*$K(J redraw $@$7$F$d$k$HLa$j$^$9!K!%(J
\medskip
\end{itemize}
\end{nquote}
\par
$@%+%i!<(J
\begin{nquote}{4ex}
\begin{itemize}
\item $@KM$N;H$C$F$$$k%^%7%s$O!$%+%i!<$G$O$J$$$N$G$$$/$D$+%P%0$,$"$k$H(J
$@;W$$$^$9!%(J
\medskip
\end{itemize}
\end{nquote}
\vspace*{2cm}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/mado.eps,hscale=0.6,vscale=0.6}
        \caption{$@$&$aAp(J}
   \end{center}
\end{figure}
\medskip
\newpage

\section{$@L>A0(J}
\medskip
\begin{nquote}{8ex}
\par
tgif+ $-$ X $@%&%#%s%I%&>e$N:n?^%D!<%k(J tgif $@$rF|K\8l2=(J + $@&A(J $@$7(J
$@$?$b$N!J0J2<!$(Jtgif $@8GM-$N5!G=$N@bL@$r$9$k>l9g$K$b(J tgif+ $@$H$7$^$9!K(J
\end{nquote}
\medskip

\section{$@7A<0(J}
\label{synopsis}
\begin{nquote}{8ex}
\par
tgif+ [$@%*%W%7%g%s(J ...]  [fileName]
\end{nquote}
\medskip
\medskip

\section{$@5!G=35MW(J}
\label{facility}
\medskip
\par
tgif+ $@$O!$(JX$@%&%$%s%I%&>e$G!$2qOCE*$K?^7A$N:n@.!&A`:n$r9T$J$&:n?^%D!<%k(J
$@$G!$(J PostScript(TM)  $@%W%j%s%?$KD>@\=PNO$7$?$j!$(JLaTeX $@$G=q$+$l$?J8=qFb$K<h$j9~(J
$@$`$3$H$,2DG=$J7A<0$N(J  PostScript  $@%U%!%$%k!J(Jencapsulated PostScript $@!K$r@8(J
$@@.$7$?$j$9$k$3$H$,$G$-$^$9!J%+%i!<$b%5%]!<%H$7$F$$$^$9!K!%(J
\medskip
\par
$@IA2h%W%j%_%F%#%V$H$7$F$O!$%F%-%9%H!$;M3Q7A!$BJ1_!$%]%j%i%$%s!$%]%j%4%s!$BJ1_8L!$(J
$@3Q4];M3Q7A!$%S%C%H%^%C%W$,$"$j$^$9!J(Jxwd $@$N7k2L$O!$%S%C%H%^%C%W$KJQ49$9$k$3$H$G!$?^Cf$K<h$j9~$a$^$9!K!%(J
\medskip
\par
tgif+ $@$GIA$$$??^$O!$%U%!%$%kL>$K!V(J.obj $@!W$H$$$&3HD%;R$,IU$$$?%U%!%$%k(J
$@!J(J.obj $@%U%!%$%k!K$KJ]B8$5$l$^$9!%(J tgif+ $@5/F0$N:]$K;XDj$9$k(J
$@%U%!%$%kL>!J>e5-!V(J\S~\ref{synopsis}~$@7A<0!W$G$N!$!V(JfileName$@!W!K$K$O!$(J.obj $@%U%!%$%k(J
$@$N%U%!%$%kL>$+$i3HD%;R$r>JN,$7$?%U%!%$%kL>$r;XDj$7$^$9!%(J

  tgif+ $@$O!$(J3$@$D%\%?%s$N%^%&%9$r2>Dj$7$?@_7W$K$J$C$F$$$^$9!J!V(J\S~\ref{mouseButton}~$@%^%&%9%\%?%s$N0UL#!W;2>H!K!%(J
\medskip
\par
$@F|K\8lF~NO$O!$(Jnemacs $@$+$i$N%3%T!<(J\&$@%Z!<%9%H$+(J kinput $@$rMQ$$$F9T$&$3$H$,(J
$@$G$-$^$9!%$b$7$/$O!$F|K\8l$,F~NO$G$-$k$J$s$i$+$N%(%G%#%?$,$"$l$P?^$KF|K\8l$r(J
$@F~$l$k$3$H$,$G$-$^$9!J!V(J\S~\ref{japanese}~$@F|K\8l!W;2>H!K!%(J
\medskip
\par
$@%U%)%s%H$OL@D+BN$H%4%7%C%/BN$r%5%]!<%H$7$F$*$j!$%5%$%:$O!V(J
8$@!$(J10$@!$(J12$@!$(J14$@!$(J18$@!$(J24$@!W$G$9!%2hLL$NI=<($K$O!$:GDc(J k14 $@$,$"$l$P2DG=$G$9!%(J
\medskip
\par
$@%=!<%9%3!<%I$O!$$9$Y$F(J X $@%i%$%V%i%j$@$1$G=q$+$l$F$$$^$9!%(J
\medskip

$@!V(J\S~\ref{synopsis}~$@7A<0!W$N%*%W%7%g%s$K$D$$$F$O!$!V(J\S~\ref{option}~$@%*%W%7%g%s!W;2>H(J.
\medskip
\medskip

\section{$@F0:n4D6-(J}
\label{environment}
\medskip
\medskip
tgif+ $@$N<B9T4D6-$O!$;HMQ$9$k5!G=$K1~$8$F0J2<$N$b$N$,I,MW$G$9(J
$@!J!VIUO?(J~\ref{install}~tgif+ $@$N%$%s%9%H!<%k4D6-!$%$%s%9%H!<%kJ}K!!W$b;2>H$7$F2<$5$$!K!%(J
\medskip
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item tgif+ $@$r5/F0$9$k!%(J
\medskip
\begin{description}
\item[$@"*(J]X11R4 $@0J>e$N(JX$@%&%#%s%I%&%7%9%F%`!J<B:]$K$O!$(JX11R3 $@$G$b2DG=!K(J
\medskip
\end{description}
\item tgif+ $@$KF|K\8l$rF~NO$9$k!%(J
\medskip
\begin{description}
\item[$@"*(J]kinput $@!$$^$?$O(J nemacs$@!$$^$?$OF|K\8l$,F~NO$G$-$k$J$s$i$+$N%(%G%#%?(J
\medskip
\end{description}
\item tgif+ $@$K(Jxwd $@$N7k2L$r<h$j9~$`!%(J
\medskip
\begin{description}
\item[$@"*(J]xwd $@%U%)!<%^%C%H$N%U%!%$%k$r(JX11 $@$N%S%C%H%^%C%W7A<0$KJQ49$9$k%U%#%k%?!JNc$($P!$(Jpbmplus $@$N(J xwdtopnm $@$H(J pbmtoxbm $@Ey!K(J
\medskip
\end{description}
\end{enumerate}
}
\par
kinput $@$O!$(JX11R4 $@$N%G%#%9%H%j%S%e!<%7%g%s%F!<%W$KIU?o$7$F$*$j!$(Jpbmplus
$@$O!$(Jexpo.lcs.mit.edu $@$N(J contrib/pbmplus.tar.Z $@$+$iF~<j$9$k$3$H$,$G$-$k(J
$@%U%j!<%&%'%"$G$9(J\footnote{$@EvK\It!$$*$h$S(Jnecspl $@$+$i(J ftp $@2DG=$JJ}$O!$(J/usr/db/pub/expo-contrib/pbmplus.tar.Z $@$+$iF~<j$9$k$3$H$,2DG=$G$9!%(J}$@!%(J
\medskip
\newpage

\section{$@5/F0!$:n?^!$J]B8!$0u:~!$=*N;(J}
\label{howToDo}
\medskip
\par
$@$3$3$G$O!$(Jtgif+ $@$r;H$C$F4JC1$J?^$r:n@.$7!$0u:~$7$F$_$^$9!%(J
\medskip
        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/figure.eps}
                \caption{$@%5%s%W%k(J}
                \label{figure}
           \end{center}
        \end{figure}

$@?^(J~\ref{figure}~$@$r<B:]$K:n@.$7$F$_$^$9!%(J
\medskip

\subsection{$@:GDc8B$N(J .Xdefaults $@$N@_Dj(J}
\medskip
\par
$@$^$:!$(J\verb+~+/.Xdefaults $@%U%!%$%k$K!$(Jtgif+ $@$G:n@.$7$??^$r%W%j%s%H%"%&%H$9$k$?$a$N%3%^%s%I$r;XDj(J
$@$7$^$9(J($@$3$3$G$O!$!V(Jnps1$@!W$H$$$&%W%j%s%?$K=PNO$9$k$h$&$K$7$F$$$^$9!%%W(J
$@%j%s%?L>$K$O(JPostScript $@%W%j%s%?$r=q$$$F2<$5$$!%0J2<!$!V(Jnps1$@!W$O(J
PostScript$@%W%j%s%?$H$7$^$9(J)
\footnote{CISC $@HG(J EWS 4800 $@$G(JX$@%&%#%s%I%&$r$*;H$$$NJ}$O!$!VIUO?(J~\ref{dXdefaults}~.Xdefaults $@$NNc!W$b;2>H$7$F2<$5$$!%(J}$@!%(J
\medskip
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% cat >> ~/.Xdefaults
Tgif*PrintCommand:      lpr -Pnps1
\end{verbatim}
\end{nquote}
\medskip

\subsection{$@5/F0(J}
\medskip
\par
$@@_Dj$7=*$o$C$?$i(J tgif+ $@$r<B9T$7$^$9!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% tgif+
\end{verbatim}
\end{nquote}
\medskip
$@5/F0$9$k$H?^(J \ref{LookAndFeel}~$@$N$h$&$J%&%#%s%I%&$,3+$+$l$^$9!%(J

        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/Look+Feel.eps}
                \caption{$@8+$?L\$H%5%V%&%#%s%I%&$NL>A0(J}
                \label{LookAndFeel}
           \end{center}
        \end{figure}
\medskip
\par
tgif+ $@$N%5%V%&%#%s%I%&$K$D$$$F!$0J2<$G@bL@$7$^$9!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item Message Window
\medskip
\par
tgif+ $@$+$i$N%a%C%;!<%8$,I=<($5$l$k%&%#%s%I%&$G$9!%%(%i!<%a%C%;!<%8Ey$b(J
$@I=<($5$l$^$9!%I=<($5$l$?%a%C%;!<%8$O!$!V:8%\%?%s!W$G%9%/%m!<%k%"%C%W!$!V1&%\%?%s!W$G%9%/%m!<%k%@%&%s$7$^$9!%(J
\medskip
\item Choice Window
\medskip
\par
$@8=:_$NIA2h%b!<%I$HIA2hB0@-!J@~$NI}!$%U%)%s%H!$?'!$EI$j$D$V$7$N(J
$@%Q%?!<%s!$%Z%s%Q%?!<%s!$@~$NC<E@7A>u!$GK@~%Q%?!<%sEy!K$rI=<($7$^$9(J
        $@!J?^(J~\ref{choiceWin}~$@;2>H!K!%(J

        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/choiceWin.eps}
                \caption{Choice $@%&%#%s%I%&(J}
                \label{choiceWin}
           \end{center}
        \end{figure}
\medskip
\item Canvas Window
\medskip
\par
$@<B:]$KIA2h$r9T$J$&%&%#%s%I%&$G$9!%(J
\medskip
\item Scrollbar
\medskip
\par
$@!V(JCanvas $@%&%$%s%I%&!W$NFbMF$r%9%/%m!<%k$7$^$9!%(J
\medskip
\item Ruler
\medskip
\par
$@L\@9$j$H8=:_$N%0%j%C%I$N4V3V$rI=<($7$^$9!%(J
\medskip
\end{enumerate}
}

\subsection{$@:n?^(J}
\medskip

\subsubsection{$@%a%$%s%a%K%e!<$NI=<((J}
\label{displayMainMenu}

$@$^$:!$?^$rIA$-;O$a$kA0$K!$%a%$%s%a%K%e!<!J?^(J~\ref{mainMenu}~$@;2>H!K$N%](J
$@%C%W%"%C%W$N;EJ}$r3P$($F2<$5$$!%(J
\medskip
\par
$@!V1&%\%?%s!W$r!V(JCanvas$@%&%#%s%I%&!W$NCf$G!$2!$7B3$1$F2<$5$$!%(J
$@?^(J \ref{mainMenu} $@$N$h$&$J%&%#%s%I%&$,I=<($5$l$^$9!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/mainMenuW.eps}
        \caption{$@%a%$%s%a%K%e!<(J}
        \label{mainMenu}
   \end{center}
\end{figure}
\medskip
\par
$@!V%a%$%s%a%K%e!<!W$+$i%"%$%F%`!J(JMode$@!$(JFile$@!$(JEdit$@!%!%!%!K$rA*Br$9$k$3$H(J
$@$G!$(Jtgif+ $@$N$9$Y$F$N5!G=$r;HMQ$9$k$3$H$,$G$-$^$9!%(J
\medskip
\par
$@%a%$%s%a%K%e!<$O!$NI$/;H$&%a%K%e!<$J$N$G!$0J2<$NJ}K!$G2hLL$KD%$jIU$1$FCV$/$3$H$K$7(J
$@$^$9!J?^(J~\ref{wiredDown}~$@;2>H!K!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@%a%$%s%a%K%e!<$rI=<($7$?>uBV$G!$%^%&%9$r1&>e$KF0$+$7$F$$$C$F2<$5$$!%(J
\item $@$7$P$i$/F0$+$9$HOH$,H4$1=P$7$^$9!%(J
\item $@$=$N8e$G!$!V1&%\%?%s!W$rN%$9$H%a%$%s%a%K%e!<$,D%$jIU$-$^$9!%(J
\item $@0\F0$5$;$k;~$O!V1&%\%?%s!W$G%I%i%C%0!$>C$9;~$O!V??Cf%\%?%s!W$G%/%j%C%/$G$9!%(J
\medskip
\end{enumerate}
}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/wiredDown.eps}
        \caption{$@%a%$%s%a%K%e!<$ND%$jIU$1(J}
        \label{wiredDown}
   \end{center}
\end{figure}
\medskip

\subsubsection{tgif+ $@$G$N%a%K%e!<$NA*Br$N;EJ}(J}
\medskip
\par
tgif+ $@$G$N%a%K%e!<$NA*Br$O!$%^%&%9$G9T$$$^$9!%A*Br$N;EJ}$O!$A4$F$N%a%K(J
$@%e!<$GF1$8$G$9!%%a%K%e!<$rA*Br$9$k$H!$A*Br$5$l$?%"%$%F%`$,<B9T$5$l$^$9!%(J
$@Nc$($P!$(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V%+!<%=%k!W$r!V(JEdit$@!W$H$$$&J8;z$N>e$K;}$C$F$$$-$^$9!%(J
\item $@!V:8%\%?%s!W$r2!$7B3$1$^$9!%(J
\item $@!V(JEdit $@%a%K%e!<!W$,I=<($5$l$^$9(J
$@!J?^(J~\ref{editMenuWin}~$@;2>H!K!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/editMenuW.eps}
        \caption{Edit $@%a%K%e!<(J $@$NI=<(Iw7J!J:8!K$H$=$NFbMF!J1&!K(J}
        \label{editMenuWin}
   \end{center}
\end{figure}
\medskip
\item $@!V:8%\%?%s!W$r2!$7$?$^$^!$!V(JEdit$@!W%a%K%e!<$N>e$r$J$>$k$H!$%"%$%F(J
$@%`$,<!!9$HH?E>I=<($7$^$9!%(J
\item $@Nc$($P!$!V(JSelectAll$@!W$N>e$G!$%\%?%s$rN%$9$H!V%*%V%8%'%/%H$r$9$Y$FA*Br$9$k!W$,<B9T$5$l$^$9!%(J
\medskip
\end{enumerate}
}
\par
$@$3$N$h$&$K!$%^%&%9%\%?%s$r2!$7$J$,$i%^%&%9$rF0$+$7!$L\E*$N%a%K%e!<$N%"(J
$@%$%F%`$N$H$3$m$G!$%\%?%s$rN%$9$3$H$r!$!V%a%K%e!<$rA*Br$9$k!W$H$$$$$^$9(J
$@!J?^(J~\ref{selectMenu} $@;2>H!K!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/selectMnu.eps}
        \caption{$@%a%K%e!<$rA*Br$9$k(J}
        \label{selectMenu}
   \end{center}
\end{figure}

$@$3$NNc$N>l9g$O!$!V(JEdit $@%a%K%e!<$+$i(J SelectAll $@$rA*Br$9$k!W$H$J$j$^$9(J
\footnote{$@$3$N>l9g$O!$$^$@(J tgif+ $@$G2?$bIA$$$F$$$J$$$N$G!$<B:]$K$O!$2?$b5/$3$j$^$;$s(J
$@$N$G!$?4G[$7$J$$$G2<$5$$!J!V(J\S~\ref{selection}~$@A*Br!W;2>H!K!%(J}$@!%(J
\medskip

\subsubsection{$@;M3Q7A$rIA$/!%(J}
\medskip
$@;M3Q7A$rIA$$$F$_$^$9!J?^(J~\ref{drawRect} $@;2>H!K!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JMode$@%a%K%e!<!W$+$i!V;M3Q7A!W$rA*Br$7$F2<$5$$(J
$@!J?^(J~\ref{modeMenuWin}~$@;2>H!K(J\footnote{$@$J$*!$8=:_$N!VIA2h%b!<%I!W$O!$!V(JChoice $@%&%#%s%I%&!W!J?^(J~\ref{choiceWin}$@!K$KI=<($5$l$^$9!%(J}$@!%(J
\item $@%+!<%=%k$,Gr$/$J$j$^$9!%(J
\item $@!V(JCanvas $@%&%#%s%I%&!W$N:8>e$NE,Ev$J>l=j$G!$!V:8%\%?%s!W$r2!$7%I%i%C%0$7$F2<$5$$!%(J
\item $@;M3Q$$%i%P!<%P%s%I$,I=<($5$l$^$9(J
\footnote{$@$3$N>l9g!$%i%P!<%P%s%I$O$"$k0lDj$N4V3V!J%0%j%C%I4V3V!K$G$7$+!$(J
$@Bg$-$/$9$k$3$H$,$G$-$J$$$H;W$$$^$9!%$3$N%0%j%C%I$r$&$^$/;H$&$H$-$l$$$J?^$,(J
$@AGAa$/IA$1$^$9!%%0%j%C%I4V3V$NJQ99$O!$!V(J\S~\ref{xyGrid}
~$@%0%j%C%I4V3V$rJQ$($?$$!W;2>H!%(J}$@!%(J
\item $@E,Ev$J>l=j$G!$%\%?%s$rN%$9$H!$;M3Q7A$,IA$1$^$9!%(J
\medskip
\end{enumerate}
}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/modeMenuW.eps}
        \caption{Mode $@%a%K%e!<(J}
        \label{modeMenuWin}
   \end{center}
\end{figure}

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/drawRect.eps}
        \caption{$@;M3Q7A$rIA$/(J}
        \label{drawRect}
   \end{center}
\end{figure}
\medskip
\par
$@$J$*!$0lEYIA2h%b!<%I$K$J$k$H!$!V(JMode $@%a%K%e!<!W$+$i!VLp0u!W$rA*Br$7$F(J
$@!VA`:n%b!<%I!W$K$9$k$^$G$O!J$D$^$j!$%+!<%=%k$,!VGr$$!W$&$A$O!K!$(J
$@F1$8?^7A$,B3$1$FIA2h$G$-$^$9!%A`:n%b!<%I$K$J$k$H%+!<%=%k$O!$!V9u$/!W$J$j$^$9(J
\footnote{$@0l;~E*$K!$!VA`:n%b!<%I!W$K$7$?$$>l9g$O!$!V(J \^{}E$@!W!J%3%s%H%m!<%k$r(J
$@2!$7$J$,$i!V(JE$@!W$r2!$9!%(JE $@$O>.J8;z!K$H%-!<%\!<%I$+$i(J
$@F~NO$7$F2<$5$$!%$b$&0lEY!V(J \^{}E$@!W$rF~NO$9$k$^$G$O!$%+!<%=%k$,!V9u$/!W$J$C$F(J
$@!VA`:n%b!<%I!W$K$J$j$^$9!%:FEY!V(J \^{}E$@!W$rF~NO$9$k$H!$85$N!VIA2h%b!<%I!W$K(J
$@La$j$^$9(J.}$@!%(J
\medskip

\subsubsection{$@1_$rIA$/!%(J}
\medskip
\par
$@!V(JMode $@%a%K%e!<!W$+$i!V1_!W$rA*Br$7$F2<$5$$!%(J
$@8e$O!$;M3Q7A$HF1$8$G$9!J?^(J~\ref{drawRect}~$@;2>H!K!%(J
\medskip
\medskip

\subsubsection{$@J8;z$rF~$l$k!%(J}
\medskip
$@J8;zNs$rF~$l$F$_$^$9!J?^(J~\ref{drawText}~$@;2>H!K!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JMode $@%a%K%e!<!W$+$i!V(JT$@!W$rA*Br$7$F2<$5$$!%(J
\item $@%+!<%=%k$,!V(JI$@!W$N;z$K$J$j$^$9!%(J
\item $@J8;z$rF~NO$7$?$$>l=j$G!$!V:8%\%?%s!W$r%/%j%C%/$7$F2<$5$$!%(J
\item $@!V(JEnder's Game!$@!W$H%-!<%\!<%I$+$iF~NO$7$F2<$5$$!%(J
\item $@4V0c$($?>l9g$O!$%P%C%/%9%Z!<%9$G>C$($^$9!%(J
\item $@%-%c%m%C%H$N0\F0$O%^%&%9$G9T$$$^$9!%(J
\medskip
\end{enumerate}
}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/drawText.eps}
        \caption{$@J8;z$rIA$/(J}
        \label{drawText}
   \end{center}
\end{figure}
\medskip

\subsubsection{$@1_$N@~$rB@$/$7!$Cf$rEI$k!%(J}
\medskip
$@$3$N$h$&$JA`:n$O!$?^7A$NB0@-$rJQ99$9$k$3$H$G9T$J$$$^$9!J(J
$@?^7A$NB0@-$O!$!V(J\S~\ref{property}~$@IA2hB0@-!W$r;2>H$7$F2<$5$$!K!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@$^$:!$!V(JMode $@%a%K%e!<!W$N!VLp0u!W$rA*Br$7$F2<$5$$!%(J
\item $@%+!<%=%k$,9u$/$J$j$^$9!%(J
\item $@%+!<%=%k$r1_<~$N>e$K;}$C$F$$$C$F!V:8%\%?%s!W$r%/%j%C%/$7$F2<$5$$!%(J
\item $@1_$N2s$j$K!$!V>.$5$J;M3Q!W$,I=<($5$l$^$9!J$3$l$r!V?^7A$rA*Br$9$k!W$H8@$$$^$9!K!%(J
$@$5$l$J$$>l9g$O!$A@$$J}$,0-$$$N$G$b$&0l2s$d$C$F$_$F2<$5$$(J:-P$@!J!V(J\S~\ref{selection}~$@A*Br!W;2>H!K!%(J
\item $@$^$:!$@~$NB@$5$rJQ$($^$9!%(J
\item $@!V(JLineStyle$@%a%K%e!<!W$+$iE,Ev$JB@$5$N@~$rA*Br$7$F2<$5$$!%1_$N(J
$@@~$NB@$5$,JQ$o$C$?$O$:$G$9!J?^(J~\ref{chOline} $@;2>H!K(J
\footnote{$@$3$N>l9g!$!V(JLineStyle $@%a%K%e!<!W$N%"%$%F%`$N$$$/$D$+$,H?E>I=<($5$l$F$$$k(J
$@$H;W$$$^$9!%$3$l$O!$$=$N%*%V%8%'%/%H!J1_!K$N!V(JLineStyle $@%a%K%e!<!W$K4X$9$k%+%l%s%H$N>uBV$rH?E>I=<($7$F$$$^$9!%(J}$@!%(J
\item $@<!$KCf$rEI$j$D$V$7$^$9!%(J
\item $@!V(JFill$@%a%K%e!<!W$+$iE,Ev$J%Q%?!<%s$rA*Br$7$F2<$5$$!%1_$NCf$,(J
\end{enumerate}
}
$@EI$j$D$V$5$l$?$O$:$G$9(J($@?^(J~\ref{chOfill}~ $@;2>H!K!%(J
\medskip
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/chOline.eps}
        \caption{$@1_$N@~$rB@$/$9$k(J}
        \label{chOline}
   \end{center}
\end{figure}

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/chOfill.eps}
        \caption{$@1_$NCf$r$L$k(J}
        \label{chOfill}
   \end{center}
\end{figure}
\medskip

\subsubsection{$@Lp0u$rIA$/!%(J}
\medskip
$@Lp0u$rIA$$$F$_$^$9!J?^(J~\ref{drawLine} $@;2>H!K!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JMode $@%a%K%e!<!W$N!V%]%j%i%$%s!W$rA*Br$7$F2<$5$$!%(J
\item $@%+!<%=%k$,Gr$/$J$j$^$9(J
\item $@!V(JCanvas$@%&%#%s%I%&!W$NE,Ev$J>l=j$r!V:8%\%?%s!W$G%/%j%C%/$7$F2<$5$$!%(J
\item $@%^%&%9$rF0$+$9$H@~$,0z$1$^$9!%(J
\item $@!V:8%\%?%s!W$r%/%j%C%/$7$F2<$5$$!%@^$l@~$K$J$j$^$9!%(J
\item $@%^%&%9$rF0$+$9$H@~$,0z$1$^$9!%(J
\item $@!V??Cf%\%?%s!W$r%/%j%C%/$9$k$H@~$,3NDj$7$^$9!%(J
\medskip
\end{enumerate}
}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/drawLine.eps,hscale=0.9,vscale=0.9}
        \caption{$@Lp0u$rIA$/(J}
        \label{drawLine}
   \end{center}
\end{figure}
\medskip

\subsubsection{$@Lp0u$N@~$rB@$/$7!$%Z%s%Q%?!<%s$rJQ$($k!%(J}
\medskip
\par
$@Lp0u$N@~$rB@$/$9$k$N$O!$1_$N@~$rB@$/$9$k$N$HF1$8$G$9!%(J
\medskip
\par
$@$^$?!$%Z%s%Q%?!<%s$rJQ$($k$N$O!$!V(JPen $@%a%K%e!<!W$r;H$&0J30$O!$1_$rEI$k$N$H(J
$@F1$8$G$9(J
$@!J?^(J~\ref{chLprop}~$@;2>H!K!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/chLprop.eps,hscale=0.9,vscale=0.9}
        \caption{$@Lp0u$rB@$/$7!$EI$k(J}
        \label{chLprop}
   \end{center}
\end{figure}
\medskip
\medskip

\subsubsection{$@;M3Q7A$r%j%5%$%:$9$k!%(J}
\medskip
$@$A$g$C$H;M3Q7A$rBg$-$/$7$F$_$^$7$g$&(J($@?^(J~\ref{resizeRect}~$@;2>H!K!%(J
$@1_$N;~$N$h$&$K!$;M3Q7A$rA*Br$7$^$9!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JMode $@%a%K%e!<!W$N!VLp0u!W$rA*Br$7$F2<$5$$!%(J
\item $@;M3Q7A$NJU$N>e$G!$!V:8%\%?%s!W$r%/%j%C%/$7$F$/$@$5$$!%(J
\item $@;M3Q7A$N2s$j$K!$!V>.$5$J;M3Q!W$,I=<($5$l$^$9!%(J
\item $@!V>.$5$J;M3Q!W$N0l$D$N>e$K%+!<%=%k$r;}$C$F$$$-!$!V:8%\%?%s!W$r2!$7$F2<$5$$!%(J
\item $@!V:8%\%?%s!W$r2!$7$?$^$^!$%^%&%9$r%I%i%C%0$9$k$HBg$-$5$,JQ$o$j$^$9!%(J
\item $@%\%?%s$rN%$9$H!$$=$NBg$-$5$K$J$j$^$9!%(J
\medskip
\end{enumerate}
}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/resizeRec.eps,hscale=0.9,vscale=0.9}
        \caption{$@%*%V%8%'%/%H$r%j%5%$%:$9$k(J}
        \label{resizeRect}
   \end{center}
\end{figure}
\medskip
\par
$@%*%V%8%'%/%H$N%j%5%$%:$N;EJ}$O!$B>$N$b$N$bF1$8$G$9(J
\footnote{$@J8;zNs$O%j%5%$%:$G$-$^$;$s!%Bg$-$/$7$?$$$H$-$O!$%F%-%9%H%5%$%:(J
$@$rJQ$($F2<$5$$!JJ8;zNs$rA*Br$7$F!$!V(JTextSize $@%a%K%e!<!W$GE,Ev$J%5%$%:$rA*Br$9$k!K!%$^$?!$%S%C%H%^%C%W$b%j%5%$%:$G$-$^$;$s!%(J
$@$J$*!$%]%j%4%s$H%]%j%i%$%s$O!$Bg$-$5$G$O$J$/7A>u$,JQ$o$j$^$9!%%j%5%$%:$9$k(J
$@>l9g$K$O!$2?$+B>$N%*%V%8%'%/%H$H%0%k!<%W2=!J!V(J\S~\ref{group}$@!W;2>H!K$7$F$+$i%j%5%$%:$7$F2<$5$$!%(J}$@!%(J
\medskip

\subsubsection{$@1_$r0\F0$9$k!%(J}
\medskip
$@1_$N>l=j$rJQ$($F$_$^$7$g$&!J?^(J~\ref{moveOval}~$@;2>H!K!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@1_$rA*Br$7$^$9!%(J
\item $@!V>.$5$J;M3Q0J30!W$NItJ,$K%+!<%=%k$r;}$C$F$$$-!$!V:8%\%?%s!W$r2!$7$J(J
$@$,$i%I%i%C%0$7$F2<$5$$!J!V>.$5$J;M3Q!W$rA*Br$9$k$H%j%5%$%:$5$l$F$7$^$$$^$9!K!%(J
\item $@%\%?%s$rN%$9$H$=$N>l=j$K0\F0$7$^$9!%(J
\medskip
\end{enumerate}
}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/moveOval.eps}
        \caption{$@%*%V%8%'%/%H$r0\F0$9$k(J}
        \label{moveOval}
   \end{center}
\end{figure}
\medskip
\par
$@%*%V%8%'%/%H$N0\F0$N;EJ}$O!$B>$N$b$N$bF1$8$G$9!%(J
\medskip

\subsubsection{$@J8;z$N%U%)%s%H$rJQ$($k!%(J}
\label{changeFontChar}

$@J8;z$N%U%)%s%H$rJQ$($F$_$^$9!J?^(J~\ref{changeFont}~$@;2>H!K!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@J8;zNs$rA*Br$7$^$9!%(J
\item $@!V(JFont $@%a%K%e!<!W$+$i!V(JTimes$@!W$rA*$S$^$9!%(J
\medskip
\end{enumerate}
}
\par
$@$J$*!$%U%)%s%H$N%9%?%$%k!J%m!<%^%sBN!$%\!<%k%IBN!$%$%?%j%C%/BN!%!%!%!K$rJQ$($k$K$O!$(J
$@!V(JTextStyle $@%a%K%e!<!W$r;H$C$F2<$5$$!%(J
\medskip
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/chFont.eps,hscale=0.9,vscale=0.9}
        \caption{$@J8;z$N%U%)%s%H$rJQ$($k(J}
        \label{changeFont}
   \end{center}
\end{figure}
\medskip

\subsubsection{$@$=$NB>(J}
\medskip
$@?^(J~\ref{figure} $@$r:n@.$9$k:n6H$K$OD>@\4X78$7$^$;$s$,!$$$$/$D$+CN$C$F$*$$$?J}$,(J
$@NI$$$b$N$K$D$$$F=q$$$F$*$-$^$9!%(J
\medskip
\begin{itemize}
\item $@!V(JLineStyle$@%a%K%e!<!W$G!$Lp0u$r!V%9%W%i%$%s2=!W$7!$!VLp0u$rD>@~!W$K$7$F$_$^$9(J
$@!J?^(J~\ref{chLspline} $@;2>H!K!%(J

        $@%9%W%i%$%s2=$O!$%]%j%4%s$KBP$7$F$b2DG=$G$9!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/chLspline.eps,hscale=0.9,vscale=0.9}
        \caption{$@%9%W%i%$%s2=$HLp0u$ND>@~2=(J}
        \label{chLspline}
   \end{center}
\end{figure}
\medskip
\item $@%G%U%)%k%H$NJQ99(J
\medskip
\end{itemize}
\par
$@IA2h$r9T$J$C$?;~$K!$$I$N$h$&$JIA2hB0@-$r;}$D%*%V%8%'%/%H$,IA2h$5$l$k$+$O!$(J
$@!V(JChoice $@%&%#%s%I%&!W!J?^(J~\ref{choiceWin} $@;2>H!K$K<($5$l$F$$$^$9!%(J
$@Nc$($P!$?^(J~\ref{choiceWin} $@$N>uBV$G@~$r0z$/$H$$$D$b!$(J
\begin{nquote}{4ex}
{
\renewcommand{\theenumi}{(\alph{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V@~$NC<E@7A>u!W$O!VLp0u!W(J
\item $@!V%Z%s%Q%?!<%s!W$O!V$Y$?EI$j!W(J
\item $@!V@~$NI}!W$O!V0lHV:Y$$!W(J
\medskip
\end{enumerate}
}
\end{nquote}
\par
$@$H$$$&B0@-$N@~$,0z$+$l$^$9!%(J
\medskip
\par
$@$3$N%G%U%)%k%H$rJQ$($?$$>l9g$K$O!$$D$.$NFs$D$NJ}K!$,$"$j$^$9!%(J
\medskip
\begin{nquote}{4ex}
{
\renewcommand{\theenumi}{(\alph{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JChoice$@%&%#%s%I%&!W$N%"%$%3%s$r!V:8%\%?%s!W$G%/%j%C%/$9$k!%(J
\item $@2?$bA*Br$7$F$$$J$$>uBV$G!$%a%K%e!<$+$i@_Dj$7$?$$B0@-$rA*$V!%(J
\medskip
\end{enumerate}
}
\end{nquote}
\par
$@$H$j$"$($:!$!V"*!W$,I=<($5$l$F$$$k!V@~$NC<E@7A>u!W$N%"%$%3%s$r!$!V:8%\%?%s!W(J
$@$G(J4$@2s%/%j%C%/$9$k$H?^(J~\ref{solidCwin}~$@$N$h$&$K$J$j$^$9!$(J
$@$3$N>uBV$G@~$r0z$$$F$_$k$H%G%U%)%k%H$G!VLp0u$J$7$N@~!W$,0z$1$k$h$&$K$J$j$^$9!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/solidCwin.eps}
        \caption{$@%G%U%)%k%H$NJQ99(J}
        \label{solidCwin}
   \end{center}
\end{figure}
\medskip

\subsubsection{$@$^$H$a(J}
\medskip
\begin{itemize}
\item $@IA2h(J
\medskip
\par
$@!V(JMode $@%a%K%e!<!W$+$iIA2h$7$?$$%*%V%8%'%/%H$rA*Br$7$F9T$J$$$^$9!%(J
\medskip
\item $@%*%V%8%'%/%H$NA*Br(J
\medskip
\par
$@!V:8%\%?%s!W$r%/%j%C%/$7$F9T$J$$$^$9!%(J
\medskip
\item $@%*%V%8%'%/%H$KBP$9$kA`:n!JB0@-$rJQ$($k!$0\F0$9$k!$3HBg$9$kEy!K(J
\medskip
\par
$@A`:nBP>]$N%*%V%8%'%/%H$rA*Br$7$F$+$i9T$J$$$^$9!%(J
\medskip
\begin{itemize}
\item $@B0@-$NJQ99(J
\medskip
\par
$@A*Br$7$?%*%V%8%'%/%H$KBP$7$F%a%K%e!<$G9T$J$$$^$9!%(J
\medskip
\end{itemize}
\begin{itemize}
\item $@3HBg!J=L>.!K(J
\medskip
\par
$@A*Br$7$?%*%V%8%'%/%H$N<~$j$N!V>.$5$J;M3Q!W$r!V:8%\%?%s!W$G(J
$@%I%i%C%0$9$k$3$H$G9T$J$$$^$9!%(J
\medskip
\item $@0\F0(J
\medskip
\par
$@A*Br$7$?%*%V%8%'%/%H$r!V:8%\%?%s!W$G!$!V>.$5$J;M3Q0J30!W$NItJ,$r(J
$@%I%i%C%0$9$k$3$H$G9T$J$$$^$9!%(J
\medskip
\medskip
\end{itemize}
\end{itemize}

\subsection{$@J]B8(J}
\medskip
\par
$@:#IA$$$??^$r%U%!%$%k$KJ]B8$7$^$9!%(J
\medskip
$@!V(JFile$@!W%a%K%e!<$+$i!V(JSave$@!W$rA*Br$7$F2<$5$$!%?^(J \ref{filePWin} $@$N$h$&(J
$@$J%&%#%s%I%&$,I=<($5$l!$%U%!%$%kL>$rJ9$$$F$-$^$9!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/filePWin.eps}
        \caption{$@%U%!%$%kL>$r?R$M$F$$$k%&%#%s%I%&(J}
        \label{filePWin}
   \end{center}
\end{figure}

\medskip
\par
$@$=$N%&%#%s%I%&$K%^%&%9%+!<%=%k$rF~$l$F!$%U%!%$%kL>$r!V(Jfigure$@!W$H(J
$@F~$l$F2<$5$$!%%+%l%s%H%G%#%l%/%H%j$K!$!V(Jfigure.obj$@!W$H$$$&%U%!%$%k$,$G(J
$@$-$F$$$k$O$:$G$9!%(J
\medskip

\subsection{$@0u:~(J}
\label{print}
\medskip
\par
$@!V(JFile$@%a%K%e!<!W$+$i!V(JPrint$@!W$rA*Br$7$F2<$5$$!%:G=i$K(J \verb+~+/.Xdefaults $@$G(J
$@;XDj$7$?%3%^%s%I$G%W%j%s%H%"%&%H$5$l$^$9(J\footnote{$@FbItE*$K$O!$(Jfigure.obj
$@$,(J PostScript $@%U%!%$%k$KJQ49$5$l$F!$$=$l$,(J .Xdefaults $@$G@_Dj$7$?(J
$@%3%^%s%I!J(Jlpr -Pnps1$@!K$G%W%j%s%H%"%&%H$5$l$k$3$H$K$J$j$^$9!%(J}$@!%(J

$@$3$3$G$D$$$G$K!$!V(J.eps $@%U%!%$%k!W$b:n@.$7$F$*$-$^$9!J!V(J\S~\ref{files}~tgif+ $@$N@8@.$9$k%U%!%$%k!W;2>H!K!%$3$l$O!$(JjLaTeX $@$G?^$r<h$j9~$`$?$a$N$b$N$G$9(J
$@!J!V(J\S~\ref{howToInclude}~jLaTeX $@$X$N?^$N<h$j9~$_!W;2>H!K!%(J
\medskip
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JChoice $@%&%#%s%I%&!W$N!V%W%j%s%H@h!W%"%$%3%s$r!V:8%\%?%s!W$G(J
$@!V(JLaTeX $@%"%$%3%s!W$K$J$k$^$G%/%j%C%/$7$F2<$5$$!J?^(J~\ref{latexCwin}~$@;2>H!K!%(J
\item $@!V(JFile $@%a%K%e!<!W$+$i!V(JPrint$@!W$rA*Br$7$F2<$5$$!%(J
\item $@%+%l%s%H%G%#%l%/%H%j$K!$!V(Jfigure.eps$@!W$H$$$&%U%!%$%k$,$G$-$F$$$k$O$:$G$9!%(J
\medskip
\end{enumerate}
}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/latexCwin.eps}
        \caption{$@%W%j%s%H%"%&%H@h$NJQ99(J}
        \label{latexCwin}
   \end{center}
\end{figure}

\medskip

\subsection{$@=*N;(J}
\medskip
\par
$@!V(JFile$@%a%K%e!<!W$+$i!V(JQuit$@!W$rA*Br$7$F2<$5$$!%(J
\medskip

\subsection{$@%-!<%P%$%s%I(J}
\medskip
\par
$@0J>e$N@bL@$G$O!$$9$Y$F$NA`:n$r%a%K%e!<$G9T$C$F$_$^$7$?$,!$(Jtgif+ $@$O!$$[$H$s$I(J
$@$N5!G=$,%-!<$K%P%$%s%I$5$l$F$$$^$9!%DL>o%P%$%s%I$5$l$F$$$k%-!<$O!$%a%K(J
$@%e!<$N%"%$%F%`L>$N2#$K<($5$l$F$$$^$9!J>\$7$/$O!$!V(J\S~\ref{menus}~$@%a%K(J
$@%e!<!W;2>H!K!%(J
\medskip
\par
$@Nc$($P!$!V(JEdit$@%a%K%e!<!W$N!V(JRedraw$@!W$O!$!V(J\verb+^+R$@!W$H=q$$$F$"$j$^$9(J
$@!J?^(J~\ref{editMenuWin}~ $@;2>H!K!%(J
\medskip
\par
$@$3$l$O!$(J
$@!V%3%s%H%m!<%k!W$r2!$7$J$,$i(J $@!V(Jr$@!W(J $@$r2!$9$3$H$r0UL#$7$^$9!J%"%k%U%!%Y(J
$@%C%H$O!$BgJ8;z$G=q$+$l$F$$$^$9$,!$$3$l$O8+0W$/$9$k$?$a$G!$<B:]$O>.J8;z(J
$@$G$9!K!%$9$J$o$A!$%-!<%\!<%I$+$i!V(J\verb+^+R$@!W$rF~NO$9$k$3$H$G!$!V(JEdit $@%a%K%e!<!W(J
$@$+$i!V(JRedraw$@!W$rA*Br$7$?$3$H$HF1$8$3$H$K$J$j$^$9!%(J
\medskip
\par
$@$^$?!$!V(J\#$@!W$O!V%a%?%-!<!W$rI=$7$^$9!%!V(J\verb+^+\#$@!W$H=q$+$l$F$$$k>l9g$K$O!V%3(J
$@%s%H%m!<%k!W$H!V%a%?%-!<!W$r!VN>J}0lEY$K2!$7$J$,$i!W$H$$$&0UL#$K$J$j$^(J
$@$9!%(J
\medskip
\par
$@>e$G9T$C$?:n6H$O!$<!$NI=$N$h$&$K%P%$%s%I$5$l$F$$$^$9!%(J
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|}
\hline
$@%U%!%$%k$N%;!<%V(J           & \verb+^+S \\
\hline
$@%U%!%$%k$N0u:~(J             & \verb+^+P \\
\hline
tgif+ $@$N=*N;(J               & \verb+^+Q \\
\hline
$@IA2h%b!<%I$r%F%-%9%H$K$9$k(J & \verb+^+W \\
\hline
$@IA2h%b!<%I$r;M3Q7A$K$9$k(J   & \#R \\
\hline
$@IA2h%b!<%I$r1_$K$9$k(J       & \verb+^+\#V\\
\hline
$@%b!<%I$r!VA*Br!W$K$9$k(J     & \#K \\
\hline
\end{tabular}\\
\end{center}
\medskip
\par
$@%-!<%P%$%s%I$r3P$($F$*$/$H!$(Jtgif+ $@$NA`:n$,B.$/$J$jJXMx$G$9(J
$@!J!VIUO?(J~\ref{keyBinding}~$@%-!<%P%$%s%I!W;2>H!K!%(J
\medskip

\subsection{$@F|K\8l$rF~$l$k$K$O!%!%!%(J}
$@!V(J\S~\ref{japanese}~$@F|K\8l!W$r8+$F2<$5$$!%(J
\medskip

\subsection{$@:$$C$?>l9g$K$O!%!%!%(J}
$@!VIUO?(J~\ref{trouble}~$@%H%i%V%k!W$,$"$j$^$9$N$G!$$=$l$r8+$F2<$5$$!%(J
\medskip
\newpage

\section{jLaTeX $@$X$N?^$N<h$j9~$_(J}
\label{howToInclude}
\medskip
\par
tgif+ $@$G!$IA$$$??^$r(J jLaTeX $@$G=q$$$?J8=q$K<h$j9~$`$K$O!$(J
$@<!$N$h$&$K$7$^$9(J($@>e$G:n@.$7$??^!V(Jfigure$@!W$r<h$j9~$s$G$_$^$9(J)$@!%(J
\medskip
jLaTeX $@$K?^$r<h$j9~$`>l9g$K$O!$!V(J\S~\ref{print}~$@0u:~!W$G:n@.$7$?!$!V(J
figure.eps$@!W%U%!%$%k$rMQ$$$^$9!%(J
\medskip
\par
$@$^$:!$(Jdoc.tex $@$NFbMF$,<!$N$h$&$@$C$?$H$7$^$9!J$H$j$"$($:!$$3$N%U%!%$%k$O!$(J
figure.eps $@$HF1$8%G%#%l%/%H%j$K:n$C$F2<$5$$!K!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% nkf -j doc.tex
        \documentstyle[epsf]{jarticle}
        \begin{document}
        $@$3$NJ8>O$O!$(Jtgif+ $@$GIA$$$??^$,F~$C$F$$$^$9!%(J
        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=figure.eps}
                \caption{tgif+ $@$GIA$$$??^(J}
                \label{figure}
           \end{center}
        \end{figure}
        $@?^(J\ref{figure}$@$,$=$&$G$9!%(J
        \end{document}
\end{verbatim}
\end{nquote}
\medskip
\par
$@$3$3$G!$!V(J$\backslash$epsfile$@!W$H$$$&$N$O!$(Jencapsulated PostScript $@7A<0$N%U%!%$%k$r(JjLaTeX $@$K(J
$@<h$j9~$`$?$a$N%3%^%s%I$G$9!%(J
\medskip
$@>e5-$N%U%!%$%k$r(J jLaTeX $@$K$+$1$F!V(J.dvi$@!W%U%!%$%k$r:n@.$7$^$9(J
\footnote{doc.tex $@$NCf$G!$!V(J$\backslash$ref$@!W$G!$?^$r;2>H$7$F$$$k$N$G(J jLaTeX $@$r(J
2$@2s$+$1$F2<$5$$!%$=$&$7$J$$$H?^$NHV9f$,@5$7$/I=<($5$l$J$$$N$G$9!%(J}$@!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% jlatex doc.tex
\end{verbatim}
\end{nquote}
\medskip
\par
$@%U%!%$%k$K(J tgif+ $@$GIA$$$??^$r<h$j9~$`$?$a$K!$(JPostScript$@$KJQ49$7$^$9!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% dvi2ps doc.dvi > doc.ps
\end{verbatim}
\end{nquote}
\medskip
\par
$@0u:~$9$k$?$a$K!$$3$N%U%!%$%k$r%W%j%s%?$KAw$j$^$9!%%W%j%s%H%"%&%H$K$O!$>/!9(J
$@;~4V$,$+$+$k$+$b$7$l$^$;$s$,!$$9$0$KD|$a$:$KBT$C$F2<$5$$!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% lpr -Pnps1 doc.ps
\end{verbatim}
\end{nquote}
\medskip
\par
$@$^$?!$(Jdvi2ps $@$N7k2L$r$=$N$^$^%W%j%s%?$KAw$k$3$H$b$G$-$^$9!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% dvi2ps doc.dvi | lpr -Pnps1
\end{verbatim}
\end{nquote}
\medskip
\medskip
\par
jLaTeX $@$N%U%!%$%k$K(J tgif+ $@$G:n@.$7$??^$r<h$j9~$`$N;H$&$N$O!$!V(J.obj $@%U%!%$%k!W(J
$@$G$O$J$/!$!V(J.eps $@%U%!%$%k!W$G$"$k$3$H$KCm0U$7$F2<$5$$!%$h$/!$%U%!%$%k$O(J
$@%;!<%V$7$?$,!$!V(J.eps $@%U%!%$%k!W$r99?7$7K:$l$k$H$$$&$3$H$,$"$j$^$9!%(Jtgif+ $@$G(J
$@?^$OIA$-D>$7$F%U%!%$%k$O%;!<%V$7$?$N$K!$$=$N?^$r<h$j9~$s$G$$$k(J jLaTeX $@$NJ8=q$r(J
$@%W%j%s%H%"%&%H$9$k$H?^$,D>$C$F$$$J$$!$$H$$$&2x8=>]$N860x$G$9!%(J
\medskip
\par
$@?^$rD>$7$?$H$-$O!$!V(J.eps $@%U%!%$%k!W$N99?7$bK:$l$J$$$h$&$K$7$F2<$5$$(J
$@!J!V(JChoice $@%&%#%s%I%&!W$N!V%W%j%s%H@h!W$r!V(JLaTeX $@%"%$%3%s!W$K$7$F!V(JPrint$@!W$9$k!K!%(J
\medskip
\par
$@K:$l$d$9$$?M$O!$!V(J$-$swe$@!W$+!V(JTgif*SaveWithEPS: on$@!W$r;XDj$7$F2<$5$$!%(J
$@$3$&$9$k$H!$(Jtgif+ $@$G%U%!%$%k$r%;!<%V$7$?$H$-$K!$F1;~$K!V(J.eps $@%U%!%$%k!W(J
$@$b@8@.$7$F$/$l$^$9!J!V(J\S~\ref{option}~$@%*%W%7%g%s!W!$!VIUO?(J~\ref{xdefaults}~
X DEFAULTS$@!W;2>H!K!%(J
\medskip
\newpage

\section{$@%*%V%8%'%/%H$NIA2h$HA`:n(J}
\label{operation}
\medskip
$@%*%V%8%'%/%H!J(Jtgif+ $@$GIA2h$G$-$k$b$N!K$NA`:n$O!$!V(JMode$@%a%K%e!<!W!J?^(J~\ref{modeMenuWin}~$@;2>H!K$N!VLp0u!W%"%$%3%s$rA*Br$7$F!$!V(J
$@A`:n%b!<%I!W$K$7$F9T$$$^$9!%%G%U%)%k%H$O!$!VA`:n%b!<%I!W$G$9!%(J
\medskip
\par
$@$=$l0J30$N$b$N$rA*Br$9$k$H!$!VIA2h%b!<%I!W$K$J$j$^$9!%(J
\medskip

\subsection{$@IA2h(J}
\medskip
\par
$@!VIA2h%b!<%I!W$G$O!$!V%F%-%9%H!W0J30$G$O!$%+!<%=%k$OGr$/$J$j$^$9!%(J
\medskip
\par
$@%*%V%8%'%/%H$NIA2h$N4pK\$O!$!V:8%\%?%s!W$GIA2h$N;OE@$r7h$a!$$=$N$^$^!V(J
$@%I%i%C%0!W$7!$%\%?%s$r!VN%$9!W$3$H$GIA2h$,=*N;$7$^$9(J
\footnote{$@%]%j%4%s!$%]%j%i%$%s!$1_8L$O>/$7%$%s%?%U%'!<%9$,0c$$$^$9!%(J}$@!%(J
\par
$@0J2<$N@bL@$O!$$*$*$^$+$K$7$+IA2hJ}K!$r@bL@$7$F$$$^$;$s$N$G!$(J
$@<B:]$KN}=,$7$FIA2hJ}K!$r3P$($F2<$5$$!%(J
\medskip
\par
$@$J$*!$%*%V%8%'%/%H$O!VA`:n%b!<%I!W$KLa$i$J$$8B$j!$F1$8$b$N$,B3$1$FIA$1(J
$@$^$9(J\footnote{$@0l;~E*$K!VA`:n%b!<%I!W$K$7$?$$>l9g$O!$!V(J\^{}E$@!W!J%3%s%H%m!<%k$r(J
$@2!$7$J$,$i!V(JE$@!W$r2!$9!%(JE $@$O>.J8;z!K$H(J
$@F~NO$7$F2<$5$$!%$b$&0lEY!V(J\^{}E$@!W$rF~NO$9$k$^$G$O!$%+!<%=%k$,!V9u$/!W$J$C$F(J
$@!VA`:n%b!<%I!W$K$J$j$^$9!%:FEY!V(J\^{}E$@!W$rF~NO$9$k$H!$85$N!VIA2h%b!<%I!W$K(J
$@La$j$^$9!%(J}$@!%(J
\medskip

\subsubsection{$@;M3Q!$1_!$3Q4];M3Q(J}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JMode$@%a%K%e!<!W$+$i!$!V;M3Q7A!$1_!$3Q4];M3Q7A!W$N$$$:$l$+$rA*Br$9$k!%(J
\item $@%+!<%=%k$,!VGr$/!W$J$j$^$9!%(J
\item $@!V:8%\%?%s!W$r2!$7$F!$%I%i%C%0$72<$5$$!%%\%?%s$r2!$7$?>l=j$,;OE@$K$J(J
$@$j?^$,I=<($5$l$^$9!%(J
\item $@%\%?%s$rN%$9$H?^$,3NDj$7$^$9!%(J
\medskip
\end{enumerate}
}

\subsubsection{$@%F%-%9%H(J}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JMode$@%a%K%e!<!W$+$i!$!V(JT$@!W$rA*Br$9$k!%(J
\item $@%+!<%=%k$,!V(JI$@!W$N;z$K$J$j$^$9!%(J
\item $@!V(JCanvas$@%&%#%s%I%&!WFb$G!$%F%-%9%H$rF~NO$7$?$$>l=j$G!$:8%\%?%s$r(J
$@%/%j%C%/$7$F2<$5$$!%!V(JI$@!W$N;z$N%+!<%=%k$,I=<($5$l$^$9!%(J
\item $@%-!<%\!<%I$rC!$1$PJ8;z$,F~$j$^$9!%$?$@$7!$!V(Jtab$@!W$OF~$j$^$;$s!%(J
\item $@F|K\8l$rF~NO$9$k>l9g$O!$!V(JI$@!W%+!<%=%k$,I=<($5$l$?>uBV$G!$!V%7%U%H!\%9%Z!<%9!W$G(Jkinput $@$r5/F0$9$k$+!$(Jnemacs $@$+$i%Z!<%9%H$7$F2<$5$$!%(J
\end{enumerate}
}
$@F|K\8lF~NO$O!$!V(J\S~\ref{japanese} $@F|K\8l!W$r;2>H$7$F2<$5$$!%(J
\medskip

\subsubsection{$@%]%j%4%s!$%]%j%i%$%s(J}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JMode$@%a%K%e!<!W$+$i!$!V%]%j%4%s!$%]%j%i%$%s!W$N$$$:$l$+$rA*Br$9$k!%(J
\item $@%+!<%=%k$,!VGr$/!W$J$j$^$9!%(J
\item $@:8%\%?%s$r%/%j%C%/$9$k$H!$;OE@$,7h$^$j$^$9!%(J
\item $@$=$N$^$^%^%&%9$rF0$+$9$H@~$,0z$+$l$^$9!%(J
\item $@!V:8%\%?%s!W$r%/%j%C%/$9$k$H@^$l@~$K$J$j!$!V??Cf%\%?%s!W$r%/%j%C%/$9$k$H3NDj$7$^$9!%(J
\medskip
\end{enumerate}
}

\subsubsection{$@1_8L(J}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!V(JMode$@%a%K%e!<!W$+$i!$!V1_8L!W$rA*Br$9$k!%(J
\item $@%+!<%=%k$,!VGr$/!W$J$j$^$9!%(J
\item $@:8%\%?%s$r%/%j%C%/$9$k$H!$1_8L$NCf?4$,7h$^$j$^$9!%(J
\item $@$=$N$^$^%^%&%9$rF0$+$9$H@~$,0z$+$l$^$9!J$3$l$,1_8L$NH>7B$K$J$j$^$9!K!%(J
\item $@!V:8%\%?%s!W$r%/%j%C%/$9$k$H;OE@$,7h$^$j$^$9!%%^%&%9$rF0$+$9$H1_8L$,IA$+$l$^$9!%(J
\item $@!V??Cf%\%?%s!W$r%/%j%C%/$9$k$H3NDj$7$^$9!%(J
\medskip
\end{enumerate}
}

\subsection{$@A`:n(J}
\medskip
\par
$@!V(JMode$@%a%K%e!<!W$N!VLp0u!W%"%$%3%s$rA*Br$7$^$9!%%+!<%=%k$O!V9u$/!W$J$j(J
$@$^$9!%(J
\medskip

\subsubsection{$@A*Br(J}
\label{selection}
\medskip
\par
$@%*%V%8%'%/%H$NB0@-!JNc$($P!$@~$NB@$5!K$rJQ$($?$$>l9g$d!$0\F0!$3HBg$7$?(J
$@$$>l9g$K$O!$$^$:%*%V%8%'%/%H$rA*Br$7$F2<$5$$!%(J
\medskip

\paragraph{$@0l$DA*Br$9$k(J}~\\
\medskip
\par
$@A*Br$7$?$$%*%V%8%'%/%H$NJU$N>e$G!$%^%&%9$N!V:8%\%?%s!W$G!$%/%j%C%/$7$F2<$5(J
$@$$!%%*%V%8%'%/%H$N<~$j$K>.$5$J!V;M3Q!W$,I=<($5$l$?$i!$$=$N%*%V%8%'(J
$@%/%H$O!VA*Br!W$5$l$?$3$H$K$J$j$^$9!J?^(J~\ref{selection1} $@$NA0H>;2>H!K!%(J

        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/select1.eps}
                \caption{$@%/%j%C%/$H%7%U%H(J+$@%/%j%C%/(J}
                \label{selection1}
           \end{center}
        \end{figure}
\medskip

\paragraph{$@Fs$D0J>eA*Br$9$k!%(J}~\\
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@%7%U%H(J+$@%/%j%C%/(J
\par
$@%7%U%H%\%?%s$r2!$7$J$,$i!$!V:8%\%?%s!W$G%*%V%8%'%/%H$r%/%j%C%/$7(J
        $@$F$$$/$HJ#?t8DA*Br$9$k$3$H$,$G$-$^$9!J?^(J~\ref{selection1} $@$N8eH>;2>H!K!%(J
\medskip
\par
$@$^$?!$$9$G$KA*Br$5$l$F$$$k%*%V%8%'%/%H$r!$%7%U%H%\%?%s$r2!$7$J(J
$@$,$i!$!V:8%\%?%s!W$G%/%j%C%/!JA*Br!K$9$k$H!VA*Br$,2r=|!W$5$l$^$9!%(J
$@Nc$($P!$$?$/$5$s%*%V%8%'%/%H$,(J
$@$"$C$F!$$=$N$[$H$s$I$rA*Br$7$?$$>l9g$K$O!$!V(J\verb+^+A$@!W$G$9$Y$F$rA*Br$7$F!$(J
$@$$$i$J$$$b$N$r!VA*Br$r2r=|!W$9$k$H4JC1$G$9(J.
\medskip
\item $@%I%i%C%0(J
\medskip
{
\renewcommand{\theenumii}{(\alph{enumii})}
\renewcommand{\labelenumii}{\theenumii}
\begin{enumerate}
\item $@%*%V%8%'%/%H$NL5$$$H$3$m$G!$!V:8%\%?%s!W$r2!$7$J$,$i%^%&%9$r%I(J
        $@%i%C%0$9$k$H;M3Q$$%i%P!<%P%s%I$,I=<($5$l$^$9!J?^(J~\ref{selection2} $@;2>H!K!%(J
\item $@%i%P!<%P%s%I$G0O$^$l$?%*%V%8%'%/%H$,$9$Y$FA*Br$5$l$^$9(J
\end{enumerate}
}
        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/select2.eps}
                \caption{$@%I%i%C%0$K$h$kA*Br(J}
                \label{selection2}
           \end{center}
        \end{figure}
\medskip
\end{enumerate}
}

\paragraph{$@$9$Y$F$rA*Br$9$k!%(J}~\\
\medskip
\par
$@!V(JEdit$@%a%K%e!<!W$G!V(JSelectAll$@!W$rA*Br$9$k$+!$!V(J\verb+^+A$@!W$G!$$9$Y$F$N%*%V%8%'(J
$@%/%H$,A*Br$G$-$^$9!%(J
\medskip

\subsubsection{$@0\F0(J}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!VA*Br!W$5$l$?%*%V%8%'%/%H$N@~$N>e$N!V>.$5$J;M3Q0J30!W$NItJ,$K%^%&%9%+!<%=%k$r;}$C$F$$$/!%(J
\item $@!V:8%\%?%s!W$r2!$7$?$^$^%I%i%C%0$9$k$H0\F0$G$-$^$9!%(J
\item $@%\%?%s$rN%$9$H$=$N>l=j$KCV$+$l$^$9!%(J
\medskip
\end{enumerate}
}

\subsubsection{$@3HBg!$=L>.(J}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@!VA*Br!W$5$l$?%*%V%8%'%/%H$N2s$j$KI=<($5$l$?!V>.$5$J;M3Q!W>e$G!V:8%\%?%s!W$r2!$9!%(J
\item $@!V:8%\%?%s!W$r2!$7$?$^$^%i%C%0$9$k$H3HBg!J=L>.!K$G$-$^$9!%(J
\item $@%\%?%s$rN%$9$H$=$NBg$-$5$K3HBg!J=L>.!K$5$l$^$9!%(J
\medskip
\end{enumerate}
}
\par
$@$?$@$7!$%]%j%4%s!$%]%j%i%$%s$N>l9g$O!$Bg$-$5$G$O$J$/7A$NJQ99$K$J$j$^$9!%(J
\medskip

\section{$@%^%&%9%\%?%s$N0UL#(J}
\label{mouseButton}
\medskip
\par
$@:#$^$G$K@bL@$7$?%^%&%9%\%?%s$N0UL#$r$^$H$a$^$9!%(J
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|c|}
\hline
$@:8%\%?%s(J   & \multicolumn{2}{|l|}{$@%*%V%8%'%/%H$NIA2h!$A*Br!$JQ7A(J             }\\
\hline
$@??Cf%\%?%s(J & $@%]%j%i%$%s!$%](J  & $@IA2h$N=*N;(J                 \\
           & $@%j%4%s$NIA2hCf(J  &                            \\
\cline{2-2}\cline{3-3}           & $@J8;zNs$NIA2hCf(J  & $@NN0h$N;XDj(J                 \\
\cline{2-2}\cline{3-3}           & $@$=$l0J30(J        & Mode $@%a%K%e!<$N%]%C%W%"%C%W(J\\
\hline
$@1&%\%?%s(J   & \multicolumn{2}{|l|}{tgif+ $@$N%a%$%s%a%K%e!<$N%]%C%W%"%C%W(J       }\\
\hline
\end{tabular}\\
\end{center}
\medskip
\newpage

\section{tgif+ $@$N@8@.$9$k%U%!%$%k(J}
\label{files}
\medskip
\medskip
\par
tgif+ $@$O!$<!$N(J4$@<oN`$N%U%!%$%k$r@8@.$9$k$3$H$,$G$-$^$9!J3g8LFb$O!$%U%!(J
$@%$%kL>$N3HD%;R$rI=$7$^$9!K!%(J
\medskip
\begin{nquote}{4ex}
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@%*%V%8%'%/%H%U%!%$%k!J(J.obj$@!K(J... $@0J2<!$(J.obj $@%U%!%$%k(J
\item $@%S%C%H%^%C%W%U%!%$%k!J(J.xbm$@!K(J... $@0J2<!$(J.xbm $@%U%!%$%k(J
\item $@%+%W%;%k2=(J PostScript $@%U%!%$%k!J(J.eps$@!K(J... $@0J2<!$(J.eps $@%U%!%$%k(J
\item PostScript $@%U%!%$%k!J(J.ps$@!K(J... $@0J2<!$(J.ps $@%U%!%$%k(J
\medskip
\end{enumerate}
}
\end{nquote}
tgif+ $@$H%U%!%$%k$N4X78$O!$?^(J \ref{TgifAndFile} $@$r;2>H$7$F2<$5$$!%(J
\medskip

\subsection{.obj $@%U%!%$%k!$(J.xbm $@%U%!%$%k(J}
\medskip
\par
$@!V(J.obj $@%U%!%$%k!W$O!$(Jtgif+ $@$,IA2h7k2L$NFI$_=q$-$K;HMQ$9$k%U%!%$(J
$@%k$G$9!%FbIt7A<0$O!$%"%9%-!<$G$9!J>\$7$$FbIt7A<0$O!$%=!<%9%3!<%I(J
$@$KIUB0$9$k(J tgif.pl $@$r;2>H$7$F2<$5$$!K!%!V(J.obj $@%U%!%$%k!W$r$=$N$^$^(J lpr $@%3%^%s%IEy(J
$@$G%W%j%s%?$KAw$C$F$b?^$O%W%j%s%H%"%&%H$G$-$^$;$s(J
\footnote{$@$J$*!$$3$N%U%!%$%k$r(J tgif+ $@$r;HMQ$;$:(J
$@$KD>@\%W%j%s%H%"%&%H$9$k>l9g$O!$!V(Jprtgif+ $@%3%^%s%I!W$r;H$C$F2<$5$$(J
$@!J!V(J\S~\ref{prtgif+}~prtgif+ $@$N;H$$J}!W;2>H!K!%(J}$@!%(J
\medskip
\par
$@$^$?!$!V(J.xbm $@%U%!%$%k!W$O!$(JX11 $@$N%S%C%H%^%C%W7A<0$N%U%!%$%k$NF~=PNO$K;H$o(J
$@$l$k%U%!%$%k$G$9!J>\$7$$FbIt7A<0$O!$(JX$@%&%#%s%I%&$N%^%K%e%"%k$r;2>H$7$F2<$5$$!K!%(J
\medskip

\subsection{.eps $@%U%!%$%k!$(J.ps $@%U%!%$%k(J}
\medskip
\par
$@;D$j$N(J2$@$D$N%U%!%$%k$N$&$A!V(J.eps $@%U%!%$%k!W$O!$(Jtgif+ $@$GIA$+$l$??^$r(J jLaTeX
$@$G=q$+$l$?J8=q$K<h$j9~$a$k$h$&$K!$(Jtgif+ $@$,JQ49@8@.$9$k%U%!%$%k$G$9!%(J
\medskip
\par
$@$^$?!$!V(J.ps $@%U%!%$%k!W$O!$$=$N$^$^(J PostScript $@%W%j%s%?$K=PNO$9$k$3$H$,$G(J
$@$-$k%U%!%$%k$G$9(J($@Nc$($P!$(Jlpr $@$d(J cat $@%3%^%s%I$G!K!%(J
\medskip
\par
$@!V(J.eps $@%U%!%$%k!W!$!V(J.ps $@%U%!%$%k!W$O!$(JjLaTeX $@MQ!$%W%j%s%?MQ$K:n@.$5$l$k%U%!%$%k$G$9(J
$@$N$G!$(Jtgif+ $@$O!$$3$l$i$rFI$_9~$`$3$H$O$G$-$^$;$s!%(J
\medskip

\subsection{$@3F%U%!%$%k$N@8@.$N;EJ}(J}
\medskip
\par
$@!V(J.obj $@%U%!%$%k!W$O!$(Jtgif+ $@$GIA2h$7$??^$r%;!<%V$9$k$3$H$G@8@.$5$l!$$=$l0J30$N%U%!%$%k$O!$%W%j%s%H%"%&%H@h$r;X(J
$@Dj$7$F!$%W%j%s%H$9$k$3$H$K$h$C$F%U%!%$%k$,@8@.$5$l$^$9!%(J
\medskip
~\\
\begin{tabular}{|l|l|}
\hline
\multicolumn{1}{|c|}{     $@<oN`(J    } & \multicolumn{1}{|c|}{                $@J}K!(J                                   }\\
\hline\hline
.obj $@%U%!%$%k(J & $@!V(JFile $@%a%K%e!<!W$G!V(JSave$@!W$7$F2<$5$$(J.                 \\
\hline
.xbm $@%U%!%$%k(J & $@!J(J1$@!K!V(JChoice $@%&%#%s%I%&!W$G!V(JX11 BM $@%"%$%3%s!W$K$9$k!%(J\\
              & $@!J(J2$@!K!V(JFile $@%a%K%e!<!W$G!V(JPrint$@!W$7$F2<$5$$(J.           \\
\hline
.eps $@%U%!%$%k(J & $@!J(J1$@!K!V(JChoice $@%&%#%s%I%&!W$G!V(JLaTeX $@%"%$%3%s!W$K$9$k!%(J \\
              & $@!J(J2$@!K!V(JFile $@%a%K%e!<!W$G!V(JPrint$@!W$7$F2<$5$$(J.           \\
\hline
.ps $@%U%!%$%k(J  & $@!J(J1$@!K!V(JChoice $@%&%#%s%I%&!W$G!V(JPS $@%"%$%3%s!W$K$9$k!%(J    \\
              & $@!J(J2$@!K!V(JFile $@%a%K%e!<!W$G!V(JPrint$@!W$7$F2<$5$$(J.           \\
\hline
\end{tabular}\\
\medskip
\medskip
\medskip
$@%W%j%s%H%"%&%H@h$NA*Br$N;EJ}$O!$?^(J \ref{SaveFiles}$@$r;2>H$7$F2<$5$$!%(J

        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/SaveFiles.eps}
                \caption{$@%W%j%s%H%"%&%H@h$NA*Br(J}
                \label{SaveFiles}
           \end{center}
        \end{figure}
\medskip
\par
$@$A$J$_$K!$!V(JChoice $@%&%#%s%I%&!W$G!V(JPrinter $@%"%$%3%s!W$K$7$F!$!V(JPrint$@!W(J
$@$9$k$H%W%j%s%?$K=PNO$9$k$3$H$K$J$j$^$9!%(J
\medskip
                \begin{figure}[htbp]
                    \begin{center}
                        \epsfile{file=FIGURE/Tgif+File.eps}
                        \caption{tgif+ $@$H3F%U%!%$%k$N4X78(J}
                        \label{TgifAndFile}
                    \end{center}
                \end{figure}
\medskip
\medskip
\newpage

\section{$@%*%W%7%g%s(J}
\label{option}
\medskip
\par
tgif+ $@$G$O!$0J2<$N%*%W%7%g%s$r%5%]!<%H$7$F$$$^$9!%$[$H$s$I$N%*%W%7%g%s(J
$@$O!$(J\verb+~+/.Xdefaults $@$G;XDj2DG=$G$9(J
$@!J!VIUO?(J~\ref{xdefaults} XDEFAULTS$@!W(J $@;2>H!K!%(J
\medskip
\par
$-$display displayName
\begin{nquote}{8ex}
\par
$@;HMQ$9$k%5!<%P$,2TF/$9$k%[%9%HL>$H%G%#%9%W%l%$HV9f$r;XDj!%(J
\end{nquote}
\medskip
\par
$-$reverse $@!J$^$?$O!$(J$-$rv$@!K(J
\begin{nquote}{8ex}
\par
$@2hLL$rH?E>I=<($7$^$9!J%G%U%)%k%H$O!$%U%)%"%0%i%s%I?'$,9u!$%P%C(J
$@%/%0%i%s%I?'$,Gr!K!%(J
\end{nquote}
\medskip
\par
$-$cd path
\begin{nquote}{8ex}
\par
path $@$G;XDj$7$?%G%#%l%/%H%j$,!$%+%l%s%H%G%#%l%/%H%j$K$J$j$^$9!%(J
\end{nquote}
\medskip
\par
$-$eps
\begin{nquote}{8ex}
\par
$@%G%U%)%k%H$N%W%j%s%H%"%&%H@h$r!V(J.eps$@%U%!%$%k!W$K$7$^$9!J%G%U%)%k%H$O!$%W%j%s%?!K!%(J
\end{nquote}
\medskip
\par
$-$euc
\begin{nquote}{8ex}
\par
PostScript  $@$NF|K\8l%U%)%s%H$H$7$F(J Ryumin$-$Light$-$Euc $@$H(J
GothicBBB$-$Medium$-$Euc $@$r;HMQ$9$k!J%G%U%)%k%H$O!$(JRyumin$-$Light$-$
Ext $@$H(J GothicBBB$-$Medium$-$Ext$@!K!%(J
\end{nquote}
\medskip
\par
$-$autokinput($@$^$?$O!$(J$-$ak$@!K(J
\begin{nquote}{8ex}
\par
        $@%+%l%s%H%U%)%s%H$,!V(JMinchou$@!W$+!V(JGothic$@!W$N>l9g$K!$%F%-%9%HF~NO;~$K!$(J
        $@<+F0E*$K(J kinput $@$H@\B3$9$k!J(J
 $@!V(J\S~\ref{japanese}~$@F|K\8l!W;2>H!K(J\footnote{$@$3$N%*%W%7%g%s$r;XDj$9$k$H(J
kinput $@$,5/F0$5$l$F$$$J$$>l9g$O!$!V(Jtgif+ $@$,F0$$$F$$$k%[%9%H!W$G(J kinput $@$r(J
$@5/F0$7@\B3$7$^$9!%$G$9$+$i!$(Jkinput $@$,%$%s%9%H!<%k$5$l$F$$$J$$%[%9%H$G;H$C$F$$$k(J tgif+
$@$G!$$3$N%*%W%7%g%s$r;H$&>l9g$K$O!$$"$i$+$8$a(J kinput $@$rJL$N!V(Jkinput $@$,%$%s%9%H!<%k$5$l$F$$$k%[%9%H!W$G5/F0$7$F$*$/I,MW$,$"$j$^$9!J!VIUO?(J~\ref{kinput}~kinput $@$N;H$$J}!W;2>H!K!%(J}$@!%(J
\end{nquote}
\medskip
\par
$-$larc
\begin{nquote}{8ex}
\par
$@1_8L$NIA2h$r(JX$@%i%$%V%i%j!J(JXDrawArc()$@!K$rMQ$$$F9T$&!J%G%U%)%k%H(J
$@$G$O!$(Jtgif+$@FbIt$G=hM}$r9T$C$F$$$^$9!K!%$3$l$O!$(JXDrawArc() $@$NJ}(J
$@$,1_8L$NIA2h$,B.$$>l9g$K;XDj$9$k$HNI$$!JNc$($P!$(JNCD $@$N(J X11R4$@!K!%(J
\end{nquote}
\medskip
\par
$-$hidedragcursor$@!J$^$?$O!$(J$-$hc$@!K(J
\begin{nquote}{8ex}
\par
$@%I%i%C%0Cf$N%^%&%9%+!<%=%k$rI=<($7$J$$!J%G%U%)%k%H$O!$$9$k!K!%(J
\end{nquote}
\medskip
\par
$-$nokanji
\begin{nquote}{8ex}
\par
$@F|K\8l5!G=$r;H$o$J$$!%4A;z$r4^$^$J$$(J .eps $@$r:n@.$9$k>l9g$K;XDj$9$k$HJXMx!%(J
\end{nquote}
\medskip
\par
$-$savewitheps$@!J$^$?$O!$(J$-$swe$@!K(J
\begin{nquote}{8ex}
\par
        $@%U%!%$%k$N%;!<%V;~$K!V(J.eps $@%U%!%$%k!W$b@8@.$7$^$9!J(J
$@!V(J\S~\ref{howToInclude}~jLaTeX $@$X$N?^$N<h$j9~$_!W;2>H!K!%(J
\end{nquote}
\newpage

\section{$@IA2h%*%V%8%'%/%H$HIA2hB0@-(J}
\label{objectAndProperty}
\medskip
\par
tgif+ $@$GIA2h$G$-$k$b$N$H!$$=$l$KE,MQ$5$l$kIA2hB0@-$K$D$$$F=R$Y$^$9!%(J
\medskip

\subsection{$@IA2h%*%V%8%'%/%H(J}
\medskip
\par
tgif+ $@$GIA2h$G$-$k$b$N$O!$<!$N(J8$@$D$G$9!%0J2<$3$l$i$r$^$H$a$FIA2h%*%V%8(J
$@%'%/%H!J$b$7$/$O!$C1$K%*%V%8%'%/%H!K$H8F$S$^$9(J
$@!J?^(J~\ref{objects}~$@;2>H!K!%(J
\begin{nquote}{4ex}
\begin{itemize}
\item $@%F%-%9%H!JJ8;zNs!K(J
\item $@;M3Q7A(J
\item $@BJ1_(J
\item $@%]%j%i%$%s!JB?3Q@~!K(J
\item $@%]%j%4%s!JB?3Q7A!K(J
\item $@BJ1_8L(J
\item $@3Q4];M3Q7A(J
\item $@%S%C%H%^%C%W(J
\medskip
\end{itemize}
\end{nquote}
        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/objects.eps}
                \caption{$@IA2h%*%V%8%'%/%H$NNc(J}
                \label{objects}
           \end{center}
        \end{figure}

\medskip

\subsection{$@IA2hB0@-(J}
\label{property}
\medskip
\par
tgif+ $@$G!$3FIA2h%*%V%8%'%/%H$KBP$7$F!$;XDj$G$-$kIA2hB0@-$O!$<!$N(J7$@$D$G$9!%(J
\medskip
\begin{itemize}
\item $@EI$j$D$V$7%Q%?!<%s(J
\par
$@JD6JLL$rEI$j$D$V$9$N$K;XDj$7$^$9(J
$@!J?^(J~\ref{menuWins}~$@;2>H!K!%(J
\item $@%Z%s%Q%?!<%s(J
\par
$@@~J,$d%F%-%9%H$rEI$j$D$V$9$N$K;XDj$7$^$9(J
$@!J?^(J~\ref{menuWins}~$@;2>H!K!%(J
\item $@@~%9%?%$%k(J
\par
$@@~J,$N7A>u$r;XDj$7$^$9!%$3$l$K$O!$<!$N(J4$@<oN`$,$"$jFHN)$K;XDj$G$-$^$9(J
$@!J?^(J~\ref{menuWins}~$@;2>H!K!%(J
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@@~I}!J(J6 $@<oN`!K(J
\item $@GK@~%Q%?!<%s!J(J4 $@<oN`!K(J
\item $@@~$NC<E@7A>u!J(J4 $@<oN`!K(J
\item $@%]%j%i%$%s"+"*(J $@%9%W%i%$%sJQ49(J
\end{enumerate}
}
\item $@%F%-%9%H%9%?%$%k(J
\par
$@<!$N$h$&$J%F%-%9%H$NB0@-$r;XDj$7$^$9!%(J
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@%m!<%^%sBN!$%\!<%k%IBN!$%$%?%j%C%/BN!$%\!<%k%I%$%?%j%C%/BN(J
\item $@1&4s$;!$Cf1{B7$(!$:84s$;(J
\end{enumerate}
}
\par
$@C"$7!$4A;z$O%m!<%^%sBN$7$+;H$($^$;$s!%(J
\item $@%F%-%9%H%5%$%:(J
\par
$@J8;z$NBg$-$5$r;XDj$7$^$9!%$3$l$O!$2<$N!V%U%)%s%H(JDPI$@!W$4$H$K(J
$@%5%$%:$,;XDj$G$-!$<!$N$h$&$K$J$C$F$$$^$9!J%5%$%:$O%]%$%s%H$G$O$"$j$^$;$s!K!%(J
\par
$@!J(J1$@!K(J75dpi $@$N>l9g$O!$(J8$@!$(J10$@!$(J12$@!$(J14$@!$(J18$@!$(J24
$@!J(J2$@!K(J100dpi $@$N>l9g!$(J11$@!$(J14$@!$(J17$@!$(J20$@!$(J25$@!$(J34
\par
$@C"$7!$4A;z$O!$!J(J1$@!K$N(J75dpi$@$G$7$+;H$($^$;$s!%(J
\item $@%U%)%s%H(J
\par
$@J8;z$N7A>u$r;XDj$7$^$9!%<!$N(J7$@<oN`$N%U%)%s%H$,;XDj$G$-$^$9!%(J
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item Times
\item Courier
\item Helvetica
\item NewCentury
\item Symbol$@!J5-9f!K(J
\item Minchou$@!JL@D+BN!K(J
\item Gothic$@!J%4%7%C%/BN!K(J
\medskip
\end{enumerate}
}
        $@?^(J~\ref{fonts}~$@$KNc$r<($7$^$9(J\footnote{NewCentury $@$O!$%W%j%s%?$K%U%)%s%H$,$J$$$?$a$K!$0u:~>e$O(J Courier $@$,;H$o$l$F$$$^$9!%(J}$@!%(J


        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/fonts.eps}
                \caption{$@%U%)%s%H$NNc(J}
                \label{fonts}
           \end{center}
        \end{figure}

\medskip
\item $@%U%)%s%H(JDPI
\end{itemize}
                $@%U%)%s%H$N(JDPI$@!J(JDot Per Inch$@!K$r;XDj$7$^$9!%(J75$@$H(J100$@$,;XDj$G$-$^$9(J
\footnote{$@%U%)%s%H(JDPI $@$,NI$/J,$+$i$J$$J}$O!$?<$/9M$($:$K(J $@!V(JFontDPI $@%a%K%e!<!W$,(J
$@!V(J75dpi$@!W$K$J$C$F$$$k$3$H$r3NG'$7$F;H$C$F;H$C$F2<$5$$!%(J}$@!%(J
\medskip

\subsection{$@IA2h%*%V%8%'%/%H$HIA2hB0@-$N4X78(J}
\medskip
\par
$@0J2<$K!$IA2h%*%V%8%'%/%H$HIA2hB0@-$N4X78$rI=$K<($7$^$9!%(J
\medskip
\begin{nquote}{4ex}
\begin{itemize}
\item $@;M3Q7A!$BJ1_!$%]%j%i%$%s!$%]%j%4%s!$BJ1_8L!$3Q4];M3Q7A(J
~\\
\begin{center}
\begin{tabular}{|l|c|l|r|r|r|l|l|}
\hline
       & $@IA2hB0@-(J  & $@EI$j$D$V$7(J & \multicolumn{1}{|l|}{$@%Z%s(J    } & \multicolumn{4}{|l|}{$@@~$N%9%?%$%k(J                }\\
\cline{2-2}\cline{5-5}\cline{6-6}\cline{7-7}\cline{8-8}       &  & $@%Q%?!<%s(J   & $@%Q%?!<%s(J &  &  &  & \\
\multicolumn{2}{|l|}{$@IA2h%*%V%8%'%/%H(J} &            &          & \multicolumn{1}{|c|}{ $@I}(J } & \multicolumn{1}{|l|}{$@GK@~(J } & $@Lp0u(J & $@%9%W%i%$%s2=(J\\
\hline
\multicolumn{2}{|l|}{$@;M3Q7A(J          } & $@!{(J         & \multicolumn{1}{|c|}{ $@!{(J     } & \multicolumn{1}{|c|}{ $@!{(J } & \multicolumn{1}{|c|}{ $@!{(J  } & $@!_(J   & $@!_(J          \\
\hline
\multicolumn{2}{|l|}{$@BJ1_(J            } & $@!{(J         & \multicolumn{1}{|c|}{ $@!{(J     } & \multicolumn{1}{|c|}{ $@!{(J } & \multicolumn{1}{|c|}{ $@!{(J  } & $@!_(J   & $@!_(J          \\
\hline
\multicolumn{2}{|l|}{$@%]%j%i%$%s(J      } & $@!{(J         & \multicolumn{1}{|c|}{ $@!{(J     } & \multicolumn{1}{|c|}{ $@!{(J } & \multicolumn{1}{|c|}{ $@!{(J  } & $@!{(J   & $@!{(J          \\
\hline
\multicolumn{2}{|l|}{$@%]%j%4%s(J        } & $@!{(J         & \multicolumn{1}{|c|}{ $@!{(J     } & \multicolumn{1}{|c|}{ $@!{(J } & \multicolumn{1}{|c|}{ $@!{(J  } & $@!_(J   & $@!{(J          \\
\hline
\multicolumn{2}{|l|}{$@BJ1_8L(J          } & $@!{(J         & \multicolumn{1}{|c|}{ $@!{(J     } & \multicolumn{1}{|c|}{ $@!{(J } & \multicolumn{1}{|c|}{ $@!{(J  } & $@!_(J   & $@!_(J          \\
\hline
\multicolumn{2}{|l|}{$@3Q4];M3Q7A(J      } & $@!{(J         & \multicolumn{1}{|c|}{ $@!{(J     } & \multicolumn{1}{|c|}{ $@!{(J } & \multicolumn{1}{|c|}{ $@!{(J  } & $@!_(J   & $@!_(J          \\
\hline
\multicolumn{2}{|l|}{$@%S%C%H%^%C%W(J    } & $@!{(J         & \multicolumn{1}{|c|}{ $@!_(J     } & \multicolumn{1}{|c|}{ $@!_(J } & \multicolumn{1}{|c|}{ $@!_(J  } & $@!_(J   & $@!_(J          \\
\hline
\end{tabular}\\
\end{center}
\medskip
\item $@%F%-%9%H!JJ8;zNs!K(J
\end{itemize}
\end{nquote}
~\\
\begin{center}
\begin{tabular}{|r|c|l|l|l|r|}
\hline
  & $@IA2hB0@-(J & $@%F%-%9%H(J        & $@%F%-%9%H(J & $@%U%)%s%H(JDPI & $@%Z%s%Q%?!<%s(J\\
\cline{2-2}  &  & $@%9%?%$%k(J        & $@%5%$%:(J   &             &             \\
\multicolumn{2}{|l|}{$@%U%)%s%H(J  } &                 &          &             &             \\
\hline
\multicolumn{2}{|l|}{Times     } & \multicolumn{1}{|c|}{ $@!{(J            } & $@!{(J       & \multicolumn{1}{|c|}{ $@!{(J        } & \multicolumn{1}{|c|}{ $@!{(J         }\\
\hline
\multicolumn{2}{|l|}{Courier   } & \multicolumn{1}{|c|}{ $@!{(J            } & $@!{(J       & \multicolumn{1}{|c|}{ $@!{(J        } & \multicolumn{1}{|c|}{ $@!{(J         }\\
\hline
\multicolumn{2}{|l|}{Helvetica } & \multicolumn{1}{|c|}{ $@!{(J            } & $@!{(J       & \multicolumn{1}{|c|}{ $@!{(J        } & \multicolumn{1}{|c|}{ $@!{(J         }\\
\hline
\multicolumn{2}{|r|}{NewCentury} & \multicolumn{1}{|c|}{ $@!{(J            } & $@!{(J       & \multicolumn{1}{|c|}{ $@!{(J        } & \multicolumn{1}{|c|}{ $@!{(J         }\\
\hline
\multicolumn{2}{|l|}{Symbol    } & \multicolumn{1}{|r|}{ $@%m!<%^%sBN$N$_(J} & $@!{(J       & \multicolumn{1}{|c|}{ $@!{(J        } & \multicolumn{1}{|c|}{ $@!{(J         }\\
\hline
\multicolumn{2}{|l|}{$@L@D+BN(J    } & \multicolumn{1}{|r|}{ $@%m!<%^%sBN$N$_(J} & $@!{(J       & \multicolumn{1}{|r|}{ 75dpi $@$N$_(J} & \multicolumn{1}{|c|}{ $@!{(J         }\\
\hline
\multicolumn{2}{|r|}{$@%4%7%C%/BN(J} & \multicolumn{1}{|r|}{ $@%m!<%^%sBN$N$_(J} & $@!{(J       & \multicolumn{1}{|r|}{ 75dpi $@$N$_(J} & \multicolumn{1}{|c|}{ $@!{(J         }\\
\hline
\end{tabular}\\
\end{center}
\medskip
\medskip
\newpage

\section{$@%a%K%e!<(J}
\label{menus}
\medskip
\par
tgif+ $@$G:n6H$r9T$&$H$-$O!$!V%a%$%s%a%K%e!<!W$b$7$/$O!V(JChoice $@%&%#%s%I(J
$@%&!W$rMQ$$$F9T$$$^$9!%$J$*!$$[$H$s$I$NA`:n$O!$%-!<$K3d$jEv$F$i$l$F$*$j!$(J
$@%-!<%\!<%I$rMQ$$$F9T$&$3$H$b$G$-$^$9!%(J
\medskip

\subsection{$@%a%$%s%a%K%e!<(J}
\medskip
\par
$@%a%$%s%a%K%e!<$GI=<($5$l$k%"%$%F%`$,!$(Jtgif+ $@$G9T$($kA`:n$rI=$7$^$9!%%a(J
$@%$%s%a%K%e!<$N%"%$%F%`$K$O!$<!$N$b$N$,$"$j!$$=$l$>$l%5%V%a%K%e!<$r;}$C(J
$@$F$$$^$9!%(J
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|}
\hline
$@%"%$%F%`L>(J & $@35MW(J                              \\
\hline\hline
Mode       & $@A`:n%b!<%I!$IA2h%b!<%I$N;XDj!&JQ99(J\\
\hline
File       & $@%U%!%$%k$K4X$9$kA`:n$r9T$&(J        \\
\hline
Edit       & $@%(%G%#%C%H$K4X$9$kA`:n$r9T$&(J      \\
\hline
TextStyle  & $@%F%-%9%H%9%?%$%k$NA*Br!&JQ99$r9T$&(J\\
\hline
TextSize   & $@%F%-%9%H%5%$%:$NA*Br!&JQ99$r9T$&(J  \\
\hline
Font       & $@%U%)%s%H$NA*Br!&JQ99$r9T$&(J        \\
\hline
FontDPI    & $@%U%)%s%H$N(JDPI $@$NA*Br!&JQ99$r9T$&(J  \\
\hline
Layout     & $@%l%$%"%&%H$K4X$9$kA`:n$r9T$&(J      \\
\hline
Arrange    & $@%*%V%8%'%/%H$NG[CVEy$NA`:n$r9T$&(J  \\
\hline
Fill       & $@JD6JLL$NEI$j$D$V$7$r9T$&(J          \\
\hline
LineStyle  & $@@~$N%9%?%$%k$K$NA*Br!&JQ99$r9T$&(J  \\
\hline
Pen        & $@@~J,$dJ8;zNs$rEI$k(J                \\
\hline
Color      & $@?'$NA*Br!&JQ99$r9T$&(J              \\
\hline
Special    & $@$=$NB>=t!9$NA`:n$r9T$&(J            \\
\hline
\end{tabular}\\
\end{center}
\medskip

\subsection{$@%5%V%a%K%e!<(J}
\medskip
\par
$@0J2<!$%a%$%s%a%K%e!<$N%5%V%a%K%e!<$NFbMF$K$D$$$F!$<gMW$J$b$N$N%-!<%P%$%s%I$H35MW$r4JC1$K@bL@$7$^$9!%(J
\medskip
\par
$@$J$*!$%5%V%a%K%e!<$N%"%$%F%`$,H?E>I=<($5$l$F$$$k$b$N$O!$%+%l%s%H$rI=$7$^$9!%(J
$@Nc$($P!$%*%V%8%'%/%H!J;M3Q7AEy!K$rA*Br$7$F!V(JLineStyle $@%a%K%e!<!W$rI=<($9$k$H!$(J
$@$=$N%*%V%8%'%/%H$N!V(JLineStyle $@%a%K%e!<!W$K4X$9$k%+%l%s%H$N>uBV$,H?E>I=<($5$l$^$9!%(J
\medskip
\medskip

\subsubsection{Mode}
\medskip
$@?^(J~\ref{menuWins}~ $@$r;2>H$7$F2<$5$$!%(J

        \begin{figure}[htbp]
           \begin{center}
                \epsfile{file=FIGURE/menuWins.eps}
                \caption{Mode, LineStyle, Fill, Pen $@%a%K%e!<(J}
                \label{menuWins}
           \end{center}
        \end{figure}
\medskip

\subsubsection{File}
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|l|}
\hline
$@%"%$%F%`L>(J    & $@%-!<(J & $@35MW(J                              \\
\hline
New           & \verb+^+N   & $@?7$7$$(J .obj $@%U%!%$%k$N%*!<%W%s(J    \\
Open          & \verb+^+O   & $@0MB8$N(J .obj $@%U%!%$%k$N%*!<%W%s(J    \\
Save          & \verb+^+S   & .obj $@%U%!%$%k$XJ]B8(J               \\
SaveNew       & \verb+^+\#S  & $@?7$7$$L>A0$N(J .obj $@%U%!%$%k$XJ]B8(J  \\
Import        & \multicolumn{1}{|c|}{ \#P } & $@JL$N(J.obj $@%U%!%$%k$NA^F~(J           \\
ImportXBitmap & \verb+^+\#.  & .xbm $@%U%!%$%k$NA^F~(J               \\
...           &      & $@!JN,!K(J                            \\
Print         & \verb+^+P   & $@%W%j%s%H%"%&%H(J                    \\
PrintWithCmd  & \verb+^+$-$   & $@;XDj$7$?%3%^%s%I$G$N%W%j%s%H%"%&%H(J\\
...           &      & $@!JN,!K(J                            \\
Quit          & \verb+^+Q   & tgif+ $@$N=*N;(J                      \\
\hline
\end{tabular}\\
\end{center}
\medskip

\subsubsection{Edit}
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|l|}
\hline
$@%"%$%F%`L>(J    & $@%-!<(J & $@35MW(J                              \\
\hline
Redraw        & \verb+^+R   & $@2hLL$N:FIA2h(J                      \\
Duplicate     & \verb+^+D   & $@%*%V%8%'%/%H$NJ#@=(J                \\
Delete        & \verb+^+X   & $@%*%V%8%'%/%H$N:o=|(J                \\
SelectAll     & \verb+^+A   & $@$9$Y$F$N%*%V%8%'%/%H$rA*Br(J        \\
Undo          & \#U   & $@A`:n$N(JUNDO                        \\
DeletePoint   & \verb+^+\#D  & $@%]%j%4%s!$%]%j%i%$%s$N@)8fE@$N:o=|(J\\
AddPoint      & \verb+^+\#A  & $@%]%j%4%s!$%]%j%i%$%s$N@)8fE@$NDI2C(J\\
Copy          & \verb+^+Y   & $@%+%C%H%P%C%U%!$X$N%3%T!<(J          \\
Paste         & \verb+^+\#Y  & $@%+%C%H%P%C%U%!$+$i$N%Z!<%9%H(J      \\
InvertXBitmap & \verb+^+\#F  & $@A*Br$7$?%S%C%H%^%C%W$NH?E>(J        \\
SameAs        & \#SP  & $@$"$k%*%V%8%'%/%H$NB0@-$rJL$N%*%V%8(J\\
              &      & $@%'%/%H$K%3%T!<$9$k!%(J              \\
\hline
\end{tabular}\\
\end{center}
\medskip

\subsubsection{TextStyle}
\medskip
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|l|}
\hline
$@%"%$%F%`L>(J & $@%-!<(J & $@35MW(J                       \\
\hline
Roman      & \verb+^+\#O  & $@A*Br$7$?J8;zNs$r(J Roman $@BN$K(J\\
Bold       & \verb+^+\#B  & \multicolumn{1}{|r|}{                  Bold $@BN$K(J}\\
Italic     & \verb+^+\#T  & \multicolumn{1}{|r|}{                Italic $@BN$K(J}\\
BoldItalic & \verb+^+\#P  & \multicolumn{1}{|r|}{            BoldItalic $@BN$K(J}\\
...        &      &                            \\
Left       & \verb+^+\#L  & $@J8;zNs$r:84s$;$K(J           \\
Center     & \verb+^+\#C  & $@J8;zNs$rCf1{B7$($K(J         \\
Right      & \verb+^+\#R  & $@J8;zNs$r1&4s$;(J             \\
\hline
\end{tabular}\\
\end{center}
\medskip
\medskip

\subsubsection{TextSize}
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|l|}
\hline
$@%"%$%F%`L>(J & $@%-!<(J & $@35MW(J                          \\
\hline
8          & \#0   & $@A*Br$7$?J8;zNs$N%5%$%:$r(J 8 $@$K(J \\
10         & \#1   & $@A*Br$7$?J8;zNs$N%5%$%:$r(J 10 $@$K(J\\
12         & \#2   & $@A*Br$7$?J8;zNs$N%5%$%:$r(J 12 $@$K(J\\
14         & \#3   & $@A*Br$7$?J8;zNs$N%5%$%:$r(J 14 $@$K(J\\
18         & \#4   & $@A*Br$7$?J8;zNs$N%5%$%:$r(J 18 $@$K(J\\
24         & \#5   & $@A*Br$7$?J8;zNs$N%5%$%:$r(J 24 $@$K(J\\
\hline
\end{tabular}\\
\end{center}
\medskip
\par
$@!V(JFontDPI$@!W$,!V(J100dpi$@!W(J $@$N>l9g$O!$!V(J11$@!$(J14$@!$(J17$@!$(J20$@!$(J25$@!$(J34$@!W$H$J$j$^$9!%(J
\medskip

\subsubsection{Font}
\medskip
~\\
\begin{center}
\begin{tabular}{|l|r|l|}
\hline
$@%"%$%F%`L>(J & $@%-!<(J & $@35MW(J                                   \\
\hline
Times      & $@$J$7(J & $@A*Br$7$?J8;zNs$N%U%)%s%H$r(J Times $@$K(J    \\
Courier    & $@$J$7(J & $@A*Br$7$?J8;zNs$N%U%)%s%H$r(J Courier $@$K(J  \\
Helvetica  & $@$J$7(J & $@A*Br$7$?J8;zNs$N%U%)%s%H$r(J Helvetica $@$K(J\\
NewCentury & $@$J$7(J & $@A*Br$7$?J8;zNs$N%U%)%s%H$r(J NewCentury$@$K(J\\
Symbol     & $@$J$7(J & $@A*Br$7$?J8;zNs$N%U%)%s%H$r(J Symbol $@$K(J   \\
Minchou    & $@$J$7(J & $@A*Br$7$?J8;zNs$N%U%)%s%H$r(J Minchou $@$K(J  \\
Gothic     & $@$J$7(J & $@A*Br$7$?J8;zNs$N%U%)%s%H$r(J Gothic $@$K(J   \\
\hline
\end{tabular}\\
\end{center}
\medskip

\subsubsection{FontDPI}
\medskip
~\\
\begin{center}
\begin{tabular}{|l|r|l|}
\hline
$@%"%$%F%`L>(J & $@%-!<(J & $@35MW(J                          \\
\hline
75 dpi     & $@$J$7(J & $@%U%)%s%H$N(J DPI $@$r(J 75 $@$K$9$k!%(J \\
100 dpi    & $@$J$7(J & $@%U%)%s%H$N(J DPI $@$r(J 100 $@$K$9$k!%(J\\
\hline
\end{tabular}\\
\end{center}
\medskip

\subsubsection{Layout}
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|l|}
\hline
$@%"%$%F%`L>(J  & $@%-!<(J & $@35MW(J                                \\
\hline
+Grid       & \#I   & $@%0%j%C%I4V3V$rA}$d$9(J                \\
$-$Grid       & \#D   & $@%0%j%C%I4V3V$r8:$i$9(J                \\
HideGrid    & \#G   & $@%0%j%C%IE@$rI=<($7$J$$(J              \\
SnapOff     & \verb+^+\#G  & $@%0%j%C%I$r8z$+$J$/$9$k(J              \\
ZoomIn      & \#Z   & $@2hLL$r3HBg$9$k(J                      \\
ZoomOut     & \#O   & $@2hLL$r=L>.$9$k(J                      \\
Landscape   & $@$J$7(J & $@;f$N%5%$%:$r(J Landscape $@$K$9$k(J       \\
Portrait    & $@$J$7(J & $@;f$N%5%$%:$r(J Portrait $@$K$9$k(J        \\
HighLand    & $@$J$7(J & $@;f$N%5%$%:$r(J HighLand $@$K$9$k(J        \\
HighPort    & $@$J$7(J & $@;f$N%5%$%:$r(J HighPort $@$K$9$k(J        \\
SlideLand   & $@$J$7(J & $@;f$N%5%$%:$r(J SlideLand $@$K$9$k(J       \\
SlidePort   & $@$J$7(J & $@;f$N%5%$%:$r(J SlidePort $@$K$9$k(J       \\
LaTeXFig    & \verb+^+\#X  & $@%W%j%s%H@h$r(J .ps $@%U%!%$%k(J $@"*(J        \\
            &      & .xbm $@%U%!%$%k(J $@"*(J .eps $@%U%!%$%k(J $@"*(J   \\
            &      & $@%W%j%s%?(J $@$K$9$k!%(J                   \\
...         &      &                                     \\
DotOnGrid   & \verb+^+/   & $@%0%j%C%IE@$N4V3V$r%0%j%C%I4V3V$K$9$k(J\\
ShowSize    & \#/   & $@%*%V%8%'%/%H$N%5%$%:$rI=<($9$k(J      \\
SaveWithEPS & $@$J$7(J & $@%U%!%$%k$N%;!<%V$9$k$H(J $@F1;~$K(J       \\
            &      & .eps $@$r@8@.$9$k%b!<%I$K$9$k(J         \\
\hline
\end{tabular}\\
\end{center}
\medskip

\subsubsection{Arrange}
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|l|}
\hline
$@%"%$%F%`L>(J      & $@%-!<(J & $@35MW(J                                  \\
\hline
Front           & \verb+^+F   & $@A*Br$7$?%*%V%8%'%/%H$rA0$K(J            \\
Back            & \verb+^+B   & $@A*Br$7$?%*%V%8%'%/%H$r8e$m$X(J          \\
Group           & \verb+^+G   & $@A*Br$7$?%*%V%8%'%/%H$r%0%k!<%W2=$9$k(J  \\
UnGroup         & \verb+^+U   & $@%0%k!<%W2=$r2r$/(J                      \\
AlignObjs       & \verb+^+L   & $@J#?t$N%*%V%8%'%/%H$I$&$7$r%"%i%$%s$9$k(J\\
AlignToGrid     & \verb+^+T   & $@%*%V%8%'%/%H$r%0%j%C%I$K%"%i%$%s$9$k(J  \\
FlipHorizontal  & \#H   & $@A*Br$7$?%*%V%8%'%/%H$r(Jy$@<4$G6@A|$K(J     \\
FlipVertical    & \#V   & $@A*Br$7$?%*%V%8%'%/%H$r(Jx$@<4$G6@A|$K(J     \\
RotateClockWise & \#W   & $@A*Br$7$?%*%V%8%'%/%H$r;~7W2s$j$K2sE>(J  \\
RotateCounter   & \#C   & $@A*Br$7$?%*%V%8%'%/%H$rH?;~7W2s$j$K2sE>(J\\
\hline
\end{tabular}\\
\end{center}
\medskip

\subsubsection{Fill, LineStyle, Pen}
\medskip
$@?^(J~\ref{menuWins}~ $@$r;2>H$7$F2<$5$$!%(J
\medskip

\subsubsection{Color$@!$(JSpecial}
\begin{nquote}{8ex}
\par
$@>JN,(J
\end{nquote}
\medskip
\medskip
\newpage

\section{$@F|K\8l(J}
\label{japanese}
\medskip

\subsection{$@F~NO(J}
\medskip
\par
tgif+ $@$X$NF|K\8lF~NO$O!$<!$N(J3$@$D$NJ}K!$G9T$&$3$H$,$G$-$^$9!%(J
\medskip
\begin{nquote}{4ex}
\begin{itemize}
\item nemacs $@$+$i$N(J copy \& paste
\item X $@%&%#%s%I%&$NF|K\8lF~NO%/%i%$%"%s%H(J $@!J(Jkinput$@!K$rMQ$$$F(J
\item .obj $@%U%!%$%k$rD>$9(J
\medskip
\end{itemize}
\end{nquote}
\par
$@$J$*!$(Jkterm $@$+$i$N(J copy \& paste $@$O!$%5%]!<%H$7$F$$$^$;$s!J(JX11R3 $@$N(J
kterm $@$+$i$O2DG=$G$9!K(J
\footnote{kterm $@$+$i$N(J copy \& paste $@$O!$(Jnemacs  $@$,(J ICCCM  $@BP1~$5$l$?$i9T$&$D$b$j$G$9(J}$@!%(J
\medskip

\subsubsection{nemacs $@$+$i$N(J copy \& paste}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item X $@%&%#%s%I%&$N%/%i%$%"%s%H$H$7$F5/F0$7$F$$$k(J nemacs $@$G$N(J X $@%&%#%s%I%&$N5!G=$r(J
$@MxMQ$7$?(J copy $@A`:n$rMQ$$$F!$(Jnemacs $@%P%C%U%!Fb$NFbMF$r%3%T!<$9$k!%(J
\item tgif+ $@$G(J $@!V(JEdit $@%a%K%e!<!W$N!V(JPaste$@!W$rA*Br$9$k$3$H$G!$(Jtgif+ $@Fb$KF|(J
$@K\8l$rF~NO$G$-$^$9!%(J
\item $@J8;z2=$1$7$?;~$O!$$=$NJ8;zNs$rA*Br$7$F!$%U%)%s%H$r!V(JMinchou$@!W$K$7$F2<$5$$!%(J
\medskip
\end{enumerate}
}
\par
nemacs $@$G$N(J copy $@A`:n$O!$!V%3%s%H%m!<%k!W$r2!$7$J$,$i!$%^%&%9$N!V:8%\%?%s!W$r2!(J
$@$7$FN%$9$^$G$NNN0h$,%3%T!<$5$l$^$9!J%_%K%P%C%U%!$K!$!V(JCopyed$@!W$H$$$&%a(J
$@%C%;!<%8$,$G$^$9!%?^(J~\ref{tgifAndNemacs}~$@;2>H!K!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/tgif+Nema.ps,hscale=0.75,vscale=0.75}
%       \epsfile{file=FIGURE/tgif+Nema.eps,hscale=0.75,vscale=0.75}
        \caption{tgif+ $@$H(J nemacs}
        \label{tgifAndNemacs}
   \end{center}
\end{figure}

nemacs $@$G(J copy \& paste $@$r$9$k>l9g$K$O!$!V(Jx-mouse.el$@!W$H!V(Jx11-cut-paste.el$@!W(J
$@$rM=$a%m!<%I$7$F$*$/I,MW$,$"$j$^$9(J\footnote{.emacs $@$K(J(load "x-mouse") $@$H(J
(load "x11-cut-paste") $@$r2C$($F$*$$$F2<$5$$!%(J}$@!%(J
\medskip

\subsubsection{$@$+$J4A;zJQ49MQ$N(J X $@%/%i%$%"%s%H$G$"$k(J kinput $@$K$h$kF~NO(J}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item tgif+ $@$G!$J8;zF~NO%b!<%I!J(JMode $@%a%K%e!<$G!$!V(JT$@!W$rA*Br$9$k!K$K$9$k!%(J
\item $@!V(JCanvas $@%&%#%s%I%&!W$r%/%j%C%/$9$k!%(J
\item $@!V%7%U%H(J + $@%9%Z!<%9!W$G(J kinput $@$N$+$J4A;zJQ49%&%#%s%I%&$,I=<($5$l$k!%(J
\item $@!V(JTitle $@%&%#%s%I%&!W$K!V(J*KINPUT*$@!W$HI=<($5$l$^$9!%(J
\end{enumerate}
}
 $@?^(J~\ref{tgifAndKinput}~$@$K(J kinput $@$GF|K\8lF~NO$7$F$$$kMM;R$r<($7$^$9!%(J

\begin{figure}[htbp]
   \begin{center}
%       \epsfile{file=FIGURE/tgif+Kinp.eps,hscale=0.6,vscale=0.6}
        \epsfile{file=FIGURE/tgif+Kinp.ps,hscale=0.6,vscale=0.6}
        \caption{tgif+ $@$H(J kinput}
        \label{tgifAndKinput}
   \end{center}
\end{figure}
\medskip
\par
$@0J9_$NA`:n$O!$(Jkinput $@$K=>$$!$%-!<F~NO$O%m!<%^;zJQ49$5$l!$!V(J\verb+^+J$@!W$G4A;zJQ49!$(J
$@!V(J\verb+^+L$@!W$G3NDj$7$^$9!%(J
\medskip
\par
$@$J$*!$(Jkinput $@$O!$(Jtgif+ $@$GF|K\8lF~NO$r(J
$@9T$&A0$K5/F0$7$F$*$/I,MW$,$"$j$^$9!J(Jkinput $@$N;H$$J}$K4X$7$F$O!$!V(J
$@IUO?(J~\ref{kinput}~kinput $@$N;H$$J}!W;2>H!K!%(Jkinput $@$N=*N;$b!V%7%U%H(J + $@%9%Z!<%9!W$G$9!%(J
\medskip
\par
kinput $@$N$+$J4A;zJQ49%&%#%s%I%&$,I=<($5$l$F$$$k$H!$(Jtgif+ $@$X$N$9$Y$F$N(J
$@%-!]F~NO$,(J kinput $@$K<h$i$l$F$7$^$&$N$GCm0U$,I,MW$G$9(J
$@!J!VIUO?(J~\ref{kinputNotice}~$@Cm0U!W;2>H!K(J\footnote{$@$3$l$O!$<!%P!<%8%g%s$N(J kinput $@$G$O!$JQ$o$k$_$?$$$G$9!%(J}.
\medskip

\paragraph{kinput $@$N<+F05/F05!G=(J}~\\
\label{autoKinput}
\medskip
\par
$@$$$A$$$A!V%7%U%H(J+$@%9%Z!<%9!W$rF~NO$9$k$N$,LLE]$J%f!<%6$K$*4+$a$N5!G=$G(J
$@$9!%(J
\medskip
\par
$@!V(J$-$ak$@!W$^$?$O!V(JTgif*AutoKinput: on$@!W$r;XDj$9$k$H!$0J2<$N>l9g$K!$(Jkinput $@$H(J
$@<+F0E*$K@\B3$7$^$9!%$^$?!$(Jkinput $@$,5/F0$5$l$F$$$J$$>l9g$K$O!$(J
$@5/F0$7$F$+$i@\B3$7$^$9(J
\footnote{$@$3$N>l9g!$(J
kinput $@$N5/F0$O!$!V(Jtgif+ $@$,F0$$$F$$$k%[%9%H!W$G9T$J$o$l$^$9!%(J
$@$G$9$+$i!$(Jkinput $@$,%$%s%9%H!<%k$5$l$F$$$J$$%[%9%H$G;H$C$F$$$k(J tgif+
$@$G!$$3$N%*%W%7%g%s$r;H$&>l9g$K$O!$$"$i$+$8$a(J kinput $@$rJL$N!V(Jkinput $@$,%$%s%9%H!<%k$5$l$F$$$k%[%9%H!W$G5/F0$7$F$*$/I,MW$,$"$j$^$9(J.}$@!%(J
\medskip
\medskip
\begin{nquote}{4ex}
\begin{itemize}
\item $@%U%)%s%H$,!V(JMinchou$@!W$+!V(JGothic$@!W$G!$(J
\item $@J8;zNsF~NO%b!<%I$G!$J8;zNs$rF~NO$9$k$?$a$K!$$I$3$+$r%/%j%C%/(J
$@$7$?!%(J
\medskip
\end{itemize}
\end{nquote}
\par
$@$3$N5!G=$O!$!V(JSpecial $@%a%K%e!<!W$N!V(JAutoKinput ON$@!W$G$b@_Dj$,2DG=$G$9!%(J
\medskip
\par
$@Cm0U(J: $@$3$N5!G=$O!$;H$&A0$K0lEY$O!J(Jtgif+ $@$r;HMQ$9$k%[%9%H$G!K%7%'%k$+$i(J kinput $@$r5/F0$7$F@5>o$KF0$/(J
$@$3$H$r3NG'$7$F$+$i$*;H$$2<$5$$!J0lEY$b(J kinput $@$rF0$+$7$?$3$H$N$J$$4D6-(J
$@$G;H$&$N$O4m81$G$9!K!%(J
\medskip

\subsubsection{.obj $@%U%!%$%k$r=$@5$9$k(J}
\medskip
\par
$@>e5-$N(J2$@$D$NJ}K!$N$I$A$i$b;H$($J$$>l9g!$:G8e$N<jCJ$H$7$F!V(J.obj$@%U%!%$%k!W(J
$@$r%(%G%#%C%H$9$k$3$H$K$h$C$FF|K\8l$r?^$KF~$l$k$3$H$,$G$-$^$9!%$3$NJ}K!$O!$4JC1$J$N(J
$@$GCN$C$F$*$/$HJXMx$G$9!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@$^$:!$(Jtgif+ $@$r5/F0$7!$?^$r:n@.$7!$!V(JMinchou$@!W%U%)%s%H$rA*$s$G!$J8(J
$@;z$rF~$l$?$$>l=j$K!V%m!<%^;z!W$G!J1Q8l$G$b9=$$$^$;$s!K!$J8;z$rF~$l$F$*(J
$@$-$^$9!%Nc$($P!$!V(Jwatashiha, tgif+ gasukidesu$@!W$H$@$1F~$l$?$H$7$^$9!%(J
\medskip
\item $@$3$l$r!$!V(J.obj$@%U%!%$%k!W$K%;!<%V$7$^$9!J!V(JFile$@%a%K%e!<!W$+$i!V(J
Save$@!W$rA*Br$9$k!%$^$?$O!$!V(J\verb+^+S$@!W!K!%:#$3$N%U%!%$%k$r!V(Jbunsho$@!W$H$7$^$9!%(J
\medskip
\item $@JL$N%&%#%s%I%&$G!$E,Ev$J%(%G%#%?$r3+$$$F!$!V(Jbunsho.obj$@!W$rFI$_9~(J
$@$s$G2<$5$$!%<!$N$h$&$K$J$C$F$$$k$H;W$$$^$9!J(Jstate $@$N?t;z$,<c430[$J$k$+$b$7$l$^(J
$@$;$s$,5$$K$7$J$$$G2<$5$$!K!%(J
\medskip
\baselineskip=10pt
\begin{verbatim}
state(0,10,0,0,0,16,1,0,1,1,0,0,1,0,1,0,1,0,4,0,0).
%
% @(#)$Header$
%
text('magenta',64,80,5,0,4,1,0,0,1,205,21,12,0,17,4,[
        "watashiha, tgif+ gasukidesu."]).
\end{verbatim}
\medskip
\item $@!V%m!<%^;z!W$NItJ,$r0J2<$N$h$&$K$+$J4A;zJQ49$7!$%U%!%$%k$r!V(JJIS $@$+(J EUC$@!W%3!<%I$G%;!<%V$7$F!$:F$S!$(J
tgif+ $@$GFI$_9~$s$G2<$5$$!J!V(JFile $@%a%K%e!<!W$N!V(JOpen$@!W!$$^$?$O!$!V(J\verb+^+O$@!W!K!%(J
\medskip
\baselineskip=10pt
\begin{verbatim}
state(0,10,0,0,0,16,1,0,1,1,0,0,1,0,1,0,1,0,4,0,0).
%
% @(#)$Header$
%
text('magenta',64,80,5,0,4,1,0,0,1,205,21,12,0,17,4,[
        "$@;d$O(J, tgif+ $@$,9%$-$G$9!%(J"]).
\end{verbatim}
\medskip
\item $@!V%m!<%^;z!W$NItJ,$,!$4A;z$K$J$C$F$$$k$H;W$$$^$9!%$b$7!$J8;z2=$1(J
$@$7$F$$$k$h$&$G$7$?$i!$$=$N%U%)%s%H$rA*Br$7$F!$%U%)%s%H$r!V(JMinchou$@!W$K(J
$@$7$F8+$F2<$5$$!%(J
\medskip
\end{enumerate}
}

\subsection{$@I=<((J}
\medskip

\subsubsection{$@I=<($K;H$o$l$k%U%)%s%H(J}
\medskip
\par
$@F|K\8l$NI=<($K;H$&%U%)%s%H$O!$<!$N$h$&$JM%@hEY$GC5$5$l8+$D$+$C$?$b$N$r(J
$@;H$$$^$9(J\footnote{$@!VIUO?(J~\ref{howToSetKanji}~.Xdefaults $@$G$N4A;z%U%)%s%H$N;XDj$K$D$$$F!W(J
$@$b;2>H$7$F2<$5$$!%$?$@$7!$%5%$%:(J18 $@$OFC<l$G$9!%$3$l$K4X$7$F$b(J
$@!VIUO?(J~\ref{howToSetKanji}~.Xdefaults $@$G$N4A;z%U%)%s%H$N;XDj$K$D$$$F!W(J
$@$r;2>H$7$F2<$5$$(J}$@!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item \verb+~+/.Xdefaults $@$G;XDj$5$l$?%U%)%s%H(J
\item tgif+ $@$N%G%U%)%k%H$N%U%)%s%H(J
\item k14
\medskip
\end{enumerate}
}
\par
$@Nc$($P!$%5%$%:(J 24 $@$NL@D+BN$N>l9g$O!$<!$N$h$&$K$J$j$^$9!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item \verb+~+/.Xdefaults $@$N(J $@!V(JTgif*KanjiMinchou24:$@!W$K@_Dj$5$l$F$$$k%U%)%s%H(J
\item tgif+ $@$N%G%U%)%k%H!J(Jneck24$@!K(J
\item k14
\medskip
\end{enumerate}
}
\par
$@$3$3$G!$%U%)%s%H$rC5$7$F!J(J3$@!K$K$J$C$?>l9g(J k14 $@$GI=<($7$F!$$=$N2s$j$KJ8;zOH$,I=<($5$l$k(J
$@$3$H$K$J$j$^$9(J
$@!J!V(J\S~\ref{displayKanji}~$@<B:]$NI=<(Iw7J!W(J $@;2>H!K!%(J
\medskip
\par
$@$^$?!$%4%7%C%/BN$N>l9g$O!$<!$N$h$&$K$J$j$^$9!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item \verb+~+/.Xdefaults $@$N(J $@!V(JTgif*KanjiGothic24:$@!W$K@_Dj$5$l$F$$$k%U%)%s%H(J
\item \verb+~+/.Xdefaults $@$N(J $@!V(JTgif*KanjiMinchou24:$@!W$K@_Dj$5$l$F$$$k%U%)%s%H(J
\item tgif+ $@$N%G%U%)%k%H!J(Jneck24$@!K(J
\item k14
\medskip
\end{enumerate}
}
\par
$@$?$@$7!$!V(JTgif*KanjiGothic24:$@!W$N%U%)%s%H0J30$N>l9g$K$O!$L@D+BN$N%U%)%s%H$r(J
$@;H$&$3$H$K$J$j$^$9$+$i!$I=<($9$k;~$K%4%7%C%/BN$K8+$($k$h$&>/$7B@$a$K(J
$@I=<($7$^$9!%(J(4)$@$,A*$P$l$?>l9g$O!$L@D+BN$HF1MM$KJ8;zOH$,I=<($5$l(J
$@$^$9!%(J
\medskip

\subsubsection{$@<B:]$NI=<(Iw7J(J}
\label{displayKanji}
\medskip
\par
X11R4 $@$^$G$O!$%G%U%)%k%H$G$O!$4A;z%U%)%s%H$O!V(Jk14$@!W0l<o$7$+%5%]!<%H$5$l$F$$$^$;$s!%(J
$@$3$l$O!$%G%U%)%k%H$N4D6-$G$O!$2hLL$K!V(Jk14$@!W$N%5%$%:$N%U%)%s%H$7$+I=<($G$-$J$$$H(J
$@$$$&$3$H$r0UL#$7$F$$$^$9!J$3$l$O!$(Jtgif+ $@$N%F%-%9%H%5%$%:!V(J14$@!W$KEv$?$j$^$9!K!%(J
\medskip
\par
$@$H$3$m$,!$(Jtgif+ $@$O!V(J8$@!$(J10$@!$(J12$@!$(J14$@!$(J18$@!$(J24$@!W$H$$$&(J6$@<oN`$N%U%)%s%H$r(J
$@%5%]!<%H$7$F$$$^$9!%$D$^$j!V(J14$@!W0J30$N%U%)%s%H$r2hLL$G!$E,Ev$KI=<($G$-$k$h$&$K(J
$@$7$J$/$F$O$J$i$J$$$o$1$G$9!%(J
tgif+ $@$G$O!$$=$N%5%$%:$N%U%)%s%H$,$J$$>l9g$K$O!$$H$j$"$($:(J k14 $@$GI=<($7$F(J
$@<~$j$K<B:]$N%5%$%:$rI=$9!VJ8;zOH!W$rI=<($9$k$h$&$K$7$F$$$^$9(J
\footnote{$@K\Ev$O!$%U%)%s%H$r%S%C%H3HBg$9$k$J$j!$=L>.$9$k$J$j$7$FI=<($9$l$P$$(J
$@$$$N$OJ,$+$C$F$$$k$N$G$9$,!$$^$"!$$H$j$"$($:!$$=$N$&$A$$$m$$$m$J%5%$%:$N(J
$@%U%)%s%H$,%5%]!<%H$5$l$k$G$7$g$&$7!%!%(J
.}.
\medskip
\par
$@$3$3$G$O!$!V(JFake $@%U%)%s%H5!G=!W$H$$$&J8;zNs$rI=<($7$F$_$^$9!%>e$,!VL@D+BN!W(J
$@$G!$2<$,!V%4%7%C%/BN!W$G$9!%:8B&$O!V%5%$%:(J 24$@!W$N%U%)%s%H$,(J X $@%5!<%P$K(J
$@$"$k>l9g!$1&B&$,$J$$>l9g$NNc$G$9!J?^(J~\ref{fakeFont} $@;2>H!K!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/fakeFont.eps}
        \caption{$@4A;z$NI=<(Iw7J(J}
        \label{fakeFont}
   \end{center}
\end{figure}
\medskip
\par
$@$3$l$O!V%4%7%C%/BN!W$N%U%)%s%H$,$J$$>l9g$G$9$N$G!$!V%4%7%C%/BN!W$O!VL@D+BN!W$N%U%)%s%H$r>/$7B@$a$KI=<($7$F$$$^$9!%(J
$@%U%)%s%H$,$J$$>l9g$K$O!$?^$K<($7$?$h$&$K!$<B:]$N%5%$%:$KAjEv$9$kJ8;zOH(J
$@$,<~$j$KI=<($5$l$^$9!%(J
\medskip

\subsubsection{$@I=<($N@bL@$,J,$+$i$J$+$C$??M$K!%!%!%(J}
\medskip
\par
$@$9$_$^$;$s!%@bL@$,2<<j$G!%!%!%(J:$-$(
\medskip
\par
$@$^$"!$4pK\E*$K%G%U%)%k%H$G(J tgif+ $@$r;H$C$F$$$F2?$bLdBj$O$J$$$N$G$9$,!$(J
$@LdBjE@$H$7$F$O!$(J
\medskip
\begin{itemize}
\item X $@%&%#%s%I%&$K$O!V(Jk14$@!W$7$+4A;z%U%)%s%H$O%5%]!<%H$5$l$F$$$J$$!%(J
\medskip
\end{itemize}
\par
$@$H$$$&$3$H$G!$(Jtgif+ $@$G$O!$%5%$%:!V(J14$@!W0J30$N%U%)%s%H$b;H$&$N$G!$(J
$@$I$&$K$+$7$J$$$H$$$1$J$$$H$$$&Lu$G$9!%(J
\medskip
\par
$@$=$3$G!$(Jtgif+ $@$G$O!$$H$j$"$($:!$(J
\medskip
\begin{itemize}
\item $@$J$$%U%)%s%H$O!$(Jk14 $@$GI=<($7$F$=$N<~$j$KJ8;zOH$rI=<($9$k$h$&$K$7$F$$$F!$(J
\item $@$^$?!$%4%7%C%/BN$b$J$$$N$G!VL@D+BN!W$GBeMQ$7$F$^$9$h!%(J
\medskip
\end{itemize}
\par
$@$H$$$&$3$H$G$9!J$3$l$G$bJ,$+$i$J$+$C$?>l9g$O!%!%!%$&!<$s!$5$$K$7$J$$$G2<$5$$(J:$-$P$@!K!%(J
\medskip
\newpage

\section{$@$5$F!$$3$&$$$&>l9g$K$O!%!%!%(J}
\label{inCaseOf}
\medskip
\par
$@NI$/J9$+$l$k<ALd$K$D$$$F$^$H$a$F$*$-$^$9!%(J
\medskip
\par
$@0J2<$N@bL@$G!$%-!<%P%$%s%I$O!V(J\verb+^+A$@!W$O!V%3%s%H%m!<%k$r2!$7$J$,$i(J A $@$r2!$9!W!$(J
$@!V(J\#A$@!W$O!V%a%?%-!<$r2!$7$J$,$i(J A $@$r2!$9!W!$$^$?!$!V(J\verb+^+\#A$@!W$O!V%3%s%H%m!<%k$H%a%?%-!<$r2!$7$J$,$i(J A $@$r2!$9!W$rI=$7$^$9!J%"%k%U%!%Y%C%H$O8+0W$5$N$?$aBgJ8;z$G=q$-$^$9$,!$F~NO$O>.J8;z$G$9!K!%(J
\medskip

\subsection{.obj $@%U%!%$%k$rFI$_9~$_$?$$(J.}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@!V(JFile$@%a%K%e!<!W$N!V(JOpen$@!W$rA*Br$7$F2<$5$$!%(J
$@?^(J \ref{namePwin} $@$N$h$&$J%&%#%s%I%&$,I=<($5$l!$!V(JOpen$@!W$9$k(J
$@%U%!%$%kL>$rJ9$$$F$-$^$9!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/openPwin.eps}
        \caption{open $@$9$k%U%!%$%kL>$r?R$M$F$$$k%&%#%s%I%&(J}
        \label{namePwin}
   \end{center}
\end{figure}
\medskip
\begin{itemize}
\item $@$3$3$G!$H?E>I=<($5$l$F$$$k$b$N$,!$%+%l%s%H$NA*Br$rI=$7$^$9!%(J
$@!V:8%\%?%s!W$r%@%V%k%/%j%C%/$+!$$^$?$O!$%j%?!<%s$G!V3NDj!W$7$^$9!%(J
\medskip
\item $@%U%!%$%kL>$NI=<($5$l$F$$$k%&%#%s%I%&$G!$!V:8%\%?%s!W$r%/%j%C(J
$@%/$9$k$H!VA*Br!W$G$-$^$9!%(J
\medskip
\item $@%G%#%l%/%H%j$r0\F0$9$k>l9g$O!$%-!<%\!<%I$+$i%Q%9L>$rF~NO$9$k$+!$(J
$@!V:8%\%?%s!W$G%G%#%l%/%H%j$rA*Br!$3NDj$7$D$D0\F0$7$^$9!%(J
\medskip
\item $@0lHV2<$N!V(JSETDIR$@!W$rA*Br$9$k$H!$8=:_$N%G%#%l%/%H%j$r5-21$7$^$9!%(J
\medskip
\end{itemize}
\item[$@%-!<%P%$%s%I(J:]\verb+^+O
\medskip
\end{description}

\subsection{$@JL$N(J .obj $@%U%!%$%k$rA^F~$7$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@!V(JFile$@%a%K%e!<!W$N!V(JImport$@!W$rA*Br$7$F2<$5$$!%(J
$@!V(JOpen$@!W;~$HF1MM$N%&%#%s%I%&$,3+$+$l$^$9!%A`:n$OF1$8$G$9!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\#P
\medskip
\end{description}

\subsection{.xbm $@%U%!%$%k$rA^F~$7$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@!V(JFile$@%a%K%e!<!W$N!V(JImportXBitmap$@!W$rA*Br$7$F2<$5$$!%(J
$@!V(JOpen$@!W;~$HF1MM$N%&%#%s%I%&$,3+$+$l$^$9!%A`:n$OF1$8$G$9!%(J
\medskip
\par
$@C"$7!$!V(J.xbm $@%U%!%$%k!W$H%G%#%l%/%H%jL>$7$+I=<($5$l$^$;$s!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\verb+^+\#.$@!J%T%j%*%I$G$9$h(J:$-$P$@!K(J
\medskip
\end{description}

\subsection{$@8e$m$K$"$k%*%V%8%'%/%H$rA0$K;}$C$F$-$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@A0$K;}$C$F$-$?$$%*%V%8%'%/%H$rA*Br$7$F!V(JArrange $@%a%K%e!<!W$N!V(JFront$@!W(J
        $@$rA*Br$7$F2<$5$$!J?^(J~\ref{toForward}~$@;2>H!K!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/toForward.eps,hscale=0.75,vscale=0.75}
        \caption{$@%*%V%8%'%/%H$rA0$K;}$C$F$/$k(J}
        \label{toForward}
   \end{center}
\end{figure}
\medskip
\item[$@%-!<%P%$%s%I(J:]\verb+^+F$@!JA0$K!K!$(J\verb+^+B$@!J8e$m$X!K(J
\medskip
\end{description}

\subsection{$@?^7A$r3HBg!&=L>.$7$F8+$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@!V(JLayout$@%a%K%e!<!W$N!V(JZoomIn$@!W!V(JZoomOut$@!W$rA*Br$7$F2<$5$$!%(J $@?^$,(J2$@G\!J(J1/2$@G\!K$K$J$j$^$9!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\#Z$@!J3HBg!K!$(J \#O$@!J=L>.!K(J
\medskip
\end{description}

\subsection{$@%0%j%C%I4V3V$rJQ$($?$$!%(J}
\label{xyGrid}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
tgif+ $@$O!$%G%U%)%k%H$G$O%*%V%8%'%/%H$NIA2h$dA`:n$O!$%0%j%C%IC10L$G(J
$@9T$J$o$l$^$9!%Nc$($P!$;M3Q7A$rIA$$$F%j%5%$%:$9$k$H!$$"$k0lDj$NI}$G$7$+(J
$@Bg$-$/$G$-$^$;$s!%$3$l$O!$(Jtgif+ $@$N%0%j%C%I5!G=$K$h$k$b$N$G$9!%(J
\medskip
\par
$@$3$N4V3V!J%0%j%C%I4V3V!K$rJQ$($k$K$O!$!V(JLayout$@%a%K%e!<!W$N!V(J+Grid$@!W!V(J$-$Grid$@!W$rA*Br$7$F2<$5$$!%%0%j%C%I(J
$@4V3V$,(J 2$@G\!J(J1/2 $@G\!K$K$J$j$^$9!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\#I$@!J(J2$@G\$K$9$k!K!$(J\#D$@!J(J1/2 $@G\$K$9$k!K(J
\medskip
\end{description}

\subsection{$@%0%j%C%I$r8z$+$J$/$7$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~

        $@>e$N!V(J\S~\ref{xyGrid}~$@%0%j%C%I4V3V$rJQ$($?$$!W$G@bL@$7$^$7$?$,!$(J
        tgif+ $@$O!$%G%U%)%k%H$G$O%*%V%8%'%/%H$NA`:n$O!$%0%j%C%IC10L$G$7$+(J
        $@$G$-$^$;$s!%$3$N>uBV$r!V%0%j%C%I$,8z$$$F$$$k!W$H8@$$$^$9!%(J
\medskip
\par
$@!V%0%j%C%I$r8z$+$J$/$9$k!W$K$O!$!V(JLayout$@%a%K%e!<!W$N!V(JSnapOff$@!W$rA*Br$7$F2<$5$$!%$b$I$9>l9g$O!$!V(JSnapOn$@!W$rA*Br$7$F2<$5$$!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\verb+^+\#G $@!J%H%0%k!K(J
\medskip
\end{description}

\subsection{$@IA2h%*%V%8%'%/%H$r%0%j%C%I$K9g$o$;$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@IA2h%*%V%8%'%/%H$rA*Br$7$F!$!V(JArrange$@%a%K%e!<!W$N!V(JAlignToGrid$@!W(J
$@$rA*Br$7$F2<$5$$!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\verb+^+T
\medskip
\end{description}

\subsection{$@IA2h%*%V%8%'%/%H$r(J 90 $@EY2sE>$7$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@IA2h%*%V%8%'%/%H$rA*Br$7$F!$!V(JArrange$@%a%K%e!<!W$N!V(JRotateClockWise$@!W(J
$@$rA*Br$7$F2<$5$$!%H?;~7W2s$j$K2s$9>l9g$O!$!V(JRotateCounter$@!W!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\#W$@!J;~7W2s$j!K!$(J\#C$@!JH?;~7W2s$j!K(J
\medskip
\end{description}

\subsection{$@%3%T!<(J\&$@%Z!<%9%H$7$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@%3%T!<$7$?$$IA2h%*%V%8%'%/%H$rA*Br$7$F!$!V(JEdit$@%a%K%e!<!W$N!V(J
Copy$@!W$rA*Br$7$F2<$5$$!%$3$l$G!$%3%T!<%P%C%U%!$K%3%T!<$5$l$^$9!%(J
$@!V(JEdit$@%a%K%e!<!W$G!V(JPaste$@!W$r<B9T$7$F2<$5$$!%$J$*!$$3$l$O!$(J
tgif+ $@$rJ#?t8DF0$+$7$F!$$=$N4V$G$b%3%T!<(J\&$@%Z!<%9%H$G$-$^$9!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\verb+^+Y $@!J%3%T!<!K!$(J\verb+^+\#Y$@!J%Z!<%9%H!K(J
\medskip
\end{description}

\subsection{$@J#?t8D$N%*%V%8%'%/%H$r!$0l$D$N%*%V%8%'%/%H$H$7$F07$$$?$$!%(J}
\label{group}
\medskip
\begin{description}
\item[$@J}K!(J:]~
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@0l$D$K$7$F07$$$?$$%*%V%8%'%/%H$r$9$Y$FA*Br$7$F2<$5$$!%(J
\item $@!V(JLayout $@%a%K%e!<!W$+$i!V(JGroup$@!W$rA*Br$7$F2<$5$$!%(J
\item $@A*Br$5$l$F$$$?%*%V%8%'%/%H$,!$0l$D$N%*%V%8%'%/%H$K$J$j$^$9!%(J
\item $@$b$H$KLa$9;~$O!$!V(JUngroup$@!W$G$9!%(J
\medskip
\end{enumerate}
}
\item[$@%-!<%P%$%s%I(J:]\verb+^+G$@!J%0%k!<%W2=!K!$(J\verb+^+U$@!J%0%k!<%W2=$r2r$/!K(J
\medskip
\end{description}

\subsection{$@F1$8%*%V%8%'%/%H$rBt;3:n$j$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@IA2h%*%V%8%'%/%H$r!VA*Br!W$7$F!$!V(JEdit$@%a%K%e!<!W$N!V(JDup$@!W$rA*(J
$@Br$7$F2<$5$$!%J#@=$,:n$i$l$^$9!%(J
\medskip
\par
$@J#@=$r:n$k:]$K!$(Jtgif+ $@$O!$A0$N%*%V%8%'%/%H$H$N5wN%$rJ]$C$FJ#@=(J
$@$r:n$j$^$9!%$3$N@-<A$r$&$^$/MxMQ$9$k$HJXMx$G$9!%(J
\medskip
\par
$@Nc$($P!$8kHW$rIA$3$&$H;W$C$?>l9g(J:
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@8kHW$N%^%9$r0l8DIA$-!$A*Br$7$F!$J#@=$7$^$9!%(J
\item $@J#@=$5$l$?%*%V%8%'%/%H$rA0$N%*%V%8%'%/%H$N??2#$K0\F0$7$^$9!%(J
\item $@$=$&$7$F!$!V(J\verb+^+D$@!W$r7+$jJV$9$H!$8kHW$N%^%9L\$,<+F0E*$KIA$+$l$F$$$-$^$9(J
\end{enumerate}
}
 $@!J?^(J~\ref{ctrlD} $@;2>H!K!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\verb+^+D
\medskip
\end{description}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/ctrlD.eps,hscale=0.75,vscale=0.75}
        \caption{\^{}D $@$N;H$$J}(J}
        \label{ctrlD}
   \end{center}
\end{figure}
\medskip

\subsection{$@??1_$rIA$-$?$$!$@5J}7A$rIA$-$?$$(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\medskip
\par
$@$H$j$"$($:!$$h$$%$%s%?%U%'!<%9$G$O$"$j$^$;$s$,!$!V(JLayout$@!W%a%K(J
$@%e!<$+$i!V(JShowSize$@!W$rA*Br$9$k$HIA2h;~$K%*%V%8%'%/%H$N%5%$%:$,(J
$@I=<($5$l$^$9$N$G!$$=$l$r8+$J$,$iIA$$$F2<$5$$!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]\#/
\medskip
\end{description}

\subsection{$@IA2h%*%V%8%'%/%H$r(JY$@<4$KBP$7$F6@A|$K$7$?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@IA2h%*%V%8%'%/%H$rA*Br$7$F!$!V(JArrange $@%a%K%e!<!W$N!V(JFlipHorizontal$@!W(J
        $@$rA*Br$7$F2<$5$$!J?^(J~\ref{flip}~$@;2>H!K!%(JX$@<4$KBP$7$F$O!V(JFlipVertical$@!W(J.
\medskip
\item[$@%-!<%P%$%s%I(J:]\#H$@!J(JY$@<4!K!$(J\#V$@!J(JX$@<4!K(J
\medskip
\end{description}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/flip.eps}
        \caption{Y$@<4$KBP$9$k6@A|(J}
        \label{flip}
   \end{center}
\end{figure}
\medskip

\subsection{$@%]%j%4%s!$%]%j%i%$%s$N@)8fE@$rA}$d$7$?$$!J8:$i$7$?$$!K!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@@)8fE@$N?t$rJQ$($?$$%*%V%8%'%/%H$rA*Br$7$F2<$5$$!%(J
\item $@!V(JEdit$@%a%K%e!<!W$+$i!V(JAddPoint$@!W$+!V(JDeletePoint$@!W$rA*Br$7(J
$@$F2<$5$$!%(J
\item $@%+!<%=%k$N2#$K!V(JADD$@!W$+!V(JDEL$@!W$H$$$&J8;z$,I=<($5$l$^$9(J
$@!J?^(J~\ref{addPoint}~$@;2>H!K!%(J
\item $@A}$d$7$?$$>l9g$O!$@)8fE@$r!V:8%\%?%s!W$G2!$7$F%I%i%C%0$7(J
$@$F2<$5$$!%(J
\item $@8:$i$7$?$$>l9g$O!$$J$/$7$?$$@)8fE@$r!V:8%\%?%s!W$G%/%j%C(J
$@%/$7$F2<$5$$!%(J
\item $@$3$N:n6H$O!$!V1&%\%?%s!W$r2!$9$^$G7+$jJV$79T$($^$9!%(J
\medskip
\end{enumerate}
}
\item[$@%-!<%P%$%s%I(J:]\verb+^+\#A $@!JA}$d$9!K!$(J\verb+^+\#D $@!J8:$i$9!K(J
\medskip
\medskip
\end{description}
\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/addPoint.eps}
        \caption{$@%]%j%i%$%s$N@)8fE@$NDI2C(J}
        \label{addPoint}
   \end{center}
\end{figure}
\medskip

\subsection{$@;f$N%5%$%:$rJQ$($?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
tgif+ $@$G$O!$<!$N(J6$@<oN`$N;f$N%5%$%:$r%5%]!<%H$7$F$$$^$9!%(J
$@;f$r2#$K$7$?$$>l9g$d?^$dJ8;z$r3HBg$7$?$$>l9gEy$K;H$&$HJXMx$G$9!%(J
\medskip
\par
$@%5%$%:$NA*Br$O!$!V(JLayout$@%a%K%e!<!W$r;H$C$F2<$5$$!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]$@$J$7(J
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|}
\hline
\multicolumn{1}{|c|}{ $@L>A0(J    } & $@%5%$%:(J\\
\hline
Landscape & A4 $@2#(J \\
Portrait  & A4 $@=D(J \\
HighLand  & A3 $@2#(J \\
HighPort  & A3 $@=D(J \\
SlideLand & A5 $@2#(J \\
SlidePort & A5 $@=D(J \\
\hline
\end{tabular}\\
\end{center}
\medskip
\end{description}

\subsection{$@J8;zNs$r1_$NCf?4$KCV$-$?$$!JJ#?t$N%*%V%8%'%/%H$I$&$7$r%"%i%$%s$7$?$$!K!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\medskip
\par
$@!V(JChoice $@%&%#%s%I%&!W$K!$!V?eJ?J}8~!W!V?bD>J}8~!W$N%"%i%$%s%a%s%H$r(J
        $@D4@a$9$k%"%$%3%s$,$"$j$^$9!J?^(J~\ref{align} $@;2>H!K!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/align.eps}
        \caption{$@?eJ?!&?bD>J}8~$N%"%i%$%s(J}
        \label{align}
   \end{center}
\end{figure}
\medskip
\par
$@1&%\%?%s$G%/%j%C%/$9$k$H!$!V?eJ?J}8~!W$N%"%$%3%s$O!V(JL$@!$(JC$@!$(JR$@!$(J$-$$@!W!$(J
$@!V?bD>J}8~!W$N%"%$%3%s$O!V(JT$@!$(JM$@!$(JB$@!$(J$-$$@!W$HJQ$o$j$^$9!%$3$l$O!$$=$l$>$l(J
L$@!J(JLeft$@!K!$(JC$@!J(JCenter$@!K!$(JR$@!J(JRight$@!K!$(J$-$$@!J(JIgnore$@!K!$$^$?!$(JT$@!J(JTop$@!K!$(J
M$@!J(JMiddle$@!K!$(JB$@!J(JButtom$@!K!$(J$-$$@!J(JIgnore$@!K$rI=$7$^$9!%(J
\medskip
\par
$@$3$l$i$rE,Ev$KAH$_9g$o$;$F!$%*%V%8%'%/%H$rA*Br$7$F!$!V(JArrange
$@%a%K%e!<!W$N!V(JAlighObjs$@!W$rA*Br$7$F2<$5$$!%(J
\medskip
        $@Nc$($P!$J8;zNs$r1_$NCf?4$KCV$-$?$$>l9g$O!$?^(J~\ref{alignObjects} $@$N(J
        $@$h$&$K$9$l$PNI$$$G$7$g$&!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/alignObje.eps}
        \caption{$@%*%V%8%'%/%H$I$&$7$N%"%i%$%s(J}
        \label{alignObjects}
   \end{center}
\end{figure}
\medskip
\item[$@%-!<%P%$%s%I(J:]\verb+^+L
\medskip
\end{description}

\subsection{$@%G%U%)%k%H$N%U%)%s%H$r!V(JMinchou$@!W$K$7$?$$(J.}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
$@!V(JTgif*InitialFont: Minchou$@!W(J $@$r(J \verb+~+/.Xdefaults $@$K2C$($F2<$5$$!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]$@$J$7(J
\medskip
\end{description}

\subsection{kinput $@$N$+$J4A;zJQ49$N%-!<%P%$%s%I$rJQ$($?$$!%(J}
\medskip
\begin{description}
\item[$@J}K!(J:]~
\par
ccdef.kinput $@$r%[!<%`%G%#%l%/%H%j$K%3%T!<=$@5$7$F2<$5$$!%(J
$@=$@5$N;EJ}$O!$%U%!%$%k$r8+$l$P$J$s$H$+$J$k$H;W$$$^$9$,!$(J
kinput $@$N%^%K%e%"%k$b;2>H$7$F2<$5$$!%(J
\medskip
\item[$@%-!<%P%$%s%I(J:]$@EvA3$"$j$^$;$s(J:$-$P
\medskip
\end{description}
\newpage

\section{prtgif+ $@$N;H$$J}(J}
\label{prtgif+}
\medskip

\subsection{$@7A<0(J}
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
prtgif+ [-ps,-eps] [fileName...]
\end{verbatim}
\end{nquote}
\medskip

\subsection{$@5!G=@bL@(J}
\medskip
\par
prtgif+ $@$O!$:n@.$7$F$*$$$?(J tgif+ $@$GIA$$$??^!J(J.obj $@%U%!%$%k!K$r%W%j%s%H(J
$@%"%&%H!$$^$?$O!$!V(J.eps $@%U%!%$%k!W$r:n@.$9$k$?$a$N%3%^%s%I$G$9!%$3$N%3%^%s%I$O(J
X $@%&%#%s%I%&$rI,MW$H$7$^$;$s$N$G!$(JX $@%&%#%s%I%&$,;HMQ$G$-$J$$4D6-$G!$!V(J.obj
$@%U%!%$%k!W$r%W%j%s%H%"%&%H$9$k$N$KJXMx$G$9!%(J
\medskip
\par
prtgif+ $@$,2r<a$9$k%*%W%7%g%s0J30$N%*%W%7%g%s$O!$(Jlpr $@%3%^%s%I$KEO$j$^$9!%(J
\medskip

\subsection{$@%*%W%7%g%s(J}
\medskip
\begin{nquote}{1ex}
\baselineskip=10pt
\begin{verbatim}
-eps
       $@!V(J.eps $@%U%!%$%k!W$,@8@.$5$l$^$9!%(J
-ps
\end{verbatim}
\end{nquote}
\begin{nquote}{8ex}
\par
$@!V(J.ps $@%U%!%$%k!W$,@8@.$5$l$^$9!%(J
\end{nquote}
\medskip

\subsection{$@;HMQNc(J}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item tgif+ $@$G:n@.$7$?%U%!%$%k!V(Jfigure.obj$@!W$r!$!V(J.eps $@%U%!%$%k!W$K$9$k!%(J
\medskip
\baselineskip=10pt
\begin{verbatim}
% prtgif+ -eps figure
\end{verbatim}
\medskip
\item tgif+ $@$G:n@.$7$?%U%!%$%k!V(Jfigure.obj$@!W$r!$%W%j%s%?!J(Jnps1$@!K$K=PNO$9$k!%(J
\medskip
\baselineskip=10pt
\begin{verbatim}
% prtgif+ -Pnps1 figure
\end{verbatim}
\medskip
\end{enumerate}
}
\newpage
\appendix
\section{X DEFAULTS}
\label{xdefaults}
\medskip
\par
$@0J2<!$(J\verb+~+/.Xdefaults $@$K=q$1$kJQ?t$H$=$NFbMF$K$D$$$F=q$-$^$9!%$J$*!$(J
\verb+~+/.Xdefaults $@$N;XDj$N;EJ}$O!$(J
$@!VIUO?(J~\ref{dXdefaults}~.Xdefaults $@$NNc!W$r;2>H$7$F2<$5$$!%$^$?!$%*%W%7%g%s$O(J
$@!V(J\S~\ref{option}~$@%*%W%7%g%s!W$r;2>H$7$F2<$5$$!%(J
\medskip
\par
$@$J$*!$(J[]$@$OA*Br;h$rI=$7!$$=$N$&$A$N$$$:$l$+$r;XDj$7$F2<$5$$!JNc$($P!$(J[on, off]
$@$J$i!$(Jon $@$+(J off $@$N$$$:$l$+!K!%(J
\medskip
\begin{description}
\item[Tgif*PrintCommand:]$@%W%j%s%H%"%&%H%3%^%s%I(J
\medskip
\par
PostScript $@%U%!%$%k$r%W%j%s%H%"%&%H$9$k$H$-$K;HMQ$9$k%3%^(J
$@%s%I$r;XDj$7$^$9!%Nc$($P!$%W%j%s%?!V(Jnps1$@!W$K=PNO$9$k>l9g$O!$(J
\baselineskip=10pt
\begin{verbatim}
Tgif*PrintCommand:      lpr -Pnps1
\end{verbatim}
\medskip
\item[Tgif*ReverseVideo:][on, off]
\medskip
\par
$@!V(Jon$@!W$r;XDj$9$k$H(J $@!V(J$-$rv$@!W$HF1$8!%(J
\medskip
\item[Tgif*DrawCursor:][top\_left\_arrow, dotbox, crosshair]
\medskip
\par
$@IA2h%b!<%I;~$N%+!<%=%k$N7A>u$r;XDj$7$^$9!J%G%U%)%k%H$O!$(Jtop\_left\_arrow$@!K!%(J
\medskip
\item[Tgif*HideDragCursor:][on, off]
\medskip
\par
$@!V(Jon$@!W$r;XDj$9$k$H(J $@!V(J$-$hc$@!W$HF1$8!%(J
\medskip
\item[Tgif*Euc:][on, off]
\medskip
\par
$@!V(Jon$@!W(J $@$r;XDj$9$k$H(J $@!V(J$-$euc$@!W(J $@$HF1$8!%(J
\medskip
\item[Tgif*WhereToPrint:]LaTeXFigure
\medskip
\par
$@!V(JLaTeXFigure$@!W(J $@$r;XDj$9$k$H(J $@!V(J$-$eps$@!W(J $@$HF1$8!%(J
\medskip
\item[Tgif*XLibArc:][on, off]
\medskip
\par
$@!V(Jon$@!W(J $@$r;XDj$9$k$H(J $@!V(J$-$larc$@!W(J $@$HF1$8!%(J
\medskip
\item[Tgif*NoKanji:][on, off]
\medskip
\par
$@!V(Jon$@!W(J $@$r;XDj$9$k$H(J $@!V(J$-$nokanji$@!W(J $@$HF1$8!%(J
\medskip
\item[Tgif*Foreground:]$@?'$NL>A0(J
\medskip
\baselineskip=10pt
\begin{verbatim}
Foreground $@%+%i!<$r;XDj$9$k!%(J
$@!JNc!K(J
        Tgif*Foreground:        yellow
\end{verbatim}
\medskip
\item[Tgif*Background:]$@?'$NL>A0(J
\medskip
\par
Background $@%+%i!<$r;XDj$9$k!%(J
\medskip
\item[Tgif*BorderColor:]$@?'$NL>A0(J
\medskip
\par
$@OH@~$N?'$r;XDj$9$k!%(J
\medskip
\item[Tgif*AutoKinput:][on, off]
\medskip
\par
$@!V(Jon$@!W(J $@$r;XDj$9$k$H(J $@!V(J$-$autokinput$@!W(J $@$HF1$8!%(J
\medskip
\medskip
\item[Tgif*SaveWithEPS:][on, off]
\medskip
\par
$@!V(Jon$@!W(J $@$r;XDj$9$k$H(J $@!V(J$-$savewitheps$@!W(J $@$HF1$8!%(J
\medskip
\medskip
\item[Tgif*InitialFont:][Times, Courier, Helvetica, NewCentury, Symbol, Minchou, Gothic]
\medskip
\par
$@%G%U%)%k%H$N%U%)%s%H$r;XDj$9$k!J%G%U%)%k%H$O!$(JCourier$@!K(J
\medskip
\item[Tgif*CurrentDirectory:]$@%Q%9(J
\medskip
\par
$@!V%Q%9!W$G;XDj$5$l$?%G%#%l%/%H%j$,%+%l%s%H%G%#%l%/%H%j$K$J$j$^$9!%(J
\medskip
\item[Tgif*LineStyle:][PLAIN, RIGHT, LEFT, DOUBLE]
\medskip
\par
$@%G%U%)%k%H$N%i%$%s%9%?%$%k$r;XDj$9$k!J%G%U%)%k%H$O!$(JRIGHT$@!K!%(J
\medskip
\item[Tgif*LineWidth:][0, 1, 2, 3, 4, 5, 6]
\medskip
\par
$@%G%U%)%k%H$N@~I}$r;XDj$7$^$9!J%G%U%)%k%H$O!$(J0$@!K!%(J
\medskip
\item[Tgif*KanjiFontReadSkip:]on
\medskip
\par
\verb+~+/.Xdefaults $@$G!$4A;z%U%)%s%H@_Dj$r$7$J$$>l9g!JAa$$OC$,%G%U%)%k%H$N(J
$@%U%)%s%H$N@_Dj$G(J tgif+ $@$r;H$&>l9g!K$K!$$3$NJQ?t$r!V(Jon$@!W(J
$@$K$9$k$H!$(Jtgif+ $@$NN)$A>e$,$j$,<c43Aa$/$J$j$^$9!%$9$J$o$A!$$3$l$,(J
$@@_Dj$5$l$F$$$J$$$H!$0J2<$N%(%s%H%j$,(J \verb+~+/.Xdefaults $@$K@_Dj$5$l$F$$$k(J
$@2DG=@-$,$"$k$H;W$C$FD4$Y$K$$$/$+$i$G$9!%(J
\medskip
\item[Tgif*KanjiMinchou8:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 8  $@$NL@D+BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$(Jk14$@!K!%(J
\medskip
\item[Tgif*KanjiMinchou10:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 10 $@$NL@D+BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$(Jk14$@!K!%(J
\medskip
\item[Tgif*KanjiMinchou12:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 12 $@$NL@D+BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$(Jk14$@!K!%(J
\medskip
\item[Tgif*KanjiMinchou14:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 14 $@$NL@D+BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$(Jk14$@!K!%(J
\medskip
\item[Tgif*KanjiMinchou16:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 16 $@$NL@D+BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$(Jneck16$@!K!%(J
\medskip
\item[Tgif*KanjiMinchou18:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 18 $@$NL@D+BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$L5$7!K!%(J
\medskip
\item[Tgif*KanjiMinchou24:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 24 $@$NL@D+BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$(Jneck24$@!K!%(J
\medskip
\item[Tgif*KanjiGothic8:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 8  $@$N%4%7%C%/BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$L5$7!K!%(J
\medskip
\item[Tgif*KanjiGothic10:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 10 $@$N%4%7%C%/BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$L5$7!K!%(J
\medskip
\item[Tgif*KanjiGothic12:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 12 $@$N%4%7%C%/BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$L5$7!K!%(J
\medskip
\item[Tgif*KanjiGothic14:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 14 $@$N%4%7%C%/BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$L5$7!K!%(J
\medskip
\item[Tgif*KanjiGothic16:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 16 $@$N%4%7%C%/BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$L5$7!K!%(J
\medskip
\item[Tgif*KanjiGothic18:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 18 $@$N%4%7%C%/BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$L5$7!K!%(J
\medskip
\item[Tgif*KanjiGothic24:]$@O@M}%U%)%s%HL>(J
\medskip
\par
$@%5%$%:(J 24 $@$N%4%7%C%/BN$N4A;z%U%)%s%HL>!J%G%U%)%k%H$O!$L5$7!K!%(J
\medskip
\end{description}
\newpage
\section{Xdefaults $@$G$N4A;z%U%)%s%H$N;XDj$K$D$$$F(J}
\label{howToSetKanji}
\medskip
\par
\verb+~+/.Xdefaults $@$G!$(Jtgif+ $@$G;H$o$l$k4A;z%U%)%s%H$rJQ$($k$3$H$,$G$-$^$9!J(J
$@!VIUO?(J~\ref{xdefaults}~ X DEFAULTS$@!W;2>H!K!%(J
\medskip
\par
$@4A;z%U%)%s%H$O!$L@D+BN$H%4%7%C%/BN$N(J2$@<oN`$N%U%)%s%H$,;XDj$G$-$^$9!%$3$l$i(J
$@$O!$(JX $@C<Kv$J$I$GFH<+$K(J ROM $@Ey$K4A;z%U%)%s%H$r;}$C$F$$$k>l9g$K;XDj$9$k$HNI(J
$@$$$G$7$g$&!%(J
\medskip
\par
$@%U%)%s%H$N;XDjJ}K!$O!$%5%$%:(J18 $@$N>l9g$,FC<l$G!$;D$j$N%5%$%:$OF1$8$G$9!%(J
\medskip

$@%5%$%:(J 16 $@$N%U%)%s%H$O!$%5%$%:(J 18 $@$N%U%)%s%H$N(J
$@BeMQ$K;H$&$3$H$,$G$-$^$9!%(J\\
$@!V(JTgif*KanjiMinchou16:$@!W$d!V(JTgif*KanjiGothic16:$@!W$O!$$=$N(J
$@$?$a$N%U%)%s%HL>$r;XDj$9$k$b$N$G$9!JCm0U(J: tgif+ $@$G!$%5%$%:(J 16 $@$N%U%)%s(J
$@%H$,;H$($k$o$1$G$O$"$j$^$;$s!%%5%$%:(J 18 $@$N%U%)%s%H$NI=<($K!$%5%$%:$N6a$$(J
$@%5%$%:(J 16 $@$N%U%)%s%H$r;H$&$H$$$&$3$H$G$9!%!K!%(J

$@Nc$($P!$!V(JTgif*KanjiMinchou18:$@!W$O!$%U%)%s%HL>$,@_Dj$5$l$F$$$J$$$H!$<+F0E*$K!V(J
Tgif*KanjiMinchou16:$@!W$N%U%)%s%H$,;H$o$l!$J8;z4V$NI}$rD4@0$9$k$3$H$G!$%5%$%:$,9g$o(J
$@$;$i$l$^$9!J%4%7%C%/BN$bF1MM!K!%$3$3$G!$(J\\
$@!V(JTgif*KanjiMinchou16:$@!W$N%U%)%s%H$,$J$$$H!$(Jtgif+ $@$N%G%U%)%k%H!J(Jneck16$@!K(J
$@$,C5$5$l!$$=$l$b$J$$$H!$(J
Fake $@%U%)%s%H5!G=$K$h$j!$<+F0E*$K!V(Jk14$@!W$N%U%)%s%H$,;HMQ$5$l$^$9!J$=$N>l9g(J
$@$O!$%U%)%s%H%5%$%:$,0c$$2a$.$^$9$N$G!$J8;zOH$,$^$o$j$KI=<($5$l$^$9!K!%(J

\par
$@4JC1$K=q$/$H<!$N$h$&$J=gHV$G%U%)%s%H$,C5$5$l$^$9!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item Tgif*KanjiMinchou18: $@$G@_Dj$5$l$F$$$k%U%)%s%H(J
\item Tgif*KanjiMinchou16: $@$G@_Dj$5$l$F$$$k%U%)%s%H(J
\item tgif+ $@$N%G%U%)%k%H!J(Jneck16$@!K(J
\item k14
\medskip
\end{enumerate}
}
\par
$@%5%$%:(J 18 $@0J30$O!$%U%)%s%H$NC5$7J}$OF1$8$G!$<!$N$h$&$K$J$j$^$9!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item \verb+~+/.Xdefaults $@$G;XDj$7$?%U%)%s%H(J
\item tgif+ $@$N%G%U%)%k%H$N%U%)%s%H(J
\item k14
\medskip
\end{enumerate}
}
\par
$@$^$?!$%4%7%C%/BN$N%U%)%s%H$O!$%U%)%s%HL>$,@_Dj$5$l$F$$$J$$$H!$<+F0E*$K(J
$@F1$8%5%$%:$NL@D+BN$N%U%)%s%H$,;H$o$l!$I}$r(J1$@%T%/%;%k8|$/$7$FI=<($7$^$9(J
$@!J?^(J~\ref{fakeFont} $@;2>H!K!%(J
\medskip
\par
$@3F%5%$%:$N%U%)%s%H$KBP$7$F!$(J\verb+~+/.Xdefaults $@$G!$$I$&@_Dj$9$l$PNI$$$+$O!$<!(J
$@$N$h$&$JI=$K$J$j$^$9!J!VM-L5!W$O!$(JX $@%5!<%P$,$=$N%9%?%$%k$N%U%)%s%H$r(J
$@;}$C$F$$$k$+$I$&$+$rI=$7$^$9!K!%(J
\medskip
~\\
\begin{center}
\begin{tabular}{|r|r|l|}
\hline
$@L@D+BN(J & $@%4%7%C%/BN(J & .Xdefaults $@$N@_Dj(J                 \\
\hline\hline
\multicolumn{1}{|c|}{  $@M-(J  } & \multicolumn{1}{|c|}{    $@M-(J    } & $@N>J}@_Dj2DG=(J                      \\
\hline
\multicolumn{1}{|c|}{  $@M-(J  } & \multicolumn{1}{|c|}{    $@L5(J    } & $@L@D+BN$@$1(J .Xdefaults $@$K@_Dj$9$k(J  \\
\hline
\multicolumn{1}{|c|}{  $@L5(J  } & \multicolumn{1}{|c|}{    $@M-(J    } & $@%4%7%C%/BN$N%U%)%s%H$rBP1~$9$k%5%$(J\\
       &            & $@%:$NL@D+BN$N(J .Xdefaults $@$K@_Dj$9$k(J\\
\hline
\multicolumn{1}{|c|}{  $@L5(J  } & \multicolumn{1}{|c|}{    $@L5(J    } & $@2?$b@_Dj$7$J$$(J                    \\
\hline
\end{tabular}\\
\end{center}
\medskip
\par
$@Nc$($P!$(JXMiNT $@$N>l9g$O!$%5%$%:(J 18 $@$N%4%7%C%/BN$N%U%)%s%H$H!$%5%$%:(J 24
$@$NL@D+BN$N%U%)%s%H$r;}$C$F$$$^$9!%$=$N>l9g$K$O!$<!$N$h$&$K$J$k$G$7$g$&!%(J
\medskip
\begin{description}
\item[Tgif*KanjiMinchou18:]$-$rom$-$fixed$-$medium$-$r$-$normal$-$$-$18$-$130$-$100$-$100$-$c$-$180$-$jisx0208.1983$-$0
\item[Tgif*KanjiMinchou24:]$-$rom$-$fixed$-$medium$-$r$-$normal$-$$-$24$-$170$-$100$-$100$-$c$-$240$-$jisx0208.1983$-$0
\medskip
\medskip
\end{description}
\newpage
\section{Xdefauts $@$NNc(J}
\label{dXdefaults}
\medskip

\subsection{$@$b$C$H$b4JC1$J$b$N(J}
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% cat .Xdefaults
Tgif*PrintCommand:      lpr -Pnps1
\end{verbatim}
\end{nquote}
\medskip
\par
$@$3$3$G!$(Jnps1 $@$O!$(JPostScript $@%W%j%s%?$G$9!%(J
\medskip

\subsection{$@$A$g$C$H6E$C$?$b$N(J}
\medskip
\par
$@%G%U%)%k%H$N%W%j%s%H%"%&%H@h$r!V(J.eps $@%U%!%$%k!W$K$7$F!$IA2h%b!<%I$N%+!<%=%k$N7A(J
$@$r%/%m%9%X%"!J==;z!K%+!<%=%k$K$9$k!%(Jtgif+ $@$N%&%#%s%I%&$NGX7J?'$r2+?'!J(J
$@$3$l$O!$$"$/$^$G$bNc$G$9(J:$-$$@!K$K$9$k!%(J
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% cat .Xdefaults
Tgif*PrintCommand:      lpr -Pnps1
Tgif*WhereToPrint:      LaTeXFigure
Tgif*Background:        yellow
Tgif*DrawCursor:        crosshair
\end{verbatim}
\end{nquote}
\medskip
\medskip

\subsection{CISC $@HG(J EWS 4800 $@MQ$N$b$N(J}
\medskip
\par
$@=tHL$N;v>p$+$i!$$A$g$C$HCm0U$,I,MW$G$9!J:GDc8B!$0J2<$N$b$N$r=q$$$F2<$5$$!K!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% cat .Xdefaults
Tgif*PrintCommand:      lpr -Pnps1
Tgif*ReverseVideo:      off
Tgif*DrawCursorColor:   red
\end{verbatim}
\end{nquote}
\medskip

\subsection{$@$A$J$_$KKM$N>l9g$O!%!%!%(J}
\medskip
\par
SUN3 $@$N9b2rA|EY%b%N%/%m%G%#%9%W%l%$$G(JX $@%&%#%s%I%&$r;HMQ$7$F$$$^$9!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% cat ~matsuda/.Xdefaults
Tgif*InitialFont:               Minchou
Tgif*SaveWithEPS:               on
Tgif*AutoKinput:                on
Tgif*ReverseVideo:              on
Tgif*PrintCommand:              lpr -Pnps1

Kinput*ReverseVideo:            off
\end{verbatim}
\end{nquote}
\newpage
\section{kinput $@$N;H$$J}(J}
\label{kinput}
\medskip
\par
$@$3$3$G$O!$(Jkinput $@$r;H$&>e$G:GDc8B$N@bL@$r$7$^$9!%>\$7$/$O!$(Jkinput $@$N%^%K%e%"%k(J
$@$r;2>H$7$F2<$5$$!%(J
\medskip

\subsection{kinput $@$N5/F0(J}
\medskip
\par
tgif+ $@$G!$$+$J4A;zJQ49MQ$N(J X $@%/%i%$%"%s%H$G$"$k(J kinput $@$rMQ$$$FF|K\8lF~NO$r9T$J$&>l9g$K$O!$(J
$@!V%7%U%H(J+$@%9%Z!<%9!W(J $@$H$d$kA0$K(J kinput $@$r5/F0$9$kI,MW$,$"$j$^$9!%(J
$@$3$l$O!$(Jtgif+ $@$r5/F0$7$?8e$G$b$+$^$$$^$;$s(J.
\medskip
kinput $@$,!$(JWnn $@$H(J tgif+ $@$N%$%s%?%U%'!<%9$r<h$j;}$C$F$/$l$^$9!J?^(J~\ref{tgifAndJapanese}~$@;2>H!K!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/tgif+Japa.eps}
        \caption{tgif+ $@$H(J kinput $@$N@\B3(J}
        \label{tgifAndJapanese}
   \end{center}
\end{figure}
\medskip
\par
$@$^$:!$(Jkinput $@$r5/F0$9$kA0$K!$4D6-JQ?t$N@_Dj$r9T$$$^$9!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% setenv JSERVER spls39
% setenv CC_DEF /usr/local/X11R4/lib/X11/ccdef/ccdef.kinput
\end{verbatim}
\end{nquote}
$@@_Dj$7=*$o$C$?$i(J\footnote{$@$3$l$i$O!$EvK\It$N4D6-$G$N@_Dj$G$9$N$G!$Cm0U$7$F2<$5$$!%(J
kinput $@$r%G%U%)%k%H$N%3%s%U%#%0$G%$%s%9%H!<%k$9$k$H!V(JCC\_DEF$@!W$O!$(J
$@!V(J/usr/lib/X11/ccdef/ccdef.kinput $@!W(J $@$K$J$j$^$9!%>\$7$/$O!$%$%s%9%H!<%k$7$??M$K(J
$@J9$$$F2<$5$$!%(J}$@!$<!$N$h$&$K$7$F(J kinput $@$r5/F0$7$^$9!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% kinput &
\end{verbatim}
\end{nquote}
\medskip
\par
$@$J$*!$(JJSERVER $@$K$O!$%P!<%8%g%s(J4 $@$N(J Wnn $@$,F0$$$F$$$k%[%9%HL>$r;XDj$7$F2<$5$$!%(J
kinput $@$N<+F05/F05!G=$K4X$7$F$O!$!V(J\S~\ref{autoKinput}~kinput $@$N<+F05/F05!G=!W;2>H!%(J
\medskip
\par
kinput $@$d(J Wnn $@$,$I$N$h$&$K%$%s%9%H!<%k$5$l$F$$$k$+$O!$%^%7%s4IM}<T$KJ9$$$F2<$5$$!%(J
\medskip

\subsection{$@$+$J4A;zJQ49$N%U%m!<(J}
\medskip
$@%G%U%)%k%H$N(J ccdef.kinput $@$G(J  kinput $@$r5/F0$7$?>l9g$O?^(J~\ref{kkFlow}~
$@$N$h$&$K$J$j$^$9!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/kkFlow.eps}
        \caption{$@$+$J4A;zJQ49%U%m!<(J}
        \label{kkFlow}
   \end{center}
\end{figure}
\medskip
\medskip

\subsection{$@$+$J4A;zJQ49Iw7J(J}
\medskip
\par
kinput $@5/F08l!$(Jtgif+ $@$G!$J8;zF~NO%b!<%I$K$7$F!$!V%7%U%H(J+ $@%9%Z!<%9!W(J
$@$H$9$k$H!$(Jkinput $@$H@\B3$rD%$j!$!V$+$J4A;zJQ49%b!<%I!W$K$J$j!$?^(J~\ref{kinputWin}~
$@$N$h$&$J%&%$%s%I%&$,I=<($5$l$^$9!J?^(J~\ref{tgifAndKinput}~$@$b;2>H!K!%(J

\begin{figure}[htbp]
   \begin{center}
        \epsfile{file=FIGURE/kinputWin.eps}
        \caption{$@$+$J4A;zJQ49%&%#%s%I%&(J}
        \label{kinputWin}
   \end{center}
\end{figure}
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@J8;z$rF~NO$7$F$$$/$H%m!<%^;z$KJQ49$7$F$/$l$^$9!%(J
\item $@!V%3%s%H%m!<%k(J + J$@!W(J $@$G!$4A;zJQ49$7$^$9!J8uJd$NI=<($b!V(J\verb+^+J$@!W$G$9!K!%(J
\item $@!V%3%s%H%m!<%k(J + L$@!W$G3NDj$7$FF~NO$5$l$^$9!%(J
\item $@%+%?%+%J$O!$BgJ8;z$GF~NO$7$^$9!%(J
\medskip
\end{enumerate}
}
$@$^$?!$;z<o$N@ZBX$($O!$!V(Jtab$@!W$G9T$$$^$9(J
\footnote{kinput $@$N%G%U%)%k%H$N%-!<%P%$%s%I$N$-$i$$$JJ}$O!$!V(Jccdef.kinput$@!W(J
$@$H$=$N%U%!%$%k$G%$%s%/%k!<%I$7$F$$$k%U%!%$%k$r%[!<%`$K%3%T!<$7!$%U%!%$%k(J
$@$rJQ99$7$F2<$5$$!%(J}$@!%%b!<%I$N=*N;$b!V%7%U%H(J+$@%9%Z!<%9!W$G$9!%(J
\medskip

\subsection{$@Cm0U(J}
\label{kinputNotice}
\medskip
\par
kinput $@$r5/F0$9$k$H!$J8;zF~NO$O$9$Y$F(J kinput $@$,<h$j$^$9$N$G!$(J
tgif+ $@$K$O!$J8;z$OAw$i$l$^$;$s!%Nc$($P!$(Jkinput $@$N%&%#%s%I%&$,I=<($5$l$F$$$k(J
$@;~$K!$%U%!%$%k$rJ]B8$7$h$&$H$7$F!V(J\verb+^+S$@!W$HF~NO$7$F$b%U%!%$%k$O!$J]B8$5$l$^$;$s!%(J
$@$3$l$O!$(Jkinput $@$,!V(J\verb+^+S$@!W$r<h$C$F$7$^$C$F!$(Jtgif+ $@$KEO$5$J$$$+$i$G$9(J
\medskip
\par
$@$=$N$h$&$J>l9g$K$O!$0lEY(J kinput $@$H$N@\B3$r@Z$k!J%7%U%H(J+$@%9%Z!<%9!K(J
$@$+!$%^%&%9$G%a%K%e!<$r;H$C$F9T$C$F2<$5$$!%(J
\medskip
\par
$@$J$*!$(Jkinput $@$H$N@\B3$r@Z$C$F$b!$(Jkinput $@$r(J kill $@$7$J$$8B$j!$:F$S(J
$@!V%7%U%H(J+$@%9%Z!<%9!W$G@\B3$rD%$k$3$H$,$G$-$^$9!%(J
\medskip
\medskip
\medskip
\newpage
\section{tgif+ $@$N%$%s%9%H!<%k4D6-!$%$%s%9%H!<%kJ}K!(J}
\label{install}
\medskip

\subsection{$@%P%$%J%j$N%$%s%9%H!<%k4D6-(J}
\medskip
\par
tgif+ $@$N%G%U%)%k%H$N%$%s%9%H!<%k4D6-$O!$<!$N$h$&$K$J$C$F$$$^$9!J%=!<%9(J
$@%3!<%I$NCV$$$F$"$k%G%#%l%/%H%jEy$O!$EvK\It%m!<%+%k$J$N$G!$Cm0U$,I,(J
$@MW$G$9!K!%(J
\medskip
\begin{itemize}
\item $@%3%^%s%I!$%i%$%V%i%j!$%^%K%e%"%k(J
~\\
\begin{tabular}{|l|l|}
\hline
$@%3%^%s%I(J   & /usr/local/bin     \\
\hline
$@%i%$%V%i%j(J & /usr/local/lib/tgif\\
\hline
$@%^%K%e%"%k(J & /usr/local/man     \\
\hline
\end{tabular}\\
\medskip
\item $@%=!<%9%3!<%I(J
\medskip
~\\
\begin{tabular}{|r|r|r|}
\hline
ftp $@%5!<%P!J(Jnecspl$@!K(J & \multicolumn{1}{|l|}{tgif $@!J(J2.2$@!K(J  } & \multicolumn{1}{|l|}{pub/tgif$-$2.2.tar.Z  }\\
\cline{2-2}\cline{3-3}                     & tgif+$@!J(J2.2.4$@!K(J & pub/tgif+2.2.4.tar.Z\\
\hline
\end{tabular}\\
\medskip
\item tgif+ $@$N5/F0$KI,MW$J%U%!%$%k(J
\medskip
~\\
\begin{tabular}{|l|l|}
\hline
$@%3%^%s%I(J   & tgif+, prtgif+                  \\
\hline
$@%i%$%V%i%j(J & .psmac, .pskanjiEUC, .pskanjiEXT\\
\hline
\end{tabular}\\
\medskip
\end{itemize}

\subsection{tgif+ $@$N%$%s%9%H!<%kJ}K!(J}
\medskip
\par
tgif+ $@$N%$%s%9%H!<%k$N;EJ}$O!$DL>o$N(JX$@$N%"%W%j%1!<%7%g%s$HF1$8$G$9!%(J
\medskip
\par
$@%=!<%9$KIU?o$9$k(J Imakefile $@Fb$N<!$N(J3$@$D$NJQ?t$r%?!<%2%C%H%^%7%s$N%G%#%l(J
$@%/%H%j9=@.$K$"$o$;$FJQ99$7$F2<$5$$!JI=Cf$N%G%U%)%k%H$O!$%=!<%9%3!<%I$N(J Imakefile
$@$N$b$N$G$9!K!%(J
\medskip
~\\
\begin{tabular}{|l|l|l|}
\hline
$@JQ?t(J             & $@FbMF(J                                    & \multicolumn{1}{|c|}{ $@%G%U%)%k%H(J                    }\\
\hline
BINDIR           & $@%3%^%s%I$rCV$/%G%#%l%/%H%j(J              & /usr/local/bin                 \\
\hline
MANPATH          & $@%^%K%e%"%k$rCV$/%G%#%l%/%H%j(J            & /usr/local/man                 \\
\hline
TGIFDIR          & PostScript $@$N%^%/%mEy$rCV$/%G%#%l%/%H%j(J & /usr/local/lib/tgif            \\
\hline
CCDEF\_FILE       & kinput $@$N%G%U%)%k%H$N(J ccdef $@%U%!%$%k(J    & /usr/lib/X11/ccdef/ccdef.kinput\\
\hline
JSERVERHOST\_NAME & jserver(Wnn4) $@$N2TF/$7$F$$$k%[%9%HL>(J    & spls41                         \\
\hline
\end{tabular}\\
\medskip
$@$3$l$i$NJQ?t$NJQ99$,=*$C$?8e(J\footnote{CCDEF\_FILE $@$N@_Dj$O!$(Jkinput $@$N%$%s%9%H!<%k$7$?4D6-$K9g$o$;$F2<$5$$!%(J
$@$^$?!$%$%s%9%H!<%k$5$l$F$$$J$$$N$G$"$l$P!$2?$r@_Dj$7$F$b9=$o$J$$$N$G$9$,!$(J
$@>-Mh%$%s%9%H!<%k$5$l$k$+$b$7$l$J$$$N$G!$0l1~!$!$I=$N%G%U%)%k%H$N@_Dj$K$7$F2<$5$$!%(J}
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% xmkmf
% make install
% make install.man
\end{verbatim}
\end{nquote}
\medskip
\par
$@$H$9$l$P!$(Jtgif+ $@$N%P%$%J%j@8@.$5$l!$%$%s%9%H!<%k$5$l$^$9!%(J
\medskip
\par
$@$3$3$G!$(Jxmkmf $@$O!$(JImakefile $@$+$i(J Makefile $@$r@8@.$9$k%3%^%s%I$G$9!J(J
X11R4 $@$N%G%#%9%H%j%S%e!<%7%g%s%F!<%W$KIU?o$7$F$$$^$9!K!%(Jxmkmf$@$,$J$$>l(J
$@9g$K$O!$%=!<%9$K$D$$$F$/$k(J Makefile.noimake $@$^$?$O!$(JMakefile $@!J(JSUN4 $@$G:n@.$7$?$b$N$G$9!K$rE,Ev$K=$@5$7$F;H$C$F2<$5$$!%(J
$@$^$?!$(JRISC EWS4800 $@!J(JSVR4$@%Y!<%9!K$N>l9g$O!$(JMakefile.RE $@$r!$(J
CISC EWS4800 $@!J(JSVR2.2 $@%Y!<%9!K$N>l9g$O(J Makefile.CE $@$r;H$C$F2<$5$$!K!%(J
\medskip
\par
$@Nc$($P!$(JRISC EWS4800 $@$N>l9g$O!$<!$N$h$&$K$J$j$^$9!%(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% make -f Makefile.RE install
% make -f Makefile.RE install.man
\end{verbatim}
\end{nquote}
\medskip
\par
$@$J$*!$%P%$%J%j$N@8@.$@$1$r9T$J$$$?$$>l9g$O(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% make
\end{verbatim}
\end{nquote}
\medskip
\par
$@$^$?!$(JRISC EWS4800 $@$N>l9g$O!$(J
\medskip
\begin{nquote}{8ex}
\baselineskip=10pt
\begin{verbatim}
% make -f Makefile.RE
\end{verbatim}
\end{nquote}
\medskip
\par
$@$H$9$l$P!$%P%$%J%j$,@8@.$5$l$^$9!%(J
\medskip

\subsection{$@%=!<%9%3!<%ICf$N(J ifdef $@$N35MW(J}
\medskip
\par
$@%=!<%9%3!<%ICf$G!$%*%j%8%J%k$N(J tgif $@$KJQ99$r9T$C$?ItJ,$O!$$9$Y$F(J
\#ifdef$@!J$^$?$O!$(J\#ifndef$@!K$G0O$^$l$F$$$^$9!%3F(J define $@$N0UL#$O!$<!$NDL(J
$@$j$G$9!%(J
\medskip
~\\
\begin{center}
\begin{tabular}{|l|l|}
\hline
define & $@FbMF(J                           \\
\hline
UC     & tgif $@$N(J UI $@$r$h$/$9$k$?$a$NJQ99(J\\
KINPUT & kinput $@$r;HMQ$9$k$?$a$NJQ99(J    \\
KANJI  & $@F|K\8l(J $@$r;HMQ$9$k$?$a$NJQ99(J    \\
ONTON  & $@K\Mh$O:o=|$9$Y$-JQ99(J           \\
SYSV   & System V $@$G;HMQ$9$k$?$a$NJQ99(J  \\
\hline
\end{tabular}\\
\end{center}
\medskip
\medskip
\newpage
\section{xwd $@$N7k2L$r(J X11 $@$N%S%C%H%^%C%W7A<0!J(J.xbm$@!KJQ49$9$kJ}K!(J}
\label{generateXbm}
\medskip
\par
pbmplus $@$rMQ$$$?>l9g$NJ}K!$r<($7$^$9!%(Jpbmplus $@$O!V(J Extended Portable Bitmap
Toolkit$@!W$G!$?t==<oN`%$%a!<%8%G!<%?$N%3%s%P!<%?$+$i$J$j$^$9(J\footnote{
$@EvK\It!$$*$h$S(Jnecspl $@$+$i(J ftp $@2DG=$JJ}$O!$(Jpub/expo-contrib/pbmplus.tar.Z $@$+$iF~<j$9$k$3$H$,2DG=$G$9!%(J}$@!%(J
\medskip
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item $@Gr9u$N>l9g(J
\medskip
\baselineskip=10pt
\begin{verbatim}
% xwd | xwdtopnm | pbmtoxbm > bitmap.xbm
\end{verbatim}
\medskip
\par
$@$H$9$k$H!V(Jbitmap.xbm$@!W$KJQ49$5$l$?7k2L$,F@$i$l$^$9!%(J
\medskip
\item $@%+%i!<$N>l9g(J
\medskip
\par
$@%+%i!<$N>l9g$O!$0lC6Gr9u$KJQ49$9$kI,MW$,$"$j$^$9!%(J
\medskip
\baselineskip=10pt
\begin{verbatim}
% xwd | xwdtopnm | ppmtopgm | pgmtopbm | pbmtoxbm > color.xbm
\end{verbatim}
\medskip
\par
$@$H$9$k$H!V(Jcolor.xbm$@!W$KJQ49$5$l$?7k2L$,F@$i$l$^$9!%(J
\medskip
\end{enumerate}
}
\par
$@$?$@$7!$(Jtgif $@%P!<%8%g%s(J 2.4 $@$+$i$O!$%+%i!<$N%$%a!<%8%G!<%?$r%T%C%/%9%^%C%W$b(J
$@JQ49$7$F<h$j9~$a$k$h$&$K$J$C$F$$$^$9!%(J
\medskip
\medskip
\newpage
\section{$@%H%i%V%k(J}
\label{trouble}
\medskip
\par
$@$$$/$D$+$h$/J9$+$l$k<ALd$,$"$j$^$9$N$G!$2r7hJ}K!$r<($7$F$*$-$^$9!%(J
\medskip

\subsection{$@%U%)%s%H$,$J$$(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\medskip
\par
$-$adobe$-$courier$-$medium$-$r$-$normal$-$$-$18$-$180$-$75$-$75$-$m$-$110$-$iso8859$-$1 $@$H$+$$$&(J
($@$b$7$/$O!$$3$l$K;w$?L>A0$N!K%U%)%s%H$,$J$$$H$$$o$l$F(Jtgif+$@$,F0$+$J$$!%(J
\medskip
\item[$@2r7hK!(J:]~
\medskip
\par
$@$3$l$O!$(JX$@C<Kv$K$h$/$"$k8=>]$G$9!%4pK\E*$K$O!$%U%)%s%H%Q%9$N@_Dj$r(J
$@$A$c$s$H@_Dj$7$F$$$J$$$N$,860x$G$9!%(J
\medskip
\par
\% xset q $-$display $@%[%9%HL>(J:0
\medskip
\par
$@$H$d$k$H!$:G8e$NJ}$K%U%)%s%H%Q%9$,I=<($5$l$^$9!%(J
\medskip
\par
$@!JNc!K(J
\par
Font Path:
\par
/usr/lib/X11/fonts/misc/, /usr/lib/X11/fonts/75dpi/, /usr/lib/X11/fonts/100dpi/
\medskip
\par
$@$3$3$G!$(J
\par
$@!V$J$s$H$+(J/75dpi/$@!W$H(J $@!V$J$s$H$+(J/100dpi/$@!W$N(J
\par
$@@_Dj$,$5$l$F$$$J$$>l9g$,$"$j$^$9!J$J$<$+(J75dpi$@$NJ}$,@_Dj$5$l$F$$$J$$(J
$@>l9g$,B?$$!K!%$G!$$3$l$i$r@_Dj$7$F$d$i$J$$$H$$$1$J$$$o$1$G$9$,!$(J
$@$3$3$N!V$J$s$H$+!W$K$"$?$k%G%#%l%/%H%j$O!$(JX$@C<Kv$J$i$P!$$=$N%[%9%H%^%7%s(J
$@$N$I$3$+$K$"$k$o$1$G!$$=$l$O!$5!<o$d@_Dj$K$h$C$F0[$J$j$^$9!%(J
\medskip
\par
$@$3$l$O!$(Jxset q $@$GD4$Y$?;~$K!$$9$G$K@_Dj$5$l$F$$$k$d$D$r$^$M$7$F(J
\medskip
\par
xset fp+ $@$J$s$H$+(J/75dpi/, $@$J$s$H$+(J/100dpi/
\medskip
\par
$@$H$d$l$P$$$$$O$:$G$9!%2?$b@_Dj$5$l$F$$$J$$>l9g$O!$$=$N(JX$@C<Kv$r(J
$@%3%s%U%#%0$7$??M$KJ9$$$F2<$5$$!J!V(JSetUp$@!W2hLL$GJ,$+$k>l9g$b$"$j$^$9!K!%(J
\medskip
\medskip
\medskip
\end{description}

\subsection{kinput $@$,F0$+$J$$(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\medskip
\par
kinput $@$,F0$+$J$$!%(J
\medskip
\item[$@2r7hK!(J:]~
\medskip
\par
$@4D6-JQ?t(J JSERVER$@!$(JCC\_DEF $@$r@_Dj$9$k!%FC$K!$(JJSERVER $@$O!$(JWnn4 $@$,F0$$(J
$@$F$$$k%[%9%H$G$J$$$H%3%M%/%H$G$-$^$;$s!%EvK\It$G$O!$(Jspls39 $@$H(J
$@$+(Jspls35 $@$H$+$,$=$&$@$H;W$$$^$9!%>\$7$/$O!$%^%7%s4IM}<T$KJ9$$$F2<$5(J
$@$$!%(JCC\_DEF $@$O!$EvK\It$N?M$O(J
/usr/local/X11R4/lib/X11/ccdef/ccdef.kinput $@$G$9!%(J
\medskip
\par
$@5/F0$O!$(Jtgif+ $@Fb$G!$!V%7%U%H(J+$@%9%Z!<%9!W$G(J kinput $@$N%&%#%s%I%&$,3+$+(J
        $@$l$^$9!J!VIUO?(J~\ref{kinput}~kinput $@$N;H$$J}!W;2>H!K!%(J
\medskip
\end{description}

\subsection{.eps $@%U%!%$%k$,@8@.$5$l$J$$(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\par
$@!V(J\verb+^+P$@!W(J $@$b$7$/$O!$!V(JFile$@!W%a%K%e!<$+$i!V(JPrint$@!W$rA*$s$G$b(J eps $@%U%!(J
$@%$%k$,@8@.$5$l$J$$!%(J
\medskip
\item[$@2r7hK!(J:]~
\medskip
\par
$@!V(JChoice $@%&%#%s%I%&!W$N%"%$%3%s$r!V(JLaTeX$@%"%$%3%s!W$K$7$F!$!V(JFile$@!W(J
        $@%a%K%e!<$G!$!V(JPrint$@!W$7$F2<$5$$!J!V(J\S~\ref{files}~$@3F%U%!%$%k$N@8@.$N;EJ}!W;2>H!K!%(J
\medskip
\end{description}

\subsection{$@%W%j%s%H%"%&%H$,$G$-$J$$(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\par
$@!V(J\verb+^+P$@!W(J $@$b$7$/$O!$!V(JFile$@!W%a%K%e!<$+$i!V(JPrint$@!W$rA*$s$G$b%W%j%s%?$K(J
$@=P$J$$!%(J
\medskip
\item[$@2r7hK!(J:]~
\end{description}
{
\renewcommand{\theenumi}{(\arabic{enumi})}
\renewcommand{\labelenumi}{\theenumi}
\begin{enumerate}
\item .Xdefaults $@$KITHw$,$"$k>l9g(J
\medskip
\par
\verb+~+/.Xdefaults $@$K(J $@<!$N0l9T$r2C$($F2<$5$$!%(J
\medskip
\baselineskip=10pt
\begin{verbatim}
Tgif*PrintCommand:              lpr -Pnps1
\end{verbatim}
\medskip
\par
$@$3$3$G!$!V(Jnps1$@!W$O!$(JPostScript $@%W%j%s%?$G$9!%(J
\medskip
\item $@%W%j%s%H%"%&%H@h$,%W%j%s%?$K$J$C$F$$$J$$>l9g(J
\medskip
\par
$@!V(JChoice $@%&%#%s%I%&!W$N%"%$%3%s$r!V(JPrinter$@%"%$%3%s!W$K$7$F%W%j%s%H(J
        $@$7$F$_$F2<$5$$!J(J\S~\ref{SaveFiles}~$@;2>H!K!%(J
\item $@F|K\8l%U%)%s%H$,$J$$>l9g(J
\medskip
\par
$@!V(J$-$euc$@!W$+!V(JTgif*Euc: on$@!W$K$7$F$d$C$F$_$F2<$5$$!%(J
\medskip
\medskip
\end{enumerate}
}

\subsection{$@J8;z$N%U%)%s%H$,JQ$o$i$J$$(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\par
$@F~NO$7$F$$$k;~$K!$%U%)%s%H$rJQ$($h$&$H$7$?$,JQ$o$i$J$$!%(J
\medskip
\item[$@2r7hK!(J:]~
\par
$@0lEY!$%+!<%=%k$r!VLp0u!W$K$7$F!$$=$NJ8;zNs$rA*$s$G2<$5$$!%$=$&$7$F$+$i!$(J
        $@%U%)%s%H$rJQ99$7$F2<$5$$!J!V(J\S~\ref{changeFontChar}~$@J8;z$N%U%)%s%H$rJQ$($k!%!W;2>H!K!%(J
\medskip
\end{description}

\subsection{tgif+ $@$,%O%s%0%"%C%W$7$F$7$^$C$?(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\par
$@JQ$J$3$H$r$7$?$i!$(Jtgif+ $@$,%O%s%0%"%C%W$7$F$7$^$C$?!%(J
\medskip
\item[$@2r7hK!(J:]~
\par
tgif+ $@$r5/F0$7$?(J kterm $@$J$j(J xterm $@$K<!$N$h$&$J%a%C%;!<%8$,I=<((J
\medskip
\baselineskip=10pt
\begin{verbatim}
Input an operation and a sub command.
\end{verbatim}
\medskip
\par
$@$5$l$F$$$k>l9g$O!$<B9T$7$?(J kterm $@$J$j(J xterm $@$K$$$C$F(J $@%j%?!<%s(J
$@$rF~NO$7$F2<$5$$!%(J
\medskip
\end{description}

\subsection{tgif+ $@$r;HMQCf$K(J tgif+ $@$,;`$s$G$7$^$C$?(J! $@$5$F!$:#$^$GIA$$$F$$$??^$O!)(J}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\par
$@IA2hCf$K!$(JX$@%5!<%P$,Mn$A$F$7$^$C$?!%$^$?!$IA2hCf$K(J tgif+ $@$,;`$s(J
$@$G$7$^$C$?!%(J
\medskip
\item[$@2r7hK!(J:]~
\par
$@$H$j$"$($:!$$$$:$l$K$;$h;`$s$G$7$^$C$?$N$O$7$g$&$,$J$$$N$G$9$,!$(J
tgif+ $@$O!$ITN8$N;vBV$G(J tgif+ $@$,;`$LA0$K(J $@!V(JEmergensySave.obj$@!W(J
(System V $@$N>l9g$O!$!V(JEmergensy.obj$@!W$G$9!K(J
$@$K%U%!%$%k$rJ]B8$7$^$9!%$3$N%U%!%$%k$,$G$-$F$$$J$$$+D4$Y$F$_$F(J
$@2<$5$$(J
\medskip
\end{description}

\subsection{$@%S%C%H%^%C%W%G!<%?$r4^$`?^$r%W%j%s%H%"%&%H$7$h$&$H$7$?$i=P$J$+$C$?(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\par
$@F1>e(J
\medskip
\item[$@2r7hK!(J:]~
\par
$@$3$l$O!$(Jtgif+ $@$N%P%0$G$O$"$j$^$;$s$,!$Bg$-$J%S%C%H%^%C%W$r%W%j(J
$@%s%H%"%&%H$7$h$&$H$9$k$H!$(JPostScript $@%$%s%?%W%j%?$,=hM}$7$-$l(J
$@$:$K%W%j%s%H%"%&%H$G$-$J$$$3$H$,$"$j$^$9!%?^$r>.$5$/$9$k$+!$=L(J
$@>.$7$F;n$7$F$_$F2<$5$$!%(J
\medskip
\end{description}

\subsection{$@4]$d;M3Q$rIA$/>l9g$K!$%+!<%=%k$,8+$($J$$(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\medskip
\par
$@F1>e(J
\medskip
\item[$@2r7hK!(J:]~
\medskip
\par
tgif+ $@$G$O!$IA2h%b!<%I;~$K%+!<%=%k$N?'$rGr$/$9$k$h$&$K$7$?$N$G$9$,!$(J
$@$"$k<o$N(J X $@%5!<%P$O!$%+!<%=%k$K%^%9%/$rIU$1$F$/$l$J$$$N$G!$(J
$@GrCO$NGrEF$G!$8+$($J$/$J$C$F$7$^$$$^$9!%(J
\medskip
\par
$@$H$j$"$($:!$(Jtgif+ $@$O!$%&%#%s%I%&$N?'$r(J .Xdefaults $@$K;XDj$G$-$k$N$G!$(J
Background $@%+%i!<$KGr$H9u0J30$N?'!JNc$($P!$@D$H$+!%!%!K$r;XDj$7$F;H$&(J
$@$h$&$K$7$F2<$5$$!%(J
\medskip
\par
$@!JNc!K(J
\baselineskip=10pt
\begin{verbatim}
Tgif*Foreground: white
Tgif*Background: blue
Tgif*BorderColor: white
\end{verbatim}
\par
$@$^$?!$<!$N$h$&$J1#$7%3%^%s%I$b$"$j$^$9!%(J
\medskip
\medskip
\baselineskip=10pt
\begin{verbatim}
Tgif*DrawCursorColor:   [blue, red, green...]
\end{verbatim}
\medskip
\par
$@Gr0J30$N$*9%$-$J?'$r(J .Xdefaults $@$K=q$$$F$*$$$F2<$5$$!%(J
\medskip
\end{description}

\subsection{$@1_$,I=<($G$-$J$$(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\par
$@$"$kBg$-$50J>e$N1_$,IA$1$J$$(J
\medskip
\item[$@2r7hK!(J:]~
\medskip
\par
$@$3$l$O!$(JX11R3 $@$N%5!<%P$G$h$/5/$3$k8=>]$G$9!%(J
$@%3%^%s%I%i%$%s$G(J $@!V(J$-$larc$@!W(J $@$r;XDj$9$k$+!$(J
\medskip
\baselineskip=10pt
\begin{verbatim}
Tgif*XLibArc:  on
\end{verbatim}
\medskip
\par
$@$r;XDj$7$F2<$5$$!%(J
\medskip
\end{description}

\subsection{$@!V%3%s%H%m!<%k(J+$@$J$s$H$+!W!$!V%a%?(J+$@$J$s$H$+!W$,8z$+$J$$(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\medskip
\par
\verb+^+A $@$d(J \verb+^+O $@$J$I$,F~NO$G$-$J$$$,!$%^%&%9$G%a%K%e!<$O=P$9$3$H$O$G$-$k!%(J
\medskip
\item[$@2r7hK!(J:]~
\medskip
\par
$@!V%a%?(J+$@$J$s$H$+!W$@$1$,8z$+$J$$>l9g$O!$!V%-%c%W%9%m%C%/!W$,2!$5$l$F$$$k(J
$@>l9g$,9M$($i$l$^$9!%(J
\medskip
\par
$@N>J}8z$+$J$$>l9g$O!$(Jkinput $@$,5/F0$5$l$F$$$^$;$s$+!)(J kinput $@$,5/F0$5$l$F$$$k$H!V(J
Title$@%&%#%s%I%&!W$K!V(J*KINPUT*$@!W$HI=<($5$l$F$$$^$9!%(J
\medskip
\par
kinput $@$O!$$9$Y$F$N%-!<F~NO$r<h$C$F$7$^$&$N$G!$5/F0Cf$OCm0U$,(J
$@I,MW$G$9!%(Jtgif+ $@$G$O!$!V%F%-%9%HF~NO%b!<%I!W$r$d$a$k$H<+F0E*$K(J
kinput $@$H$N%3%M%/%7%g%s$r@Z$k$h$&$K$J$C$F$$$^$9$N$G!$(J $@J8;zF~NO$,=*$o$C(J
\end{description}
        $@$?;~E@$G!$!VA`:n%b!<%I!W$K$b$I$7$F$*$/$HNI$$$G$7$g$&!J!VIUO?(J~\ref{kinputNotice}~
$@Cm0U!W;2>H!K!%(J
\medskip
\medskip

\subsection{$@?^$r(JjLaTeX $@$K<h$j9~$s$G%W%j%s%H%"%&%H$9$k$H!$?^$,@Z$l$k(J!}
\medskip
\begin{description}
\item[$@>I>u(J:]~
\medskip
\par
tgif+ $@$GIA$$$??^$r!$(JjLaTeX $@$NJ8=q$K<h$j9~$s$G%W%j%s%H%"%&%H$9$k$H(J
$@?^$N$O$8$,>/$7@Z$l$FHa$7$$!%(J
\medskip
\item[$@2r7hK!(J:]~
\medskip
\par
$@$3$l$O!$?^$N(J Bounding Box $@!J?^$rJq4^$9$k:G>.$N;M3Q7A!K$N7W;;$,$*$+$7$/$F5/$3$k8=>]$G$9$,!$(J
$@K\Mh$J$i5/$-$J$$$O$:$N$b$N$G$9!%$^$"!$$=$&$J$C$?>l9g$O!$K\Mh(J
$@$,$I$&$3$&$K$+$+$o$i$:!$Ha$7$$$N$G!$:FEY(J tgif+ $@$r5/F0$7$F?^$N(J
$@1&2<$N$O$8$K;M3Q!J4]$G$b9=$$$^$;$s!K$rIA$$$F2<$5$$!%$=$N$^$^$@(J
$@$H%W%j%s%H%"%&%H$G0u:~$5$l$F$7$^$&$N$G!$;M3Q$N@~$rF)L@$K$7$F2<(J
$@$5$$!%$3$&$9$k$H!$(JBounding Box $@$,Bg$-$/$J$k$N$GBg>fIW$G$9!%(J
\medskip
\medskip
\end{description}
\newpage
\section{jLaTeX $@$X$N$b$&0l$D$N<h$j9~$_J}(J}
\medskip
\par
$@K\J8Cf$G$O!$!V(J$\backslash$epsfile$@!W$rMQ$$$?<h$j9~$_J}$r>R2p$7$^$7$?$,!$(J
$@$3$l0J30$K!V(J$\backslash$psfig$@!W$d!V(J$\backslash$special$@!W$rMQ$$$F<h$j9~$_J}$,$"$j$^$9!%$=$l$i(J
$@$NNc$H$7$F!$(Jtgif $@$N%=!<%9$KIU?o$9$k!V(Jexample.tex$@!W$r0J2<$K0zMQ$7$F$*$-(J
$@$^$9!%(J
\baselineskip=10pt
\begin{verbatim}
%
% @(#)$Header: /tmp_mnt/n/kona/tangram/u/william/X11/TGIF2/RCS/example.tex,v 2.0 91/03/05 12:50:47 william Exp $
%
\documentstyle[12pt]{article}
%
\oddsidemargin=0in
\textwidth=6.5in
\topmargin=0in
\textheight=609pt
\parskip=14pt
\setlength{\unitlength}{0.5cm}
\begin{document}
%
Figure \ref{fig:psfig} is an example of a tgif \LaTeX\,\, figure printed with
the ``psfig'' macro.  As you can see, the figure fits nicely between the
surrounding text.
%
\begin{figure*}[htb]
\input{psfig}
\centerline{\psfig{figure=an-sr-flip-flop.eps}}
\caption{Example Tgif Diagram Printed with ``psfig'' Macro.\label{fig:psfig}}
\end{figure*}
%
The bounding box seems to be pretty tight.  In
Figure \ref{fig:epsf} below, the same figure is
generated with the ``epsf'' construct.
%
\begin{figure*}[htb]
\input{epsf}
\centerline{\epsfile{an-sr-flip-flop.eps}}
\caption{Example Tgif Diagram Printed with ``epsf'' Macro.\label{fig:epsf}}
\end{figure*}
%
All these are just trying the illustrate that the bounding boxes
generated by tgif is pretty tight, and there's really no need for
the ``psfile'' construct.  I don't know if there are cases where
some documentation generation software only supports the ``psfile''
construct, but I an putting in the example anyways.

Figure \ref{fig:psfile} is an example of a tgif \LaTeX\,\, figure
printed with the ``psfile'' macro.  Notice the \\rule command.
%
\begin{figure*}[htb]
\special{psfile=FIGURE/"an-sr-flip-flop.eps" hoffset=-40}
\rule{0in}{1.1in}
\caption{Example Tgif Diagram Printed with ``psfile'' Macro.\label{fig:psfile}}
\end{figure*}
%
It is reserving a space of 1.1 inch for the figure.  This can
be figured out from the ruler within tgif.  As a general rule,
the ``psfile'' construct should be avoided.
\end{document}
\end{verbatim}
\medskip
\medskip
\newpage
\section{$@%-!<%P%$%s%I0lMw(J}
\label{keyBinding}
\medskip
~\\
\begin{tabular}{ll}
$@%-!<(J & \multicolumn{1}{c}{ $@5!G=(J}\\
 & \\
\verb+^+A   & $@A4$F$N%*%V%8%'%/%H$rA*Br$9$k!%(J\\
\verb+^+B   & $@A*Br$7$?%*%V%8%'%/%H$r8e$m$K;}$C$F$$$/!%(J\\
\verb+^+C   & $@%I%a%$%s(J ($@%+%l%s%H%G%#%l%/%H%j(J) $@$rJQ99$9$k(J\\
\verb+^+E   & $@%+%l%s%H%b!<%I$r%W%C%7%e$9$k!J%H%0%k!K!%(J\\
\verb+^+D   & $@A*Br$7$?%*%V%8%'%/%H$rJ#@=$9$k!%(J\\
\verb+^+F   & $@A*Br$7$?%*%V%8%'%/%H$rA0$K;}$C$F$/$k!%(J\\
\verb+^+G   & $@A*Br$7$F$$$k%*%V%8%'%/%H$r%0%k!<%W2=$9$k!%(J\\
\verb+^+I   & $@%S%k%G%#%s%0%V%m%C%/%*%V%8%'%/%H$r%$%s%9%?%s%92=$9$k!%(J\\
\verb+^+K   & $@>e$N%l%Y%k$KLa$j!$%7%s%\%k%U%!%$%k$r%/%m!<%:$9$k(J (\verb+^+V $@$N5U(J) $@!%(J\\
\verb+^+L   & $@A*Br$7$?%*%V%8%'%/%H$r%"%i%$%s$9$k!%(J\\
\verb+^+N   & $@?7$7$$!$L>A0$N$J$$(J.obj $@%U%!%$%k$r%*!<%W%s$9$k!%(J\\
\verb+^+O   & .obj $@%U%!%$%k$r!$%(%G%#%C%H$9$k$?$a$K%*!<%W%s$9$k!%(J\\
\verb+^+P   & $@%+%l%s%H%Z!<%8$r%W%j%s%H$9$k!%(J\\
\verb+^+Q   & tgif $@$r=*N;$9$k!%(J\\
\verb+^+R   & $@%Z!<%8$r:FIA2h$9$k!%(J\\
\verb+^+S   & $@%+%l%s%H%*%V%8%'%/%H$"$k$$$O%7%s%\%k$r%U%!%$%k$K%;!<%V$9$k!%(J\\
\verb+^+T   & $@A*Br$7$?%*%V%8%'%/%H$r%0%j%C%I$K%"%i%$%s$9$k!%(J\\
\verb+^+U   & $@A*Br$7$?%*%V%8%'%/%H$r%"%s%0%k!<%W2=$9$k!%(J\\
\verb+^+V   & $@%S%k%G%#%s%0%V%m%C%/%*%V%8%'%/%H(J ($@%"%$%3%s%*%V%8%'%/%H$G$"$kA*Br(J\\
     & $@$7$?%*%V%8%'%/%H(J) $@$NDj5AItJ,$r%W%C%7%e(J ($@$"$k$$$O%(%G%#%C%H(J) $@$9$k!%(J\\
\verb+^+W   & $@IA2h%b!<%I$r%F%-%9%H$K$9$k!%(J\\
\verb+^+X   & $@A*Br$7$?%*%V%8%'%/%H$r:o=|$9$k!%(J\\
\verb+^+Y   & $@A*Br$7$?%*%V%8%'%/%H$r%+%C%H%P%C%U%!$K%3%T!<$9$k!%(J\\
\verb+^+Z   & $@%I%i%$%P$K%(%9%1!<%W$9$k!%(J\\
\verb+^+,   & $@:8$X%9%/%m!<%k$9$k!%(J\\
\verb+^+.   & $@1&$X%9%/%m!<%k$9$k!%(J\\
\verb+^+$-$   & $@;XDj$7$?%3%^%s%I$G%+%l%s%H%Z!<%8$r%W%j%s%H$9$k!%(J\\
\verb+^+/   & $@%0%j%C%I$NE@$NI=<(4V3V$r%0%j%C%I$N4V3V$K$9$k!J%H%0%k!K!%(J\\
\#A   & $@A*Br$7$?J#;($J%*%V%8%'%/%H$K!$A*Br$7$?B0@-$r%"%?%C%A$9$k!%(J\\
\#B   & $@%I%i%$%P$K%(%9%1!<%W$9$k!%(J\\
\#C   & $@A*Br$7$?%*%V%8%'%/%H$rH?;~7W2s$j$K(J 90 $@EY2sE>$9$k(J\\
\#D   & $@%0%j%C%I%5%$%:$r8:$i$9!%(J\\
\#E   & $@A*Br$7$?%]%j%i%$%s$K%H!<%/%s$rAw$k!%(J\\
\#F   & $@A*Br$7$?%]%j%i%$%s$r%U%i%C%7%e$9$k!%(J\\
\#G   & $@%0%j%C%I%]%$%s%H$rI=<($9$k(J / $@I=<($7$J$$!%(J\\
\#H   & $@A*Br$7$?%*%V%8%'%/%H$r?eJ?$K%U%j%C%W$9$k!%(J\\
\#I   & $@%0%j%C%I%5%$%:$rA}$d$9!%(J\\
\#J   & $@A*Br$7$?%*%V%8%'%/%H$NB0@-$NL>A0$r1#$9!%(J\\
\#K   & $@%b!<%I$r!VA*Br!W$K$9$k!%(J\\
\#L   & $@9T%9%?%$%k$H9TI}$r%]%C%W%"%C%W%a%K%e!<$KEO$9!%(J\\
\#M   & $@A*Br$7$?%*%V%8%'%/%H$NB0@-$rF0$+$9!%(J\\
\#N   & $@A*Br$7$?%*%V%8%'%/%H$NA4$F$NB0@-L>$rI=<($9$k!%(J\\
\#O   & $@%:!<%`%"%&%H!%(J\\
\end{tabular}\\
~\\
\begin{tabular}{ll}
\#P  & $@%+%l%s%H%U%!%$%k$K(J .obj $@%U%!%$%k$rFI$_9~$`!%(J\\
\#Q  & $@IA2h%b!<%I$r%]%j%i%$%s(J / $@3+$$$?%9%W%i%$%s$K$9$k!%(J\\
\#R  & $@IA2h%b!<%I$r;M3Q7A$K$9$k!%(J\\
\#S  & $@%I%i%$%P$X$N%(%9%1!<%W!%(J\\
\#T  & $@A*Br$7$?%*%V%8%'%/%H$NB0@-$r%G%?%C%A$9$k!%(J\\
\#U  & $@D>A0$K9T$J$C$?%*%Z%l!<%7%g%s$r<h$j>C$9!%(J\\
\#V  & $@A*Br$7$?%*%V%8%'%/%H$rD>3Q$K%U%j%C%W$9$k!%(J\\
\#W  & $@A*Br$7$?%*%V%8%'%/%H$r;~7W2s$j$K(J 90 $@EY2s$9!%(J\\
\#X  & $@%I%i%$%P$K%(%9%1!<%W$9$k!%(J\\
\#Y  & $@%I%i%$%P$K%(%9%1!<%W$9$k!%(J\\
\#Z  & $@%:!<%`%$%s!%(J\\
\#0  & $@%U%)%s%H%5%$%:$r(J 8 (11) $@$KJQ99$9$k!%(J\\
\#1  & $@%U%)%s%H%5%$%:$r(J 10 (14) $@$KJQ99$9$k!%(J\\
\#2  & $@%U%)%s%H%5%$%:$r(J 12 (17) $@$KJQ99$9$k!%(J\\
\#3  & $@%U%)%s%H%5%$%:$r(J 14 (20) $@$KJQ99$9$k!%(J\\
\#4  & $@%U%)%s%H%5%$%:$r(J 18 (25) $@$KJQ99$9$k!%(J\\
\#5  & $@%U%)%s%H%5%$%:$r(J 24 (34) $@$KJQ99$9$k!%(J\\
\#,  & $@%9%/%m!<%k%"%C%W!%(J\\
\#.  & $@%9%/%m!<%k%@%&%s!%(J\\
\#/  & $@A`:n;~$K%*%V%8%'%/%H$N%5%$%:$rI=<($9$k!J%H%0%k!K!%(J\\
\verb+^+\#A & $@A*Br$7$?%]%j%i%$%s!J%]%j%4%s!K$"$k$$$O%9%W%i%$%s$K@)8fE@$r2C$($k!%(J\\
\verb+^+\#B & $@%F%-%9%H%9%?%$%k$r%\!<%k%I$K$9$k!%(J\\
\verb+^+\#C & $@%F%-%9%H$N%"%i%$%s$r??Cf$K$9$k!%(J\\
\verb+^+\#D & $@A*Br$7$?%]%j%i%$%s!J%]%j%4%s!K$"$k$$$O%9%W%i%$%s$+$i@)8fE@$r:o=|$9$k!%(J\\
\verb+^+\#E & $@IA2h%b!<%I$r!$3Q4];M3Q7A$K$9$k!%(J\\
\verb+^+\#F & $@A*Br$7$?%S%C%H%^%C%W%*%V%8%'%/%H$r%j%P!<%9$9$k(J\\
\verb+^+\#G & $@%0%j%C%I%]%$%s%H$X%9%J%C%W$9$k!J%H%0%k!K!%(J\\
\verb+^+\#I & $@A*Br$7$?J#;($J%*%V%8%'%/%H$r%"%$%3%s$K$9$k!%(J\\
\verb+^+\#J & $@A*Br$7$?%"%$%3%s%*%V%8%'%/%H$r%0%k!<%W%*%V%8%'%/%H$K$9$k!%(J\\
\verb+^+\#K & $@%W%j%s%H%"%&%H$r%+%i!<$K$9$k$+Gr9u$K$9$k$+$rA*Br$9$k!%(J\\
\verb+^+\#L & $@%F%-%9%H$N%"%i%$%s$r:8$K$9$k!%(J\\
\verb+^+\#M & $@A*Br$7$?J#;($J%*%V%8%'%/%H$r%7%s%\%k$K$9$k!%(J\\
\verb+^+\#N & $@A*Br$7$?%7%s%\%j%C%/%*%V%8%'%/%H$r%0%k!<%W%*%V%8%'%/%H$K$9$k!%(J\\
\verb+^+\#O & $@%F%-%9%H%9%?%$%k$r%m!<%^%s$K$9$k!%(J\\
\verb+^+\#P & $@%F%-%9%H%9%?%$%k$r%\!<%k%I%$%?%j%C%/$K$9$k!%(J\\
\verb+^+\#Q & $@IA2h%b!<%I$r%]%j%4%s(J/ $@JD$8$?%9%W%i%$%s$K$9$k!%(J\\
\verb+^+\#R & $@%F%-%9%H$N%"%i%$%s$r1&$K$9$k!%(J\\
\verb+^+\#S & $@?7$7$$L>A0$G%U%!%$%k$r%;!<%V$9$k!%(J\\
\verb+^+\#T & $@%F%-%9%H%9%?%$%k$r%$%?%j%C%/$K$9$k!%(J\\
\verb+^+\#U & $@A*Br$7$?%*%V%8%'%/%H$N%"%$%3%sI=8=$r%"%C%W%G!<%H$9$k!%(J\\
\verb+^+\#V & $@IA2h%b!<%I$r1_$K$9$k!%(J\\
\verb+^+\#W & $@B?3Q7A"+"*%9%W%i%$%s!J%H%0%k!K!%(J\\
\verb+^+\#X & $@$5$^$6$^$J=PNO%U%!%$%k%U%)!<%^%C%H$r=[4D$9$k!%(J\\
\verb+^+\#Y & $@%+%C%H%P%C%U%!$+$i%Z!<%9%H$9$k!%(J\\
\verb+^+\#Z & $@IA2h%b!<%I$r1_8L$K$9$k!%(J\\
\verb+^+\#. & .xbm $@%U%!%$%k$r<h$j9~$`!%(J\\
\end{tabular}\\
                    \begin{center}
                        \epsfile{file=FIGURE/tgificon.eps}
                    \end{center}
\medskip
\newpage
\section{tgif+ $@$N%3%T!<%i%$%H(J}
\medskip
\par
tgif+ $@$N%3%T!<%i%$%H$O!$(Jtgif $@$N$=$l$K=>$$$^$9!%(J tgif $@$N%3%T!<%i%$%H$r(J
$@0J2<$K0zMQ$7$^$9!%(J
\medskip
\baselineskip=10pt
\begin{verbatim}
Copyright (C) 1990, 1991, William Chia-Wei Cheng
\end{verbatim}
\medskip
\baselineskip=10pt
\begin{verbatim}
William Cheng ("Author") grants to the party hereby receiving "tgif" from
an authorized provider ("Recipient") a non-exclusive, royalty-free
license to copy, display, distribute, and produce derivative works of
"tgif", provided that the above copyright notice appears in all copies
made of "tgif" and both the copyright notice and this license appear
in supporting documentation, and that the name of Author not be used
in advertising or publicity pertaining to "tgif".
\end{verbatim}
\medskip
\baselineskip=10pt
\begin{verbatim}
"Tgif" is provided "as is" without express or implied warranty. Author
does not and cannot warrant the performance of "tgif" or the results
that may be obtained by its use or its fitness for any specific use by
Recipient or any third party.  In no event shall Author become liable
to Recipient or any other party, for any loss or damages,
consequential or otherwise, including but not limited to time, money,
or goodwill, arising from use or possession, authorized or not, of
"tgif" by Recipient or any other party.
\end{verbatim}
\medskip
\medskip
\end{document}
